IFRS Accounting Standards in Practice

IFRS 18 Presentation and Disclosure in Financial Statements (Presentation in the Statement of Profit or Loss)

a ha



2024/2025



Contents

1.	1.1.	r <mark>oduction</mark> Overview of changes introduced by IFRS 18 Background on Primary Financial Statements Project	5
2.	2.1.	erview Key changes introduced by IFRS 18 Topics substantially unchanged from IAS 1	. 7
3.		ssification of Income and expenses	
	3.1.	General considerations	
		3.1.1. Classification of income and expenses - general requirements	
		3.1.2. Classification of income and expenses – entities with specified main business activities	. 12
	3.2.	Investing category	
		3.2.1. General classification requirements – specified assets	
		3.2.2. Specified income and expense classified in the investing category	. 13
		3.2.3. Practice aid – criteria to classify income and expenses in the investing category for entities without	
		specified main business activities	
		3.2.4. Investments in associates, joint ventures and unconsolidated subsidiaries3.2.5. Cash and cash equivalents	
		3.2.6. Other assets that generate a return individually and largely independently of the entity's other	. 17
		resources	17
		3.2.7. Assets that do not generate a return individually and largely independently of the entity's other	
		resources	. 18
		3.2.8. Derecognition of an asset or liability and remeasurement of an asset held for sale	
		3.2.9. Derecognition of groups of assets and liabilities and remeasurement of an asset held for sale	
		3.2.10. Change in use of an asset	20
		3.2.11. Practice aid - classification of gains and losses on the derecognition of an asset or liability (or group of	
		assets and liabilities), the remeasurement of an asset (or disposal group) as held for sale, or a change in use	
		3.2.12. Presentation of income and expenses classified in the investing category	22
	3.3.	Financing category	
		3.3.1. General classification requirements	
		3.3.2. Liabilities that arise from transactions that involve only the raising of finance	
		3.3.3. Liabilities that arise from transactions that do not involve only the raising of finance	
		3.3.4. Income and expenses recognised on the derecognition of liabilities	28
		3.3.5. Practice aid – criteria to classify income and expenses in the financing category for entities without	_
		specified main business activities	
		3.3.6. Exceptions to classification principles	
		3.3.7 Presentation of income and expenses classified in the financing category	30

<u>Linn</u>



	3.4. Operating category					
	3.5.	Income taxes category	32			
		Discontinued operations category				
	3.7.	Entities with specified main business activities – an entity's assessment of specified main business activities.	34			
		3.7.1. Purpose of specified main business activities requirements	.34			
		3.7.2. Level of assessment	.34			
		3.7.3. Multiple main business activities				
		3.7.4. Evidence to be considered when making assessment	35			
		3.7.5. Investing in associates, joint ventures and unconsolidated subsidiaries	. 37			
		3.7.6. Examples of entities with specified main business activities	. 37			
		3.7.7. Considerations applicable to groups	38			
		3.7.8. Change in assessment				
		3.7.9. Disclosures related to an entity's assessment of specified main business activities	40			
	3.8.	Additional requirements for entities with specified main business activities				
		3.8.1. Investing in assets as a main business activity				
		3.8.2. Providing financing to customers as a main business activity	42			
	3.9.	Considerations for specific items of income and expense				
		3.9.1. Cash and cash equivalents				
		3.9.2. Hybrid contracts containing a host that is a liability				
		3.9.3. Foreign exchange differences				
		3.9.4. Gains and losses on derivatives and designated hedging instruments				
		3.9.5. Guidance on common items of income and expense				
		3.9.6. Interim financial reports prepared in accordance with IAS 34	. 70			
4.	Prir	nciples of aggregation and disaggregation	71			
		The roles of the primary financial statements and notes				
	4.2.	Principles of aggregation and disaggregation				
		4.2.1. Line items required by IFRS Accounting Standards				
		4.2.2. Additional line items				
		4.2.3. Characteristics to use in assessing the need to aggregate and disaggregate information				
		4.2.4. Application of requirements to the statement of profit or loss	. 74			
		Line items in statement of profit or loss	. 74			
		Analysis of the most useful structured summary of operating expenses – use of characteristics of nature and	70			
	runc	tion.				
		4.4.1. Factors considered in assessing which presentation to use				
		4.4.2. Presentation of operating expenses by nature				
		4.4.3. Presentation of operating expenses by function	. 77			



	4.4.4. Presentation of operating expenses - mixed presentation4.4.5. Consistency of presentation, disclosure and classification	
4.5	 Labels and descriptions used. 4.5.1. Use of 'other'. 4.5.2. Labelling of subtotals. 	81
	tals and subtotals presented in the statement of profit or loss	
5.1.	Mandatory subtotals	
	5.1.1. Operating profit	
	5.1.2. Profit or loss before financing and income taxes	84
	5.1.3. Profit or loss before financing and income taxes – entities that provide financing to customers as a main business activity	84
5.2	. Additional subtotals	85
	5.2.1. Additional subtotals and management-defined performance measures	85
6. Ef	fective date and transition requirements	
6.1	. Required transition approach	87
	2. Disclosures required upon transition	
	8. Change in election for associates and joint ventures (IAS 28)	
6.4	Interim financial reports	
	6.4.1. Disclosures required upon transition – interim financial statements	90
	amples of statements of profit or loss prepared in accordance with IFRS 18	
	Example A – entity that does not have any specified main business activities	
	Example B – entity that invests in financial assets as a main business activity.	
7.4	 Example C – entity that provides financing to customers as a main business activity Example D – entity that provides financing to customers as a main business activity and invests in investm 	ent
	operty measured at fair value as a main business activity	
7.5	. Example E – insurance company that invests in financial assets as a main business activity	101
	ndices	
7.6	. Appendix 1 – Defined terms	. 105
Conta	act	107

1. Introduction

1.1. Overview of changes introduced by IFRS 18

IFRS 18 *Primary Financial Statements* was issued by the International Accounting Standards Board (IASB) in April 2024. IFRS 18 supersedes IAS 1 *Presentation of Financial Statements* and has resulted in numerous consequential amendments to IFRS® Accounting Standards including IAS 8 *Basis of Preparation of Financial Statements* (renamed from Accounting Policies, Changes in Accounting Estimates and Errors).

IFRS 18 introduces significant changes to numerous requirements, primarily how an entity:

- Presents its statement of profit or loss, including the classification of income and expenses and new mandatory subtotals;
- > Aggregates and disaggregates information disclosed in financial statements; and
- ▶ Discloses information about management-defined performance measures.

This publication focuses on the first two of the above items. A separate IFRS Accounting Standards In Practice publication is being prepared for management-defined performance measures. All of BDO's IFRS Accounting Standards In Practice publications may be accessed on BDO's IFRS and Corporate Reporting microsite.

IFRS 18 is mandatorily effective for annual reporting periods beginning on or after 1 January 2027 (i.e. 31 December 2027 annual financial statements for entities with calendar year-ends), subject to any relevant jurisdictional endorsement. Earlier application is permitted.

The issuance of IFRS 9 *Financial Instruments*, IFRS 15 *Revenue from Contracts with Customers*, IFRS 16 *Leases*, and IFRS 17 *Insurance Contracts* fundamentally changed the measurement requirements of IFRS Accounting Standards for many transactions entities commonly enter into, and IFRS 18 is expected to have a similarly significant effect on how entities present their financial statements.

While IFRS 18 does not change any of the recognition and measurement requirements in IFRS Accounting Standards, the effect of the new standard should not be underestimated by entities. The IASB has given preparers approximately 3 years to prepare for the adoption of the standard, which is a similar implementation timeframe similar to that given for the implementation of IFRS 9, 15 and 16.

Before exploring the requirements of IFRS 18 relating to the statement of profit or loss, it is useful to understand the history that led to the development and issuance of IFRS 18.

1.2. Background on Primary Financial Statements Project

IFRS 18 was developed as part of the primary financial statements project. This project was undertaken by the IASB because the board received feedback from stakeholders that IFRS Accounting Standards did not have detailed requirements on:

- classification of income and expenses in the statement of profit or loss;
- presentation of subtotals above 'profit or loss' (e.g. operating profit) in the statement of profit or loss; or
- > aggregation and disaggregation of information presented in the primary financial statements or disclosed in the notes.

This lack of detailed requirements led to diversity in practice, such as many entities presenting an 'operating profit' subtotal, with entities defining this subtotal differently. This lack of consistency made comparison of entities' financial performance difficult.

From 2015 to 2019, the IASB undertook research activities to determine how they may improve IFRS Accounting Standards to address the above noted concerns. The IASB decided to introduce a basis for classifying items of income and expense in the statement of profit or loss into categories, which would result in mandatory subtotals in the statement of profit or loss and increased comparability between entities.

The IASB also decided to require entities to disclose information about management-defined performance measures, which are subtotals of income and expenses that an entity uses in public communication that communicate management's view of an aspect of financial performance (e.g. 'adjusted profit'). IFRS 18 contains certain exclusions for specified subtotals, which do not trigger disclosures.

After performing research, the IASB issued an exposure draft *General Presentation and Disclosures* in December 2019.

The IASB deliberated on feedback received in response to the exposure draft and issued IFRS 18 in April 2024.





For a brief summary of IFRS 18 (and all IFRS Accounting Standards), refer to BDO's <u>IFRS Accounting Standards At</u> <u>a Glance publication</u>.

2.1. Key changes introduced by IFRS 18

Note that this section is not an exhaustive summary of all the effects of IFRS 18 and consequential amendments.

The most significant effects of IFRS 18 (and consequential amendments to IFRS Accounting Standards other than IFRS 18) relate to the following topics:

Key changes	Summary
Classification of income and expenses in the statement of profit or loss	 All income and expenses are classified into one of five categories: The investing category The financing category The operating category The income taxes category The discontinued operations category
	Income and expenses are generally classified based on the characteristic of the expense (i.e. the type of asset or liability to which the income or expense relates); however, certain exceptions exist for entities with specified main business activities, resulting in certain income and expenses being classified in the operating category that would otherwise be classified in the investing and/or financing categories.
	See section $\underline{3}$ of this publication.
Principles of aggregation and disaggregation	A new set of principles has been introduced for how assets, liabilities, equity, reserves, income, expenses and cash flows are aggregated and disaggregated. Applying these requirements results in aggregations and disaggregations of items being presented in the primary financial statements and disclosed in the notes.
	See section $\underline{4}$ of this publication.
Totals and subtotals presented in the statement of profit and loss	Once an entity has classified individual items of income and expense into the appropriate categories and it has aggregated those items into appropriate levels of aggregation for presentation in the statement of profit or loss, mandatory and additional subtotals are presented as a result of the previous steps. For example, all entities are required to present 'operating profit' in the statement of profit or loss, which is the total of all income and expenses classified in the operating category.
	See section $5 = 5$ of this publication.
Disclosure of management-defined	IFRS 18 requires entities to disclose information about management-defined performance measures (MPMs), which are a subtotal of income and expense that:
performance measures	 (a) an entity uses in public communications outside financial statements; (b) an entity uses to communicate to users of financial statements management's view of an aspect of the financial performance of the entity as a whole; and
	(c) is not listed in IFRS 18.118, or specifically required to be presented or disclosed by IFRS Accounting Standards.
	A common example of an MPM could be an 'adjusted profit' measure, which excludes share-based payments expenses and impairment of goodwill.
	A separate IFRS Accounting Standards In Practice publication is being prepared for management-defined performance measures.

Key changes	Summary
Consequential amendments to IAS 7 Statement of Cash Flows	For entities that use the indirect method to prepare their statement of cash flows, the starting point will be typically operating profit or loss. Prior to the adoption of IFRS 18, the starting point in the statement of cash flows was profit or loss.
	IFRS 18 also eliminates the classification options for interest and dividend cash flows, which will increase consistency.
Other changes	 Statement of financial position IFRS 18 requires goodwill to be presented as a separate line item in the statement of financial position, separate from other intangible assets. IAS 1 did not require this disaggregation (IAS 1.54(c)).
	 IAS 33 Earnings per Share Prior to the effective date of IFRS 18, IAS 33 permitted entities to disclose additional earnings per share amounts using a reported component of the statement of comprehensive income as the numerator. IFRS 18 amends IAS 33 and permits additional earnings per share amounts only when the numerator is: A total or subtotal specified by IFRS 18 in IFRS 18.69, 86 or 118; or A management-defined performance measure. Additional disclosure requirements in IAS 33 when an entity presents additional amounts per share other than basic and diluted earnings per share (e.g. adjusted operating profit per share). Disclosure requirements are more significant if the disclosed additional per share amount uses a management-defined performance measure as the numerator in the calculation.

2.2. Topics substantially unchanged from IAS 1

Many requirements in IAS 1 were brought forward into IFRS 18 (or other IFRS Accounting Standards) substantially unchanged. A significant number of requirements from IAS 1 have been relocated to IAS 8 Basis of Preparation of Financial Statements (renamed from Accounting Policies, Changes in Accounting Estimates and Errors).

Certain other requirements were relocated from IAS 1 to other IFRS Accounting Standards, such as IFRS 7 *Financial Instruments: Disclosures.*

Торіс	Summary
Most requirements applicable to the preparation of primary financial statements other than the statement of profit or loss	 IFRS 18 does not significantly affect how the other primary financial statements (e.g. statement of financial position, statement of cash flows, statement of changes in equity, etc.) are prepared other than the introduction of new aggregation and disaggregation requirements (see section <u>4</u>), with some exceptions, such as those noted above applicable to: The statement of financial position (separate presentation of goodwill); and The statement of cash flows (operating profit or loss being the starting point for the indirect method and the reduction of classification options for interest and dividend cash flows).
Fair presentation and compliance with IFRS Accounting Standards	Relocated to IAS 8 substantially unchanged.
Going concern	Relocated to IAS 8 substantially unchanged.
Offsetting	Substantially unchanged from IAS 1.
Frequency of reporting	Substantially unchanged from IAS 1.
Comparative information	Substantially unchanged from IAS 1.
Consistency of presentation	Substantially unchanged from IAS 1.

Торіс	Summary
Current/non-current classification of assets and liabilities	Substantially unchanged from IAS 1, including amendments made to IAS 1 affecting the classification of loans as current vs. non-current, which became effective for annual reporting periods beginning on or after 1 January 2024. See BDO's IFRS Accounting Standards In Practice Classification of Loans as Current or Non-Current.
Structure of notes	Substantially unchanged from IAS 1.
Disclosure of accounting policy information	Relocated to IAS 8 substantially unchanged.
Disclosures about judgements and sources of significant estimation uncertainty	Relocated to IAS 8 substantially unchanged.
Capital disclosures	Substantially unchanged from IAS 1.
Disclosures about puttable financial instruments classified as equity	Relocated to IFRS 7 substantially unchanged.
Miscellaneous other disclosures (declared dividends, cumulative preference dividends not recognised, domicile and legal form, description of the nature of the entity's operations, etc.)	Substantially unchanged from IAS 1.





3. Classification of Income and expenses



3.1. General considerations

3.1.1. Classification of income and expenses - general requirements

IFRS 18 requires entities to classify all items of income and expense into one of five categories (IFRS 18.47):

- ▶ The investing category (section <u>3.2</u>)
- ▶ The financing category (section 3.3)
- The operating category (section <u>3.4</u>)
- The income taxes category (section 3.5)
- The discontinued operations category (section <u>3.6</u>)

The structure of the statement of profit or loss is then based upon how items of income and expense are classified. For example, for an entity without specified main business activities where certain exceptions to the classification requirements apply (see section 3.8), a typical statement of profit or loss may be as follows (this example includes certain subtotals which are not mandatory, but are commonly used by entities, such as gross profit - see section 5):

Line item		Classification
Revenue	XXX	
Cost of sales	XXX	
Gross profit	XXX	
Other operating income	XXX	Operating category
Selling expenses	XXX	
Research and development	XXX	
General and administrative	<u>xxx</u>	
Operating profit	XXX	Mandatory specified subtotal
Fair value gains on investments in equity instruments	XXX	Investing category
Profit before financing and income taxes	XXX	Mandatory specified subtotal ¹
Interest expense on borrowings and lease liabilities	XXX	Financing category
Profit before income taxes	XXX	Additional subtotal
Income tax expense	XXX	Income taxes category
Profit from continuing operations	XXX	Additional subtotal
Loss from discontinued operations	XXX	Discontinued operations category
Profit	XXX	Mandatory total

¹ Profit before financing and income taxes is a mandatory subtotal for most entities, though an exception exists for entities that provide financing to customers as a main business activity and make a particular accounting policy choice – see section <u>5.1</u>.

BDO comment - labelling of categories

IFRS 18 does not require an entity to label items of income and expense based on the five categories in IFRS 18. For example, in the illustration above, the fair value gains on investments in equity instruments is classified in the investing category, however, the term 'investing category' does not appear in the statement of profit or loss. The classification of items of income and expense into the five categories noted above is used to produce the mandatory specified subtotals in IFRS 18 as these subtotals are based on how income and expenses are classified. See section <u>5.1</u>.

For example:

- 'Operating profit or loss' is required to be presented by IFRS 18.69(a), and it is defined as comprising all income and expenses classified in the operating category (IFRS 18.70).
- 'Profit or loss before financing and income taxes' is required to be presented by IFRS 18.69(b), and is defined as the total of (1) operating profit or loss (defined above) and (2) all income and expenses classified in the investing category (IFRS 18.71). Therefore, that subtotal comprises all income and expenses included in the operating and investing categories.

For more information on mandatory and additional subtotals, refer to section 5.

BDO comment - category names and inconsistency with the statement of cash flows

The operating, financing and investing categories are similarly titled to operating, financing and investing activities in IAS 7, however, the IASB did not aim to achieve alignment between how income and expenses are classified in the statement of profit or loss and how the associated cash flows are classified in the statement of cash flows. During the development of IFRS 18, some stakeholders suggested that the IASB avoid this matter by using different titles in IFRS 18 (e.g. 'non-operating' rather than the investing category), however, the IASB decided to use the category labels as proposed in the exposure draft (IFRS 18.BC87).

For example, an entity that operates a factory may sell items of property, plant and equipment, with the cash flows arising from the sale being classified as an investing activity in IAS 7 because the cash flows relate to 'the acquisition and disposal of long-term assets and other investments not included in cash equivalents' (IAS 7.6). However, the associated income or expense (the gain or loss on disposal of the item of property, plant and equipment) would typically be classified in the operating category in the statement of profit or loss because it would not meet the criteria to be classified in the investing category (IFRS 18.B48(a)) – see section <u>3.2</u>.

The aggregation and disaggregation requirements (i.e. how items of income and expense are presented within the statement of profit or loss) are addressed in section <u>4</u>. The requirements for presenting mandatory and additional subtotals are addressed in sections <u>5.1</u> and <u>5.2</u>.

For entities that do not have specified main business activities, items of income and expenses are generally classified into these five categories according to the characteristics of the items of income and expense and the assets and liabilities associated with them. For example, fair value gains and losses on financial assets (e.g. equity instruments) measured at fair value through profit or loss are classified in the investing category because of the nature of the asset; the financial asset generates a return individually and largely independently of the entity's other resources (IFRS 18.B46(a) and 53(c)).

BDO comment - operating category as the 'residual'

It is important to emphasise that the operating category is a 'residual category', meaning that items of income and expense are always classified into the operating category unless they meet criteria to be classified into another category (IFRS 18.52). Therefore, it is necessary for entities to determine whether each item of income and expense may be (or is required to be) classified into one of the other four categories.

Because the operating category is the default or residual category for all items of income and expense, this publication first explains the classification requirements applicable to the other two primary categories: investing (section 3.2) and financing (section 3.3).



3.1.2. Classification of income and expenses – entities with specified main business activities

IFRS 18 contains exceptions to the general classification requirements for entities that have **specified main business activities** (IFRS 18.49). That is to say, for certain entities, the primary activities that are undertaken to run its business will affect how items of income and expense are classified. For entities with specified main business activities, certain items of income and expense that would otherwise be classified in the investing and/or financing categories are classified in the operating category. For example, if an entity has a main business activity of investing in financial assets (i.e. particular types of assets - IFRS 18.49(a)), then the fair value gains and losses on financial assets (e.g. debt or equity instruments) measured at fair value through profit or loss are classified in the operating category.

The specified main business activities requirements of IFRS 18 may be depicted graphically as follows, with certain income and expenses being classified out of the investing and financing categories into the operating category:

Line item		Classification	
Revenue	XXX		_
Cost of sales	XXX		
Gross profit	XXX		/
Other operating income	XXX	Operating category	\leftarrow
Selling expenses	XXX		For entities with
Research and development	XXX		specified main
General and administrative	XXX		business activities, certain income and
Operating profit	XXX	Mandatory specified subtotal	expenses classified in the investing and
Fair value gains on investments in equity instruments	XXX	Investing category	financing categories may be classified in th operating category if
Profit before financing and income taxes	XXX	Mandatory specified subtotal ²	criteria are met.
Interest expense on borrowings and lease liabilities	XXX	Financing category	
Profit before income taxes	XXX	Additional subtotal	
Income tax expense	XXX	Income taxes category	
Profit from continuing operations	XXX	Additional subtotal	
Loss from discontinued operations	XXX	Discontinued operations category	
Profit	XXX	Mandatory total	

There are also special considerations for certain items of income and expense, such as income and expenses related to cash and cash equivalents, hybrid contracts containing a host that is a liability, foreign exchange differences, etc. These special considerations are discussed in section <u>3.9</u>.

Structure of Section 3

The following sections address how:

- ▶ Items of income and expense are classified into the five categories generally (sections <u>3.2</u> through <u>3.6</u>);
- An entity assesses whether it has specified main business activities (section <u>3.7</u>);
- ▶ If an entity does have specified main business activities, it applies additional requirements in classifying income and expenses (section <u>3.8</u>); and
- ▶ An entity classifies specific items of income and expense where special requirements apply (section <u>3.9</u>).

² Profit before financing and income taxes is a mandatory subtotal for most entities, though an exception exists for entities that provide financing to customers as a main business activity and make a particular accounting policy choice – see section 5.1.3.

3.2. Investing category

This section discusses the requirements applicable to entities that do not invest in assets as a main business activity. See section <u>3.8.1</u> for guidance that applies to entities with such specified main business activities.

3.2.1. General classification requirements – specified assets

Except when an entity has specified main business activities, an entity classifies in the investing category specified income and expenses from (IFRS 18.53) certain assets (i.e. specified assets):

(a) Investments in associates, joint ventures and unconsolidated subsidiaries;

- (b) Cash and cash equivalents; and
- (c) Other assets if they generate a return individually and largely independently of the entity's other resources.

'Specified income and expenses' restricts the types of income and expense that are classified in the investing category. Said another way, not all income and expenses that relate to the assets listed in IFRS 18.53 may be classified in the investing category.

Guidance on each of these three classes of assets is provided below.

3.2.2. Specified income and expense classified in the investing category

IFRS 18.53 requires that only <u>specified</u> income and expenses to be classified in the investing category if they arise from the three categories of assets noted above. These specified expenses are:

Income and expenses (IFRS 18.54)	Common examples (IFRS 18.54 and B47)	
(a) The income generated by the assets	 Interest Dividends Rental income 	
(b) The income and expenses that arise from the initial and subsequent measurement of the assets, including on derecognition of the assets	 Depreciation Impairment losses and reversal of impairment losses Fair value gains and losses 	
(c) The incremental expenses directly attributable to the acquisition and disposal of the assets	 Transaction costs on financial assets classified as fair value through profit or loss Costs to sell assets, such as broker commissions on financial instruments 	

IFRS 18.53, which lists three classes of assets and IFRS 18.54, which specifies types of income and expenses, must be applied together. Only income and expenses that are listed in 18.54 and that arise from the classes of assets listed in IFRS 18.53 may be classified in the investing category.

BDO comment – incremental expenses (IFRS 18.54(c))

IFRS 18.54(c) includes only 'incremental' expenses directly attributable to the acquisition and disposal of assets, therefore, not all directly attributable costs will meet the criteria to be classified in the investing category.

The IASB considered, but rejected alternative approaches, such as classifying all directly attributable expenses in the investing category, as this approach was seen as too complex and costly for an entity to apply. Such an approach could require, for example, an allocation of employee benefits if some employees engage in managing both an entity's investments and other activities (IFRS 18.BC109).

3.2.3. Practice aid – criteria to classify income and expenses in the investing category for entities without specified main business activities

This practice aid summarises the requirements to determine whether income or expenses are classified in the investing category, assuming that an entity does not have a main business activity of investing in such assets. As discussed in section 3.8.1, the requirements for classifying income and expense are significantly more complex in such cases.



Example 3.2-1 - classification of depreciation in the investing category

IFRS 18.B47 notes that depreciation is an example of 'income and expenses that arise from the initial and subsequent measurement of the assets', however, this does not mean that depreciation in general will be classified in the investing category.

For entities without specified main business activities, only depreciation relating to assets that meet the criterion in IFRS 18.53(c) may be classified in the investing category.

Fact pattern

Z

Entity A owns an item of property, plant and equipment that it uses to manufacture widgets. Entity A also owns a building that it rents out to tenants to earn rental income. Entity A accounts for the building as investment property in the scope of IAS 40 *Investment Property*, and uses the cost model. Entity A does not have specified main business activities.

<u>Analysis</u>

Depreciation on the property, plant and equipment must be classified in the operating category because the depreciation does not arise from (IFRS 18.53):

- (a) Investments in associates, joint ventures and unconsolidated subsidiaries;
- (b) Cash and cash equivalents; and
- (c) Other assets if they generate a return individually and largely independently of the entity's other resources.

IFRS 18.B48 and B49 also emphasises that income and expenses from assets that an entity uses in combination to produce or supply goods or services do not generate a return individually and largely independently of the entity's other resources, and are therefore classified in the operating category. For example, income and expenses relating to assets that arise from the production or supply of goods and services, such as trade receivables.

Depreciation on the investment property is classified in the investing category because it is 'income and expenses

that arise from the initial and subsequent measurement of the asset' (IFRS 18.54) and the investment property is an asset that generates a return individually and largely independently of the entity's other resources (IFRS 18.53(c)). The associated lease income would also be classified in the investing category because that lease income is 'income generated by the asset'.

The above analysis may be demonstrated by applying the classification practice aid as follows:

Property, plant and equipment depreciation



Example 3.2-2 - non-specified expenses related to specified assets

There may be instances where certain income and expenses relate to assets specified in IFRS 18.53, but because the income and/or expenses are not specified in IFRS 18.54, they must be classified in the operating category. This may produce a 'mismatch' between income and expenses related to a single asset.

For example, assume the same fact pattern as example 3.2.1., with additional case facts.

Fact pattern

P

Entity A owns an item of property, plant and equipment that it uses to manufacture widgets. Entity A also owns a building that it rents out to tenants to earn rental income. Entity A accounts for the building as investment property in the scope of IAS 40 Investment Property, and uses the cost model. Entity A does not have specified main business activities.

Entity A incurs cleaning and maintenance fees on the investment property.

<u>Analysis</u>

The rental income on the investment property is classified in the investing category for the reasons noted in Example 3.2-1. However, the cleaning and maintenance fees must be classified in the operating category, despite the fact that they are related to the investment property and are necessary in order to generate rental income. That is because they are not specified income and expenses in IFRS 18.54.

The cleaning and maintenance fees are not:

- (a) The income generated by the assets;
- (b) The income and expenses that arise from the initial and subsequent measurement of the assets, including on derecognition of the assets; or
- (c) The incremental expenses directly attributable to the acquisition and disposal of the assets

As the cleaning and maintenance fees do not meet the criteria in IFRS 18.53-54, they must be classified in the operating category.

The above analysis may be demonstrated by applying the classification practice aid as follows:



3.2.4. Investments in associates, joint ventures and unconsolidated subsidiaries

Associates and joint ventures are generally accounted for using the equity method in accordance with IAS 28 *Investments in Associates and Joint Ventures*. If the equity method is applied to such investments, then the investor's share of profit or loss recognised in accordance with IAS 28 will always be classified in the investing category, regardless of whether the entity has specified main business activities or not (IFRS 18.55(a); IFRS 18.BC131-BC132).

However, there may be cases in which an entity accounts for investments in associates, joint ventures and unconsolidated subsidiaries using a basis of measurement other than the equity method. For example, an investment entity accounts for such investments at fair value through profit or loss (see IFRS 10 *Consolidated Financial Statements*).

The basis under which an investor accounts for its associates, joint ventures and unconsolidated subsidiaries is not relevant unless the entity has a main business activity of investing in such assets. If the entity does <u>not</u> have such a main business activity, the resulting income and expense from such investments will <u>always</u> be classified in the investing category.

See section <u>3.8.1</u> for considerations for entities with specified main business activities. In certain cases, such entities may classify income and expenses related to investments in associates, joint ventures and unconsolidated subsidiaries not accounted for using the equity method in the operating category.

3.2.5. Cash and cash equivalents

Income and expense related to cash and cash equivalents (e.g. interest income) will always be classified in the investing category unless an entity has specified main business activities (IFRS 18.56).

The classification of cash and cash equivalents for entities that invest in financial assets or provide financing to customers as a main business activity is complex. Refer to section <u>3.9.1</u> for further information on these requirements and a flowchart that may be used in classifying income and expenses related to cash and cash equivalents for entities with specified main business activities.



3.2.6. Other assets that generate a return individually and largely independently of the entity's other resources

The third category of assets where specified income and expenses are classified in the investing category, is assets that generate a return individually and largely independently of the entity's other resources (IFRS 18.53(c)). This is the broadest of the three types of specified assets listed in IFRS 18.53, as this may include many different types of assets. These types of assets typically include (IFRS 18.B46):

- Debt or equity investments; and
- ▶ Investment properties, and receivables for rent generated by those properties.

It should be noted that the 'return' generated may be positive or negative; a negative return being generated does not preclude an asset from generating an individual and largely independent return (IFRS 18.B45).

Income and expenses that arise from such assets typically include (IFRS 18.B47):

- (a) Interest;
- (b) Dividends;
- (c) Rental income;
- (d) Depreciation;
- (e) Impairment losses and reversals of impairment losses;
- (f) Fair value gains and losses; and

(g) Income and expenses from the derecognition of the asset, or its classification and remeasurement as held for sale.

3.2.7. Assets that do not generate a return individually and largely independently of the entity's other resources

IFRS 18.B48 notes that assets an entity uses in combination to produce or supply goods or services do not generate a return individually and largely independently of the entity's other resources. Such assets typically include: (a) Property, plant and equipment;

(b) Assets that arise from the production or supply of goods and services for which the income and expenses are classified in the operating category (for example, receivables for such goods and services); and

(c) If the entity provides financing to customers as a main business activity, any loans to a customer.

BDO comment – interaction between IFRS 18.B48 and classification of assets

How an asset is classified in the statement of financial position will affect how an entity assesses whether that asset generates a return individually and largely independently of the entity's other resources. For example, depending on how an entity intends to use a building, it may be classified as:

- Inventory: if the entity regularly sells buildings to customers (e.g. a homebuilder);
- Property, plant and equipment: if the entity uses the building in its operations (e.g. head office or a distribution centre); or
- Investment property: if the entity holds the building for rental income or for capital appreciation or both (e.g. commercial real estate).

Inventory and property, plant and equipment will typically not generate a return individually and largely independently of the entity's other resources because the production and sale of those assets requires the entity to combine a variety of different resources in order to earn a return.

Investment property generates a return individually and largely independently of the entity's other resources because rental income and capital appreciation are earned independently of the entity's business activities. Therefore, income and expenses specified in IFRS 18.54 are classified in the investing category unless an entity invests in investment property as a main business activity (see section <u>3.8.1</u>).

Example 3.2-3 – classification of income and expenses relating to loans receivable

Assessing income and expenses related to assets that generate a return individually and largely independently of the entity's other resources will depend on the characteristics of the asset and the entity's operations.

Fact pattern

D

Entity H is a manufacturer and produces widgets. Entity H has loaned CU500 to a related party (Entity I) to provide that entity with financing. Entity H does not regularly make loans and issuing loans is not one of its main business activities. Entity H has recorded interest income on the loan.

Entity J is a financial institution and issues loans to borrowers as its main business activity. Entity J has loaned CU700 to a borrower to construct a new facility. Entity J has recorded interest income on the loan.

Assessment

Entities H and J both have to assess whether the loans receivable generates a return individually and largely independently of the entity's other resources. This is necessary because the entities have to determine how to classify the interest income in the statement of profit or loss.

Entity H determines that the loan receivable does generates a return individually and largely independently of the entity's other resources because the loan receivable is a debt instrument as noted in IFRS 18.B46. The interest income earned on the loan is income generated by the asset (IFRS 18.54(a)), therefore, Entity H classifies the interest income on the loan receivable in the investing category because Entity H's main business activities do not include issuing loans.

Entity J determines that the loan receivable does not generate a return individually and largely independently of the entity's other resources because Entity J provides financing to customers as a main business activity (IFRS 18.B48(c)). As Entity J invests in loans as a main business activity, income and expenses related to those assets are classified in the operating category.

This example demonstrates that two entities that own identical underlying asset may classify income and expenses relating to that asset differently depending on whether they invest in those assets as a main business activity.

See section 3.8.1 for further explanations of investing in assets as a main business activity.

3.2.8. Derecognition of an asset or liability and remeasurement of an asset held for sale

IFRS 18.B60 requires that an entity classify income and expenses on the derecognition of an asset, or its classification as held for sale (e.g. a write-down required by IFRS 5) and any subsequent measurement while held for sale, in the same category as it classified the income and expenses from the asset immediately before its derecognition.

IFRS 18.B47(g) requires that income and expenses from the derecognition of a specified asset, or its classification and remeasurement as held for sale be classified in the investing category.

This principle essentially 'backwards traces' the income and expenses arising from derecognition or classification as held for sale to the classification of income and expenses associated with that asset.

For example (IFRS 18.B60):

Gains and losses arising from the:	Classification (assuming no specified main business activities)
Disposal of property, plant and equipment	Operating category
Disposal of investment property	Investing category
Remeasurement of an associate previously accounted for using the equity method on the step acquisition of a subsidiary	Investing category

This same principle applies to the remeasurement of an asset when it is classified as held for sale in accordance with IFRS 5 *Non-current Assets Held for Sale and Discontinued Operations*. IFRS 5.15 requires that an entity measure a non-current asset (or disposal group) classified as held for sale at the lower of its carrying amount and fair value less costs to sell. This may result in gains or losses being recognised throughout the period when an asset (or disposal group) is classified as held for sale.

These gains and losses are classified in a similar manner to the table above. For example:

Gains and losses arising from the remeasurement of asset classified as held for sale (IFRS 5.15)	Classification (assuming no specified main business activities)
Property, plant and equipment; intangible assets	Operating category
Investment property	Investing category

Gains and losses arising from the measurement and remeasurement of an asset in accordance with IFRS 5 are not necessarily classified in the discontinued operations category, despite the fact that they arise from the application of the requirements of IFRS 5, however, this conclusion may differ for groups of assets where the criteria to present the gain or loss as a discontinued operation are met – see the following section on groups of assets. Only income and expense from discontinued operations in accordance with IFRS 5 are classified in the discontinued operation (see section 5.6).

Similar requirements exist for the derecognition of liabilities. See section 3.3.

These requirements apply to individual assets; however, a simplified classification approach exists for groups of assets and liabilities.

3.2.9. Derecognition of groups of assets and liabilities and remeasurement of an asset held for sale

An entity may dispose of groups of assets and liabilities in a single transaction, such as the sale of a group of related assets, or the sale of a subsidiary that holds assets and liabilities.

In certain cases, such a disposal may be classified as a discontinued operation in accordance with IFRS 5, and therefore, the resulting income and expenses (including a gain or loss on disposal) would be classified in the discontinued operations category (see section <u>3.6</u>). That is because IFRS 5.33(a)(ii) requires the post-tax gain or loss recognised on the measurement to fair value less costs to sell or on the disposal of the assets or disposal group(s) constituting the discontinued operation to be included in the 'single amount' presented as results of discontinued operations.

However, not all such disposals of groups of assets and liabilities will meet the criteria to be classified as discontinued operations. An entity may dispose of a group of assets and liabilities that do not meet the criteria to be presented as a discontinued operation in accordance with IFRS 5, and therefore, income and expenses cannot be classified in the discontinued operations category. This raises the question of how an entity classifies the gains and losses arising from such a sale. For example, whether an entity is required to disaggregate the total gain or loss and allocate it to the various assets and liabilities, classifying a portion of the total gain or loss in the operating category (e.g. the portion related to property, plant and equipment) and another portion in the investing category (e.g. the portion related to investment property).

IFRS 18.B63 simplifies the requirement by requiring that such a gain always be classified in the operating category unless <u>all the assets</u> in the group (other than income tax assets) generated income and expenses that the entity classified in the investing category immediately before the transaction or other event.

This is a simplification of the overall principles in IFRS 18, however, the IASB decided that requiring an entity to allocate the gain or loss on derecognition of a group of assets (or a group of assets and liabilities) to the categories related to those assets (or assets and liabilities) could result in arbitrary allocations and be costly for preparers to apply (IFRS 18.BC204).

3.2.10. Change in use of an asset

A change in use of an asset may change the assessment of whether the asset is an asset specified in IFRS 18.53, and therefore, that specified income and expenses shall be classified in the investing category. For example, an entity may use an item of property, plant and equipment as its head office, but then change the use of the asset by moving to another facility and renting out the previous head office as an investment property. The asset would not have met the criteria in IFRS 18.53 (i.e. it is not a specified asset) prior to the change in use, but it would upon reclassification as an investment property because the building would then generate a return individually and largely independently of the entity's other resources (IFRS 18.53(c)).

Income and expense up until the point of reclassification of the asset would be presented in accordance with the requirements of IFRS 18 as applied to that asset, with only income and expense from the date of reclassification being treated differently. For example, income and expenses related to the property plant and equipment would be classified in the operating category until the date of reclassification to investment property (e.g. depreciation on the building until the date of reclassification).

Only from that date of reclassification onwards would the income and expenses be classified in the investing category (assuming the entity does not invest in such assets as a main business activity).

IFRS 18.B62 requires an entity to classify the income and expenses from the transaction or other event that results in the reclassification in the category in which it classified income and expenses from the asset immediately before the transaction or event. For example, the transfer of an item of property, plant and equipment from the scope of IAS 16 to IAS 40 may result in amounts being recognised in profit or loss (IAS 40.62(a) and 62(b)(i)). These amounts must

be classified in the operating category because that is the category in which the entity classified income and expenses from the building immediately before the reclassification to investment property.

If a change in use applies to a group of assets as a result of a transaction or other event, then the requirements of IFRS 18.B63 discussed in the previous section apply. IFRS 18.B63 requires that such income or expenses always be classified in the operating category unless all the assets in the group (other than income tax assets) generated income and expenses that the entity classified in the investing category immediately before the transaction or other event.

3.2.11. Practice aid – classification of gains and losses on the derecognition of an asset or liability (or group of assets and liabilities), the remeasurement of an asset (or disposal group) as held for sale, or a change in use

This practice aid summarises which guidance an entity should apply in classifying income and expenses from the derecognition and/or remeasurement of assets (or groups of assets) while classified as held for sale or from a change in use.





3.2.12. Presentation of income and expenses classified in the investing category

IFRS 18 does not require a subtotal to be presented for all items classified in the investing category (e.g. 'investing profit or loss'). IFRS 18 requires 'profit or loss before financing and income taxes' to be presented (IFRS 18.69(b)), therefore, this subtotal will comprise the sum of income and expenses classified in the operating and investing categories. This relationship is depicted below:

Line item		Classification	Explanation
Revenue	XXX		
Cost of sales	XXX		
Gross profit	XXX		
Other operating income	XXX	Operating category	
Selling expenses	XXX		
Research and development	XXX		
General and administrative	XXX		
Operating profit	XXX	Mandatory specified subtotal	Sum of the operating category
Fair value gains on investments in equity instruments	XXX	Investing category	
Profit before financing and income taxes	XXX	³ <u>Mandatory</u> specified subtotal	Sum of the operating and investing categories
Interest expense on borrowings and lease liabilities	XXX	Financing category	
Profit before income taxes	XXX	Additional subtotal	
Income tax expense	XXX	Income taxes category	
Profit from continuing operations	XXX	Additional subtotal	
Loss from discontinued operations	XXX	Discontinued operations category	
Profit	XXX	Mandatory total	Sum of all categories

See section 5 for further information on subtotals and totals presented in the statement of profit or loss.



³ Profit before financing and income taxes is a mandatory subtotal for most entities, though an exception exists for entities that provide financing to customers as a main business activity and make a particular accounting policy choice – see section 5.1 'profit or loss before financing and income taxes – entities that provide financing to customers as a main business activity'.



3.3. Financing category

The classification requirements for the financing category focus on liabilities, whereas the investing category focuses on assets.

Classification of income and expenses in the financing category affects the determination of the mandatory subtotal 'profit or loss before financing and income taxes' (see section <u>5.1</u>). The financing category (and the related mandatory subtotal) is required by IFRS 18 because in the view of the IASB, many users of financial statements seek to analyse an entity's performance independently of how the entity is financed (IFRS 18.BC148). By isolating income and expenses related to financing in a separate category, and presenting a subtotal before those items, the operations of entities may be compared independently of how they are financed (e.g. debt vs. equity, the cost of debt financing, etc.).

3.3.1. General classification requirements

This section discusses the requirements applicable to entities that do not provide financing to customers as a main business activity. See section 3.8.2 for guidance applies to entities with such specified main business activities (e.g. providing financing to customers as a main business activity).

To classify income and expenses arising from liabilities, an entity is required to distinguish liabilities between (IFRS 18.59):

- (a) liabilities that arise from transactions that involve only the raising of finance; and
- (b) liabilities other than those described in (a) that is, liabilities that arise from transactions that do not involve only the raising of finance.

Separate requirements apply to income and expenses that relate to these two different types of liabilities.

3.3.2. Liabilities that arise from transactions that involve only the raising of finance

Liabilities that arise from transactions that involve only the raising of finance are where an entity (IFRS 18.B50):

- (a) receives finance in the form of cash, or an extinguishment of a financial liability, or receipt of the entity's own equity instruments; and
- (b) at a later date, will return in exchange cash or its own equity instruments.

For such liabilities, IFRS 18.60 requires entities to classify in the financing category the amounts included in the statement of profit or loss for:

- (a) income and expenses that arise from the initial and subsequent measurement of the liabilities, including on derecognition of the liabilities; and
- (b) the incremental expenses directly attributable to the issue and extinguishment of the liabilities for example, transaction costs.

Certain exceptions to the income and expenses listed in IFRS 18.60 exist, such as income and expenses related to hybrid contracts that contain a host that is a liability and gains and losses on derivatives and designated hedging instructions – see 'exceptions to classification principles' below as well as sections <u>3.9.2</u>, <u>3.9.3</u> and <u>3.9.4</u>.





Examples of such liabilities include, along with associated specified income and expenses in IFRS 18.60:

Liabilities that arise from transactions that involve only the raising of finance (IFRS 18.B51)	Specified income and expenses classified in the financing category (IFRS 18.B52)
A debt instrument that will be settled in cash, such as debentures, loans, notes, bonds and mortgages—an entity receives cash and will return cash in exchange	
A liability under a supplier finance arrangement when the payable for goods or services is derecognised—an entity is discharged of the financial liability for the goods or services and will return cash in exchange;	 Interest expenses (for example, on debt instrument issued); Fair value gains and losses (for example, on a liability designated at fair value through profit or loss);
A bond that will be settled through delivery of an entity's shares—an entity receives cash and will return its own equity instruments in exchange;	 Dividends on shares issued classified as liabilities; and Income and expenses from the modification and/or derecognition of the liability⁴.
An obligation for an entity to purchase its own equity instruments—an entity receives its own equity instruments and will return cash in exchange.	

As IFRS 18 requires only a binary assessment – whether a liability arises from a transaction that involves only the raising of finance or not, examples of liabilities that do not meet this criterion are included in the following section.

3.3.3. Liabilities that arise from transactions that do not involve only the raising of finance

For liabilities that arise from transactions that do not involve only the raising of finance, the income and expenses that are to be classified in the financing category are more limited than 'pure financing' transactions.

For liabilities that arise from transactions that do not involve only the raising of finance (for example, the items listed below), an entity classifies in the financing category (IFRS 18.61):

- (a) interest income and expenses, but only if the entity identifies such income and expenses for the purpose of applying other requirements in IFRS Accounting Standards; and
- (b) income and expenses arising from changes in interest rates, but only if the entity identifies such income and expenses for the purpose of applying other requirements in IFRS Accounting Standards.

Any other income and expenses not included in IFRS 18.61 must be classified in the operating category. As mentioned above see 'exceptions to the classification principles' below as well as sections <u>3.9.2</u>, <u>3.9.3</u> and <u>3.9.4</u> for special requirements applicable to hybrid contracts, foreign exchange differences and gains and losses on derivatives. Therefore, the practical distinction between liabilities that arise from transactions that do and do not involve only the raising of finance is that for the latter, a more restrictive number of income and expenses will be classified in the financing category.



⁴ If as part of a supplier finance arrangement an entity derecognises a payable to a supplier and recognises a liability under that arrangement, then any income and expenses arising from that derecognition are classified in the operating category (IFRS 18.B61(b)).



Liabilities that arise from transactions that do not involve only the raising of finance (IFRS 18.B53)	Rationale for classification (i.e. why the liability does not arise from transactions that involve only the raising of finance) (IFRS 18.B53)	Examples of income and expenses that will be classified in the financing category (IFRS 18.B54)	
Payables for goods or services that will be settled in cash (IFRS 9)	The entity receives goods or services, not finance in the form specified by IFRS 18.B50(a).	Interest expenses on payables arising from the purchase of goods or services, applying IFRS 9.	
Contract liabilities (IFRS 15)	The entity receives cash, but will settle the liability by delivering goods or services rather than cash or its own equity instruments as required by IFRS 18.B50(b).	Interest expenses on a contract liability with a significant financing component as specified by IFRS 15.	
Lease liabilities (IFRS 16)	The entity receives a right-of- use asset, not finance in the form specified by IFRS 18.B50(a).	Interest expenses on a lease liability, applying IFRS 16.	
Defined benefit pension liabilities (IAS 19)	The entity receives employee services, not finance in the form specified by IFRS 18.B50(a).	Net interest expense (income) on a net defined benefit liability (asset), applying IAS 19.	
Decommissioning or asset restoration provisions (IAS 37)	The entity receives an asset (the increase in the carrying amount of assets), not finance in the form specified by IFRS 18.B50(a).	The increase in the discounted amount of a provision arising from the passage of time and the effect of any change in the discount rate on	
A litigation provision (IAS 37)	The entity does not receive finance in the form specified by IFRS 18.B50(a).	provisions, applying IAS 37.	

BDO comment – interaction between IFRS 18 financing category requirements and IAS 23 Borrowing Costs

IFRS 18's requirements applicable to interest income and expenses indirectly interact with the requirements of IAS 23, which requires entities to capitalise borrowing costs into the carrying amount of assets if certain criteria are met. If inventories or items of property, plant and equipment take a substantial period of time to get ready for their intended use or sale (i.e. they are 'qualifying assets' in IAS 23), then borrowing costs must be capitalised. For example, the long-term construction of a building may commonly include borrowing costs being capitalised into the carrying amount of the building as it is constructed.

Borrowing costs capitalised as a component of the carrying amount of a qualifying asset may result from liabilities that do and do not arise from transactions that involve only the raising of finance. That is because IAS 23 defines borrowing costs to include components from both categories (e.g. interest expense calculated using the effective interest rate method as described in IFRS 9, which would be applicable to financial liabilities such as bank loans, as well as interest in respect of lease liabilities recognised in accordance with IFRS 16).

Once borrowing costs are capitalised in the carrying amount of a qualifying asset, the classification requirements of IFRS 18 applicable to interest income and expenses no longer apply. That is because they are no longer interest income and expenses recognised in profit or loss, they are 're-characterised' based on the nature of the asset.

For example, Entity H constructs a building which it intends to occupy as its head office, therefore, the building is in the scope of IAS 16. Over the period of construction, Entity H recognised borrowing costs as part of the carrying amount of the building as required by IAS 23. Once construction is complete, the carrying amount comprises the following amounts:

Components of carrying amount	Amount
Materials and labour	CU10,000
Capitalised borrowing costs relating to bank loan – specific borrowings (IAS 23.12)	CU800
Capitalised borrowing costs relating to lease liabilities – general borrowings (IAS 23.14)	<u>CU400</u>
Total	CU11,200

As Entity H depreciates the building over its useful life, Entity H does not apply the requirements in IFRS 18.59-61 to classify the component of the depreciation expense that relates to capitalised borrowing costs (CU800 + CU400 = CU1,200) in the financing category. The entire amount of depreciation expense is reflected in the operating category (IFRS 18.B49(c)) because once those amounts are capitalised into the borrowing cost of an asset, the eventual recognition of those amounts in profit or loss are no longer characterised as being interest income and expense recognised in profit or loss; they are depreciation of an item of property, plant and equipment.

Therefore, Entity H would reflect a portion of interest income and expenses in profit or loss (those expenses not capitalised in accordance with IAS 23), with those expenses being subject to the classification requirements of IFRS 18 applicable to the financing category. The portion of interest income and expenses capitalised in accordance with IAS 23 would be subject to the classification requirements applicable to the income or expense recognised when the qualifying asset is eventually recognised in profit or loss (e.g. via depreciation, impairment, etc.).

BDO comment – income and expenses relating to liabilities that arise from transactions that do not involve only the raising of finance that will not be classified in the financing category (examples where IFRS 18.61 criteria are not met)

IFRS 18.B53 is only a list of examples, and does not consider every liability that may be recognised as a result of applying IFRS Accounting Standards. Income and expenses relating to liabilities that arise from transactions that do not involve only the raising of finance must be analysed to determine if they meet the criteria in IFRS 18.61 to be classified in the financing category.

Following are several examples of liabilities that arise from transactions that do not involve only the raising of finance and related income and expenses that do not meet the criteria in IFRS 18.61 to be classified in the financing category. The following examples are non-exhaustive.

Cash-settled share-based payment liabilities (IFRS 2)

An entity may recognise a liability for a cash-settled share-based payment transaction that it incurs to receive a service (e.g. cleaning services). The services received would be classified in the operating category, however, the cash-settled share-based payment liability must be remeasured at each reporting period to fair value with changes in fair value recognised in profit or loss (IFRS 2.30). The movement in fair value from one period to the next would consider two elements:

1. The passage of time (i.e. the unwinding of the discount recognised); and

2. Any changes in estimate relating to the liability (e.g. an increase in the estimated liability due to changes in the share price of the entity).

For liabilities that arise from transactions that do not involve only the raising of finance, IFRS 18.61 permits only two types of income and expenses to be classified in the financing category (**emphasis added**):

- (a) interest income and expenses, <u>but only if the entity identifies such income and expenses</u> for the purpose of applying other requirements in IFRS Accounting Standards; and
- (b) income and expenses arising from changes in interest rates, **<u>but only if the entity identifies such income and</u>** <u>**expenses**</u> for the purpose of applying other requirements in IFRS Accounting Standards.

IFRS 2.30 requires the cash-settled share-based payment liability to be remeasured from one period to the next, however, IFRS 2 does not require the entity to identify interest income and expenses in performing this remeasurement. Said another way, IFRS 2 requires fair value measurement for such liabilities at each reporting date, with the total movement in that liability being recognised in profit or loss, with no disaggregation between the unwinding of the discount and other changes in fair value.



Therefore, in our view, the remeasurement of a cash-settled share-based payment liability must be classified in the operating category.

Modification gains and losses on contract liabilities (IFRS 15)

IFRS 15.21(b) requires an entity to account for a contract modification as if it were a part of the existing contract if the remaining goods or services are not distinct and, therefore, form part of a single performance obligation that is partially satisfied at the date of the contract modification. The effect that the modification has on the transaction price and the entity's measure of progress towards complete satisfaction of the performance obligation is recognised as an adjustment to revenue.

The effect recognised in profit or loss must be classified in the operating category because such a gain or loss is not interest income and expenses (IFRS 18.61(a)) or income and expenses arising from changes in interest rates (IFRS 18.61(b)). Additionally, because IFRS 15.21(b) requires the effect to be reflected as an adjustment to revenue, the effect must be reflected in the operating category because revenue is also reflected in the operating category (IFRS 18.849(a)).

Modification gains and losses on lease liabilities (IFRS 16)

IFRS 16.46 requires a lessee to account for a lease modification that is not accounted for as a separate lease by decreasing the carrying amount of the right-of-use asset, with an amount recognised in profit or loss relating to the partial or full termination of the lease (see section 5.7 of <u>IFRS Accounting Standards In Practice IFRS 16 Leases</u> for a worked example).

This gain or loss recognised in profit or loss must be classified in the operating category because such a gain or loss is not interest income and expenses (IFRS 18.61(a)) or income and expenses arising from changes in interest rates (IFRS 18.61(b)).

Change in estimate relating to a legal provision (IAS 37)

IAS 37.45-47 requires that a provision reflect the time value of money when the effect is material (i.e. provisions must be discounted). IAS 37.59 requires that a provision be reviewed at the end of each reporting period and adjusted to reflect the current best estimate. Therefore, from one reporting period to the next, the change in the carrying amount of a provision that is discounted will reflect:

- ▶ The unwinding of the discount due to the passage of time; and
- Any updates to the current best estimate.

While IAS 37 does not specify the presentation of these two items in the statement of profit or loss (IAS 37.60 refers to the unwinding of the discount as a borrowing cost), and diversity in practice exists, these two items will generally be reflected in different categories in the statement of profit or loss.

For example, Entity B recognises a provision of CU100 as at 31 December 20x7 for a legal settlement. The liability is discounted to reflect the time value of money, as settlement is not expected for several years. As at 31 December 20x8, Entity B has revised its best estimate for the legal settlement, and the carrying amount of the provision is now CU125. Assume this total CU25 change in carrying amount is comprised of:

- ▶ CU8 reflecting the unwinding of the discount as time passes; and
- ▶ CU17 reflecting a change in the best estimate of the legal settlement.

CU8 is classified in the financing category because the legal provision is a liability that arise from transactions that do not involve only the raising of finance and it is interest income and expenses (IFRS 18.61(a)). IFRS 18.B54(e) specifies that the increase in the discounted amount of a provision arising from the passage of time and the effect of any change in the discount rate is an example of interest income and expenses that meet the criteria in IFRS 18.61 to be classified in the financing category.

CU17 is classified in the operating category because the legal provision is a liability that arise from transactions that do not involve only the raising of finance and it is not interest income and expenses (IFRS 18.61(a)) or income and expenses arising from changes in interest rates (IFRS 18.61(b)).

Therefore, updating the measurement of a liability may result in an entity needing to 'split' the total effect recorded in profit or loss between more than one category.

3.3.4. Income and expenses recognised on the derecognition of liabilities

The general classification requirements of IFRS 18 apply in classifying income and expenses recognised on the derecognition of a liability (IFRS 18.B61). Said another way, an entity must identify whether a derecognised liability arises from transactions that involve only the raising of finance, or not. with The income and expenses recognised on derecognition will being classified accordingly.

For example, assuming an entity does not provide financing to customers as a main business activity (IFRS 18.B61):

Income or expense recognised on derecognition of liability	Classification
A gain or loss is recognised on the early settlement of a bank loan	Financing category - the bank loan arises from a transaction that involves only the raising of finance (IFRS 18.60) and the gain or loss on derecognition arises from the derecognition of the liability (IFRS 18.60(a)).
A gain or loss is recognised when a payable to a supplier is derecognised as a result of a supplier financing arrangement with the recognition of a new liability	Operating category – the payable arises from a transaction that does not involve only the raising of finance (IFRS 18.61) and the gain or loss is not an item specified in IFRS 18.61(a)-(b), therefore, it must be classified in the operating category.



3.3.5. Practice aid – criteria to classify income and expenses in the financing category for entities without specified main business activities

This practice aid summarises the requirements to determine whether income or expenses are classified in the financing category, assuming that an entity does not have a main business activity of providing financing to customers. As discussed in section <u>3.8.2</u>, the requirements for classifying income and expenses for entities with specified main business activities are significantly more complex in such cases.





3.3.6. Exceptions to classification principles

The classification principles applicable to liabilities are set out above, however, there are a number of exceptions:

Exception	Reference
Hybrid contracts that contain a host that is a liability	Section <u>3.9.2.</u>
Foreign exchange differences	Section <u>3.9.3.</u>
Gains and losses on derivatives and designated hedging instruments	Section <u>3.9.4.</u>
Income and expenses from issued investment contracts with participation features recognised applying IFRS 9	See below.
Insurance finance income and expenses included in the statement of profit or loss applying IFRS 17	See below and section 7.5 for further information on the presentation requirements of IFRS 18 applicable to entities that issue insurance contracts in the scope of IFRS 17.

An entity always includes the following items in the operating category (IFRS 18.64):

- ▶ Income and expenses from issued investment contracts with participation features recognised applying IFRS 9; and
- ▶ Insurance finance income and expenses included in the statement of profit or loss applying IFRS 17.

BDO comment – insurance finance income and expenses

The requirement to classify insurance finance income and expense in the operating category applies even if an entity is not an 'insurance company' in the traditional sense. For example, a corporate entity may issue contracts that are permitted or required to be classified as insurance contracts in the scope of IFRS 17. Even if the issuance of such contracts is incidental to the entity's operations and the entity is not an 'insurance company' (i.e. it is not considered as such by law and regulation), IFRS 18.64 still requires the insurance finance income and expense to be classified in the operating category (IFRS 18.BC196-BC197) along with other income and expenses relating to the contracts (e.g. insurance revenue, insurance service result, etc.).

3.3.7. Presentation of income and expenses classified in the financing category

Prior to IFRS 18, IAS 1.82(b) required entities to present 'finance costs' as a separate line item in the statement of profit or loss. Finance costs were not defined in IAS 1, and entities applied a variety of accounting policies to determine which expenses were included in that line item. These may have included interest income and expenses, foreign exchange gains and losses, fair value gains and losses on derivatives used to manage risks associated with financing such as interest rate swaps, etc.

IFRS 18 did not bring this requirement forward from IAS 1, therefore, IFRS 18 contains no explicit requirement to present a single line item for finance costs. IFRS 18 also has no explicit requirement to present a subtotal for all items classified in the financing category (see section 5 for further information on subtotals).

Instead, entities are required to:

- Classify income and expenses in the appropriate categories, which may result in certain expenses being classified in the financing category (e.g. interest expense on bank loans if the entity does not provide financing to customers as a main business activity);
- 2. Apply the aggregation and disaggregation requirements to determine appropriate groupings of items (see section <u>4</u>); and
- 3. Consider whether disaggregation of the line items presented in the statement of profit or loss is necessary in the notes (see section 4).

The application of these requirements may change how an entity presents income and expenses related to financing compared to practice under IAS 1.

3.4.Operating category

The operating category is the residual category in IFRS 18, meaning that unless income and expenses are classified in another category, they must be classified in the operating category. Applying the requirements of IFRS 18 should result in an entity classifying in the operating category income and expenses from its main business activities (IFRS 18.B42), except for any such income and expenses from investments accounted for using the equity method, which are always classified in the investing category (see section <u>3.2</u>).

For entities with specified main business activities (see section 3.8), the general requirements applicable to classifying income and expenses in the investing and financing categories are modified. That is because if an entity, for example, invests in assets as a main business activity (e.g. investing in real estate such as investment property), then specified income and expenses are presented in operating rather than investing (e.g. the fair value gains or losses on investment property when the fair value model is applied) to better reflect the entity's main business activities.

As the operating category is defined as a residual (i.e. it must be used when other categories do not apply), IFRS 18 does not contain specific requirements about criteria that must be met in order for income and expenses to be classified in the operating category. The following table is a non-exhaustive summary of income and expenses that are generally expected to be classified in the operating category, assuming that an entity does not have specified main business activities (see section <u>3.8</u>):

Income and expenses	Examples of income and expenses classified in the operating category (assuming an entity does not have specified main business activities)
Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48) See section <u>3.2</u> .	 Revenue from the sale of goods and services, including: Revenue from contracts with customers (IFRS 15) Insurance revenue (IFRS 17) Depreciation, impairment and gains and losses on disposal of property, plant and equipment Expenses arising from inventories (e.g. expenses when inventories are derecognised such as 'cost of sales' and similar items, net realisable value write-downs) Expenses related to trade receivables and contract assets (e.g. expected credit losses applying IFRS 9)
 Income and expenses related to liabilities that arise from transactions that do not involve only the raising of finance and that are not (IFRS 18.61): Interest income and expenses; or Income and expenses arising from changes in interest rates. 	 Expenses related to the consumption of goods and services for which a liability is recognised (e.g. many executory contracts such as the acquisition of cleaning services, catering, repairs and maintenance not recognised as an asset, etc.) Remeasurement of cash-settled share-based payment liabilities (IFRS 2) Contract modifications gains and losses (IFRS 15) Modification gains and losses on lease liabilities (IFRS 16) Changes in the best estimate required to settle a provision (IAS 37)
See section <u>3.3</u> . Other items required to be classified in the operating category	 Income and expenses from issued investment contracts with participation features recognised applying IFRS 9 (see section <u>3.3</u>) Insurance finance income and expenses included in the statement of profit or loss applying IFRS 17 (see section <u>3.3</u>) Income and expenses related to business combinations, such as bargain purchase gains, income and expenses relating to the remeasurement of contingent consideration (IFRS 18.B49(f), BC144) (see section <u>3.9.5</u>)

Income and expenses in the operating category include all income and expenses that are not included in other categories. Income and expenses cannot be excluded from the operating category on account of the income and expenses being volatile, unusual or non-recurring (IFRS 18.B42).

Example 3.4-1 – classification of 'extraordinary items'

IAS 1.87 specifically forbade entities from presenting line items labelled as 'extraordinary items' in the statement of profit or loss. IFRS 18 does not contain any explicit requirements concerning extraordinary items, however, entities would apply the general requirements of IFRS 18 applicable to classify income and expenses that an entity may consider 'extraordinary'.

Fact pattern

P

Entity G has significant manufacturing facilities that are destroyed by a tornado, resulting in material property, plant and equipment and inventories being derecognised as a result of their destruction.

Analysis

Entity G applies IFRS 18 to determine how to classify these expenses relating to the derecognition of property, plant and equipment and inventories. These assets do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48), and therefore, income and expenses relating to their derecognition are classified in the operating category (IFRS 18.B60).

See section $\underline{4}$ for further information on how an entity aggregates and disaggregates information in the profit or loss, which may affect how Entity G presents and discloses the expenses relating to the tornado.

3.5. Income taxes category

Income and expenses are classified in the income taxes category if they are (IFRS 18.67):

- 1. Within the scope of IAS 12 Income Taxes; or
- 2. Foreign exchange differences related to income and expenses within the scope of IAS 12 (see section 3.9.3 for further information on the classification of foreign exchange differences).

BDO comment – the scope of IAS 12

In many cases, identifying income and expenses within the scope of IAS 12 to be classified in the income taxes category will be straightforward. For example, current and deferred tax expenses and recoveries recognised in accordance with IAS 12 are generally recognised in profit or loss, with certain exceptions (e.g. the tax effect of gains and losses presented in other comprehensive income).

In our view, the income taxes category also includes the write-down and any recovery of deferred tax assets recognised in profit or loss applying the recoverability requirements of IAS 12. This will also include the effect of uncertain tax treatments reflected in profit or loss within the scope of IFRIC 23 *Uncertainty over Income Tax Treatments*, because IFRIC 23 interprets how to measure items in the scope of IAS 12.

However, in some cases, determining whether income and expenses are within the scope of IAS 12 will be more challenging. For example, whether penalties and interest on income taxes are within the scope of IAS 12 or not.

The IFRS Interpretations Committee (the Committee) published an agenda decision in September 2017 – *Interest and penalties related to income taxes*. The Committee observed that entities do not have an accounting policy choice between applying IAS 12 and applying IAS 37 *Provisions, Contingent Liabilities and Contingent Assets* to interest and penalties. Instead, if an entity considers a particular amount payable or receivable for interest and penalties to be an income tax, then the entity applies IAS 12 to that amount. If an entity does not apply IAS 12 to a particular amount payable or receivable for interest and penalties, it applies IAS 37 to that amount.

The agenda decision also referenced a March 2006 decision – *Scope* (IAS 12), which noted that IAS 12 applies to income taxes, which are defined as taxes that are based on taxable profit. That implies that (i) not all taxes are within the scope of IAS 12 but (ii) because taxable profit is not the same as accounting profit, taxes do not need to be based on a figure that is exactly accounting profit to be within the scope. The latter point is also implied by the requirement in IAS 12 to disclose an explanation of the relationship between tax expense and accounting profit. The Committee further noted that the term 'taxable profit' implies a notion of a net rather than gross amount.

Applying the March 2006 and September 2017 agenda decisions means that an entity needs to determine whether interest and penalties related to income taxes are within the scope of IAS 12 (and are therefore classified in the income taxes category of the statement of profit or loss). If interest and penalties are not within the scope of IAS 12, then they must be classified in the operating category.

IFRS Accounting Standards are not clear about how this assessment should be performed, however, the basis under which interest and penalties are calculated (i.e. are they based on taxable profit, such as a percentage of taxable profit) may assist entities in making this determination.

3.6. Discontinued operations category

Income and expenses are classified in the discontinued operations category if they are income and expenses from discontinued operations as required by IFRS 5 (IFRS 18.68).

IFRS 5.33(a) requires the following total to be presented as a single line item in the statement of profit or loss: (i) the post-tax profit or loss of discontinued operations; and

(ii) the post-tax gain or loss recognised on the measurement to fair value less costs to sell or on the disposal of the assets or disposal group(s) constituting the discontinued operation.

This single line item specified by IFRS 5.33(a) must be classified in the discontinued operations category in accordance with IFRS 18.68.

BDO comment - income and expense that will not be included in the 'single line item' - IFRS 5.33(a)

This publication does not summarise all of the requirements of IFRS 5, however, a discontinued operation in defined in IFRS 5 as:

A component of an entity that either has been disposed of or is classified as held for sale and:

(a) represents a separate major line of business or geographical area of operations;

(b) is part of a single co-ordinated plan to dispose of a separate major line of business or geographical area of operations; or

(c) is a subsidiary acquired exclusively with a view to resale.

A 'component of an entity' is defined as 'operations and cash flows that can be clearly distinguished, operationally and for financial reporting purposes, from the rest of the entity.'

Therefore, these criteria must be met for an entity to present the results of a discontinued operation in accordance with IFRS 5.33(a), and consequently, in the discontinued operations category in accordance with IFRS 18.68.

IFRS 5.33(a)(ii) also requires the post-tax gain or loss recognised on the measurement to fair value less costs to sell or on the disposal of the assets or disposal group(s) constituting the discontinued operation to be included as part of the single line item. This measurement to the lower of carrying amount and fair value less costs to sell is required by IFRS 5.15, however, not all write-downs as a result of applying IFRS 5.15 will be included in the single line item required by IFRS 5.33(a)(ii).

For example, an entity may plan to dispose of an item of property, plant and equipment, such as a distribution centre, and classify it as an asset held for sale. IFRS 5.15 requires that asset to be measured at the lower of its cost and fair value less costs to sell. However, an impairment loss recognised as a result of applying IFRS 5.20 in this fact pattern would not be presented as a discontinued operation and would be excluded from the single line item to be reported in accordance with IFRS 5.33(a)(ii). That is because IFRS 5.33(a)(ii) includes only the post-tax gain or loss recognised relating to <u>discontinued operations</u>, meaning the definition of a discontinued operation must be met.

Similarly, the disposal of a single item of property, plant and equipment (such as a distribution centre in this example) will not meet the definition of a discontinued operation, therefore, the write-down recognised in accordance with IFRS 5.15 is not included in the single line item required by IFRS 5.33(a)(ii), and is not classified in the discontinued operations category. Instead, the entity applies the requirements applicable to the remeasurement of an asset (or disposal group) that is not a discontinued operation. See section 3.2.3 – practice aid

- classification of gains and losses on the derecognition of an asset or liability (or group of assets and liabilities), the remeasurement of an asset (or disposal group) held for sale or a change in use.

3.7. Entities with specified main business activities - an entity's assessment of specified main business activities

3.7.1. Purpose of specified main business activities requirements

The general classification requirements of IFRS 18 (as explained in sections 3.2 - 3.6) apply when an entity does not have specified main business activities, which are instances when the reporting entity:

- Invests in assets that generate a return individually and largely independently of the entity's other resources as a main business activity (IFRS 18.55-58); and/or
- ▶ Provides financing to customers as a main business activity (IFRS 18.65-66).

If an entity has one or more of these activities as a main business activity, then special considerations apply, which will result in certain income and expenses being classified in the operating category rather than the investing and financing categories. In some cases, entities may also have accounting policy choices relating to the classification of certain income and expenses (see sections 3.9.1 and 3.9.4).

The IASB decided to require certain income and expenses to be classified in the operating category rather than the investing and/or financing categories for entities with specified main business activities because otherwise, operating profit would not include all of the income and expenses related to an entity's main business activities (IFRS 18.BC89(a)). Additionally, classifying certain income and expenses in the investing and financing categories would prevent certain entities from presenting key measures of operating performance in the operating category (IFRS 18.BC90).

For example, without the specified main business activities requirements, an entity that sells goods and services but also provides financing to customers as a main business activity (e.g. an entity that sells farm equipment and financing for each item of equipment sold) would not be permitted to present interest income and expenses in the operating category. If the entity provides financing to customers as a main business activity, the entity may use 'net financial margin' as a key measure of profits it generates from providing financing to customers. If the entity were unable to present finance income and expenses in the operating category, this would mean that the operating category would not include all of the entity's main business activities.

3.7.2. Level of assessment

An entity assesses whether it has specified main business activities at the reporting entity level (IFRS 18.B37, BC98). Therefore, a reporting entity that consists of its parent and consolidated subsidiaries performs this assessment at the group level. See section <u>3.7.7</u> 'considerations applicable to groups' below for further specific considerations for groups.

If an entity prepares separate financial statements, then the reporting entity would be the legal entity only, as any subsidiaries would not be consolidated. For an entity preparing separate financial statements, it would assess whether it invests in assets as a main business activity using groups of assets that are consistent with the categories used to determine their measurement basis applying IAS 27.10 (IFRS 18.B38). See section <u>3.7.5</u> for further considerations relating to whether investment in unconsolidated subsidiaries are a main business activity.

When assessing if an entity invests in assets as a main business activity, this assessment is performed by assessing an individual asset or group of assets with shared characteristics. When making this assessment for financial assets, an entity is required to use groups of financial assets that are consistent with the classes of financial assets identified applying IFRS 7.6 (IFRS 18.B40).

BDO comment – grouping of financial assets for assessment of whether an entity invests in assets as a main business activity

IFRS 7.6 requires an entity to prepare disclosures by class of financial instrument, with an entity grouping financial instruments into classes that are appropriate to the nature of the information disclosed and that take into account the characteristics of those financial instruments.

Therefore, an entity may, for example, group financial assets into classes such as equity instruments vs. debt instruments. When assessing if it invests in such assets as a main business activity, it is possible for an entity to reach different conclusions for these classes of assets. For example, the entity may conclude that it invests in debt instruments as a main business activity, but not equity instruments.

3.7.3. Multiple main business activities

An entity may have more than one main business activity (IFRS 18.B30), and it is possible to have more than one specified main business activity. An entity also assesses whether it has specified main business activities independently for the two types of specified main business activities:

- Invests in assets that generate a return individually and largely independently of the entity's other resources as a main business activity (IFRS 18.55-58); and/or
- ▶ Provides financing to customers as a main business activity (IFRS 18.65-66).

For example, an entity may conclude that it:

- 1. Invests in assets as a main business activity (e.g. real estate), but it does not provide financing to customers as a main business activity;
- 2. Does not invest in assets as a main business activity, but it provides financing to customers as a main business activity (e.g. selling goods with associated financing); or
- 3. Invests in assets as a main business activity (e.g. financial assets) and also provides financing to customers as a main business activity (e.g. loans provided to borrowers by a bank).

An entity may also conclude that it invests in multiple types of assets as a main business activity. For example, an entity may conclude that it invests in both real estate and financial assets as a main business activity, which would result in many items of income and expense being classified in the operating category (e.g. fair value gains and losses on investment property, fair value gains and losses on financial assets, etc.).

3.7.4. Evidence to be considered when making assessment

The assessment of whether an entity has specified main business activities is a matter of fact and not an assertion or an accounting policy choice (IFRS 18.B33, BC96). It is not optional for an entity to assess whether it has specified main business activities, and if it does, it is required to apply the requirements for entities with specified main business activities. Assessing whether an entity has specified main business activities will require judgement.

IFRS 18 notes two sources of evidence that an entity may use in assessing whether an entity has specified main business activities.

Use of subtotals as an indicator of operating performance, including indicators used outside of financial statements:

- If an entity uses a particular type of subtotal as an important indicator of operating performance, then investing in assets or providing financing to customers is likely to be a specified main business activity (IFRS 18.B34).
- These subtotals would be similar to gross profit (see section 5.2), which includes income and expenses that would be classified in the investing or financing categories if the entity did not have specified main business activities.
- Evidence that subtotals similar to gross profit are important indicators of operating performance includes using subtotals to (IFRS 18.B35):

(a) Explain operating performance externally; or

(b) Assess or monitor operating performance internally.

BDO comment - use of subtotals as an indicator of operating performance

IFRS 18.B34-B35 does not include examples of subtotals which if used as an important indicator of operating performance may indicate that an entity invests in assets or provides financing to customers as a main business activity.

However, subtotals similar to gross profit are common in certain industries. For example, banking and lending institutions commonly use a subtotal of interest income less interest expense, which may be labelled as 'net financial margin' or 'net interest income'. Use of a subtotal in external and/or internal communications may indicate that an entity provides financing to customers as a main business activity.

IFRS 18.B34 refers to IFRS 18.B123 when it refers to subtotals similar to gross profit which if used, may indicate that an entity invests in assets or provides financing to customers as a main business activity. For example, IFRS 18.B123(a) notes 'net interest income' as a subtotal similar to gross profit or loss. An entity using net interest income to communicate information about financial performance may provide evidence that the entity has specified main business activities (e.g. providing financing to customers).

IFRS 8 Operating Segments (IFRS 18.B36):

- Information about segments may provide evidence that investing in assets or providing finance to customers is a main business activity if an entity applies IFRS 8, however, IFRS 8 is only mandatory for certain entities (IFRS 8.2), therefore, segment information may not provide evidence of main business activities for all entities that apply IFRS Accounting Standards..
- An entity's operating segments and reportable segments may provide evidence of whether the entity has specified main business activities. This assessment may vary depending on whether the operating segment is reportable or not (IFRS 18.B36):
 - (a) Reportable segment: if a reportable segment comprises a single business activity, this indicates that the performance of the reportable segment is an important indicator of the entity's operating performance and that the business activity of the reportable segment is a main business activity of the entity.
 - (b) Operating segment: if an operating segment comprises a single business activity, this indicates that the business activity might be a main business activity of the entity if the performance of the operating segment is an important indicator of the entity's operating performance.



BDO comment – reportable segments and operating segments

Reportable segments

If a reportable segment comprises a single business activity, then it is logical that that single business activity is an important indicator of the entity's performance. That is because a reportable segment is an operating segment that is identified according to the requirements of IFRS 8. Identification as a reportable segment in accordance with IFRS 8 means that the operating segment is (1) a component of an entity whose operating results are regularly reviewed by the entity's chief operating decision maker (IFRS 8.5(b)); and/or (2) meets quantitative thresholds in IFRS 8.13.

IFRS 18.B36(a) deems an entity with a reportable segment that comprises a <u>single business activity</u> as having as a main business activity, however, not all reportable segments comprise a single business activity, therefore, care must be taken in applying IFRS 18.B36(a) only to certain reportable segments.
Operating segments

IFRS 18.36(b) is less definitive for operating segments that are not reportable segments, that is, components of an entity that meet the definition of an operating segment but are not deemed to be reportable applying the requirements of IFRS 8. If a non-reportable segment comprises a single business activity, it is still possible that that operating segment's performance is an important indicator of the entity's performance, and therefore, the business activity of that segment may be a main business activity of the entity overall, but IFRS 18.36(b) does not deem this to be the case as it does for reportable segments.

The assessment of a non-reportable operating segment's main business activities may therefore be part of the evidence an entity uses to assess whether the entity has specified main business activities, but it is not definitive.

3.7.5. Investing in associates, joint ventures and unconsolidated subsidiaries

It is not necessary for an entity to assess whether it invests in associates, joint ventures and unconsolidated subsidiaries as a main business activity if the entity accounts for such investments using the equity method. That is because if the equity method is applied to such investments, then income and expenses relating to such investments will always be classified in the investing category, regardless of whether an entity has specified main business activities (IFRS 18.B38; see section 3.2).

If an entity does not apply the equity method to such investments, then it must assess whether it invests in associates, joint ventures and unconsolidated subsidiaries as a main business activity. For example, an investment entity as defined in IFRS 10 may invest in associates and joint ventures together with subsidiaries as a main business activity, or a venture capital organisation may also invest in associates and joint ventures as a main business activity and not apply the equity method if certain accounting policies are elected (IAS 28.18).

In making an assessment of whether it invests in assets as a main business activity, an entity performs this assessment by individual asset or using groups of assets with shared characteristics (IFRS 18.B38). Therefore, it is possible for an entity to conclude that it invests in certain associates, joint ventures and unconsolidated subsidiaries as a main business activity, but not others.

3.7.6. Examples of entities with specified main business activities

Examples of entities that might invest in assets as a main business activity include (IFRS 18.B31):

- (a) investment entities as defined by IFRS 10;
- (b) investment property companies; and

(c) insurers.

Examples of entities that might provide financing to customers as a main business activity include (IFRS 18.B32): (a) banks and other lending institutions;

(b) entities that provide financing to customers to enable those customers to buy the entity's products; and (c) lessors that provide financing to customers in finance leases.

BDO comment – examples of entities with specified main business activities

The examples given in IFRS 18.B31 and B32 are non-exhaustive, and there may be many other types of entities with specified main business activities.

An example of an entity that provides financing to customers to enable those customers to buy the entity's products would include entities that provide significant financing terms for the sale of goods. For example, a company that sells heavy machinery may lease equipment to customers under a finance lease (IFRS 18.B32(c)) or legally transfer title to the equipment with a corresponding loan agreement to pay the selling entity over time through instalments of principal and interest.



As noted above, the assessment of whether an entity has specified main business activities is made at the reporting entity level, being the entity and its subsidiaries as one unit of account. This assessment may differ, for example, at the consolidated level of the parent entity compared to a subsidiary, or intermediate parent entities' individual financial statements.

This may result in significant consolidation adjustments when a subsidiary is consolidated and its parent's assessment of whether it has specified main business activities differs from its own assessment. This may occur due to materiality. For example, a subsidiary may clearly invest in assets as a main business activity at the level of its own financial statements, but when this assessment is made at the consolidated level of its parent, taken together with other subsidiaries, the operations of that subsidiary may not be significant, resulting in a different assessment.

Example 3.7-1 – consolidation adjustments required for subsidiary upon consolidation

Fact pattern

Z

Entity J is a holding company with no substantive operations. Entity J has three subsidiaries:

- 1. Entity K: a manufacturer of automotive parts
- 2. Entity L: a private lender providing loans to customers
- 3. Entity M: a real estate company that invests in commercial real estate for rental income and capital growth and measures the investment properties at fair value

Entity J prepares consolidated financial statements, and Entities K, L and M all prepare their own individual financial statements as well.

Entities K, L and M make the following assessments of whether they have specified main business activities in their individual financial statements:

Entity	Specified main business activities?
Entity K	No
Entity L	Yes; providing financing to customers and investing in assets (loans receivable from customers)
Entity M	Yes; investing in assets (real estate, being investment property measured at fair value – IAS 40)

At the consolidated level, Entity J assesses that it has no specified main business activities. That is because the operations of Entity K are significantly larger than Entities L and M. Entities L and M are not reportable segments in accordance with IFRS 8 and at the level of Entity J, subtotals similar to gross profit are not used as an important indicator of operating performance.

Assessment

Entities L and M are required to apply the requirements applicable to entities with specified main business activities in their individual financial statements. For example:

- Entity L: interest income and expenses are classified in the operating category rather than the investing and financing categories (IFRS 18.65(a)(i), 58).
- Entity M: rental income and fair value gains and losses on investment property are classified in the operating category rather than the investing category (IFRS 18.58).

In Entity J's consolidated financial statements, the requirements applicable to entities with specified main business activities do not apply, therefore, when consolidating Entities L and M, Entity J makes the following consolidation adjustments:

- Entity L: reclassifies interest income and expenses from the operating category to the investing and financing categories.
- Entity M: reclassifies rental income and fair value gains and losses on investment property from the operating category to the investing category.

This example is straightforward, however, for entities with complex structures and many subsidiaries, making appropriate consolidation adjustments for assessments of specified main business activities may be complex and require consolidation procedures to be adjusted (e.g. adjustments to consolidation software to allow for income and expenses to be classified into different categories at different levels of consolidation).

3.7.8. Change in assessment

An entity's assessment of whether it has specified main business activities is not a 'one time' assessment, because it is possible for an entity's assessment to change as its operations change over time.

IFRS 18 requires any change in an entity's assessment of whether it has specified main business activities to be reflected prospectively (IFRS 18.B41), therefore, comparatives are not restated.

IFRS 18.51(c)(i) requires an entity to disclose the fact the outcome of the assessment has changed and the date of the change.

IFRS 18.51(c)(ii) requires an entity to disclose the amount and classification of items of income and expense before and after the change in the assessment.

D

Example 3.7-2 – change in assessment of specified main business activities

Fact pattern

As at 31 December 20x4, Entity Y concludes that it does not invest in real estate assets (e.g. investment properties measured at fair value) as a main business activity. Therefore, it presents rental income of CU100 and fair value gains on investment property of CU240 in the investing category for the year-ended 31 December 20x4.

As at 1 January 20x5, Entity Y concludes that its main business activities have changed and it does invest in real estate assets as a main business activity. Therefore, it presents rental income of CU110 and fair value gains on investment property of CU225 in the operating category for the year-ended 31 December 20x5.

<u>Assessment</u>

In preparing its 31 December 20x5 financial statements, Entity Y applies IFRS 18.B41 and reflects its change in its assessment of whether it invests in assets as a main business activity prospectively (i.e. the 31 December 20x4 comparative period is not restated).

Entity Y provides disclosures required by IFRS 18.51(c)(i), being the fact of the change of the assessment and the date of the change (1 January 20x5). Entity Y also meets the disclosure requirement of IFRS 18.51(c)(ii) by disclosing the amount and classification of the items reclassified as a result of the change of the assessment:

Item of income/expense	31 December 20x5 classification	31 December 20x4 classification
Rental income	CU110 operating category	CU100 investing category
Fair value gains on investment property	CU225 operating category	CU240 investing category

3.7.9. Disclosures related to an entity's assessment of specified main business activities

- IFRS 18.51 requires the following disclosures:
- (a) If an entity invests in assets as a main business activity, it shall disclose that fact.
- (b) If an entity provides financing to customers as a main business activity, it shall disclose that fact.
- (c) Those disclosures required when an entity changes its assessment of whether it has specified main business activities (see section 3.7.8 'change in assessment' above).

If any entity does not have any specified main business activities and no change in that assessment in the current period, none of the disclosures in IFRS 18.51 will be relevant.

3.8. Additional requirements for entities with specified main business activities

Sections <u>3.8.1</u> and <u>3.8.2</u> explain how the classification requirements of IFRS 18 differ for entities with specified main business activities. If an entity does not have any specified main business activities, these sections are not relevant, and an entity would apply the general classification requirements (see sections <u>3.1</u> - <u>3.6</u>).

3.8.1. Investing in assets as a main business activity

If an entity invests in assets as a main business activity, the general classification requirements of IFRS 18 are modified. The classification requirements applicable to assets (and related income and expenses) are summarised as follows:



The following sections explain how the classification requirements of IFRS 18 apply to entities that invest in assets as a main business activity, by each of the four classes of assets in the above flowchart.



3.8.1.1. Cash and cash equivalents

See section <u>3.9.1</u> for the requirements applicable to classifying income and expenses relating to cash and cash equivalents.

In summary, if an entity invests in financial assets as a main business activity (and therefore classifies incomes and expenses from financial assets in the operating category), then it is also required to classify income and expenses from cash and cash equivalents in the operating category as well (IFRS 18.56(a)).

If the entity provides financing to customers as a main business activity (see section <u>3.8.2</u>), then the classification requirements are more complex.

3.8.1.2. Investments in associates, joint ventures and unconsolidated subsidiaries

As noted in sections <u>3.2</u> and <u>3.8</u>, it is not necessary for an entity to assess whether it invests in associates, joint ventures and unconsolidated subsidiaries as a main business activity if the entity accounts for such investments using the equity method. That is because if the equity method is applied to such investments, then income and expenses relating to such investments will always be classified in the investing category, regardless of whether an entity has specified main business activities (IFRS 18.B38; see section <u>3.2</u>).

However, if the equity method is not applied to such investments, then an entity must assess if it invests in those assets as a main business activity (see section <u>3.8</u> for guidance on this assessment applicable to investments in associates, joint ventures and unconsolidated subsidiaries). If the entity determines that it does have a main business activity of investing in those assets, then it classifies income and expenses relating to those assets in the operating category.

For example, an investment entity as defined in IFRS 10 may invest in associates and joint ventures together with subsidiaries as a main business activity and would therefore measure those investments at fair value through profit or loss. An investment entity would classify those fair value gains and losses in the operating category.

3.8.1.3. Other assets that generate a return individually and largely independently of the entity's other resources

If an entity invests in assets as a main business activity, it classifies income and expenses relating to those assets in the operating category rather than the investing category (IFRS 18.58).

These types of assets typically include (IFRS 18.B46):

- Debt or equity investments; and
- ▶ Investment properties, and receivables for rent generated by those properties.

As noted in section <u>3.2</u>, these income and expenses are:

Income and expenses (IFRS 18.54)	Common examples (IFRS 18.54 and B47)
(a) The income generated by the assets	 Interest Dividends Rental income
(b) The income and expenses that arise from the initial and subsequent measurement of the assets, including on derecognition of the assets	 Depreciation Impairment losses and reversal of impairment losses Fair value gains and losses
(c) The incremental expenses directly attributable to the acquisition and disposal of the assets	 Transaction costs on financial assets classified as fair value through profit or loss Costs to sell assets, such as broker commissions on financial instruments



BDO comment - classification of income and expenses from multiple classes of assets

As noted in section <u>3.8</u>, an entity may invest in one or more classes of assets as a main business activity but not others. Said another way, an entity may only classify in the operating category income and expenses specified in IFRS 18.54 for assets that it invests in as a main business activity.

For example, an entity may invest in real estate as a main business activity but not financial assets. Therefore, it would classify income and expenses relating to its real estate assets (e.g. rental income, fair value gains and losses on investment property) in the operating category, but not income and expenses relating to its financial assets.

'Investing in assets' as a main business activity is not a binary classification for <u>all asset classes</u>, it is an assessment made for individual classes of assets, with the conclusion affecting the classification requirements applicable to classes of assets separately.

3.8.1.4. Other assets

For assets that do not generate a return individually and largely independently of the entity's other resources, associated income and expenses are classified in the operating category. See section 3.2.

3.8.2. Providing financing to customers as a main business activity

If an entity provides financing to customers as a main business activity, the general classification requirements of IFRS 18 are modified. The classification requirements applicable to liabilities (and related income and expenses) are summarised as follows:





3.8.2.1. Liabilities that arise from transactions that involve only the raising of finance

For entities that provide financing to customers as a main business activity, liabilities that arise from transactions that involve only the raising of finance must be further sub-categorised into liabilities that (IFRS 18.65(a)): (i) Relate to providing financing to customers; and

(ii) Do not relate to providing financing to customers.

For 'pure financing' liabilities that relate to providing financing to customers, associated income and expense are always classified in the operating category.

BDO comment – classification of income and expenses from 'pure financing' liabilities that relate to providing financing to customers

If an entity provides financing to customers as a main business activity, then income and expenses related to liabilities that arise from transactions that involve only the raising of finance are classified in the operating category.

For example, banks and other lending institutions commonly receive loans from investors or other lenders and then lend those funds to the bank's customers with the goal of earning a margin on the interest rate differential between the loan payable and the loans receivable. If a lending institution borrows money in order to loan the proceeds to its customers, then that borrowing would be 'related to providing financing to customers'. Therefore, the income and expenses noted in IFRS 18.60 would be classified in the operating category in accordance with IFRS 18.65(a)(i):

- (a) income and expenses that arise from the initial and subsequent measurement of the liabilities, including on derecognition of the liabilities; and
- (b) the incremental expenses directly attributable to the issue and extinguishment of the liabilities—for example, transaction costs.

Determining whether liabilities are related to providing financing to customers may be challenging to determine in practice.

For 'pure financing' liabilities that do not relate to providing financing to customers, an accounting policy choice exists to classify the related income and expenses specified in IFRS 18.60 in either the operating category or the financing category (IFRS 18.65(a)(ii)). However, there are two restrictions on this accounting policy choice:

- if an entity cannot distinguish between the liabilities that relate to providing financing to customers and those that do not, an entity is required to classify income and expenses from all such liabilities in the operating category (IFRS 18.66).
- Consistency with IFRS 18.56(b)(ii): the accounting policy choice must be consistent with the accounting policy selected for the classification of income and expenses related to cash and cash equivalents for entities that provide financing to customers as a main business activity, do not invest in financial assets as a main business activity and where the income and expenses from cash and cash equivalents do not relate to providing financing to customers (IFRS 18.56(b)(ii)). See 3.9.1 for further information on classifying income related to cash and cash equivalents.

BDO comment – accounting policy choice in IFRS 18.56(b)(ii) and effect on the classification of income and expenses arising from certain derivatives (IFRS 18.859)

The accounting policy selected above applies to the classification of all income and expenses relating to 'pure financing' liabilities that do not relate to providing financing to customers as well as gains and losses on certain derivative financial instruments (see section <u>3.9.4</u>) if:

(1) a derivative is used for purposes other than managing identified risks;

- (2) the derivative is related to a transaction that involves only the raising of finance;
- (3) the entity provides financing to customers as a main business activity; and
- (4) the derivative does not relate to providing financing to customers.

The accounting policy selected above for 'pure financing' liabilities that do not relate to providing financing to customers affects the classification of gains and losses on certain derivatives because IFRS 18.B73(a) and B59

requires consistency in how income and expenses from derivatives are classified if those derivatives have the four characteristics noted in the preceding paragraph. Therefore, if an entity elected to present income and expenses from <u>all</u> 'pure financing' liabilities in the operating category, the entity would also present income and expenses from derivatives that have the four characteristics noted above in the operating category as well.

See section 3.9.4 for discussion of this accounting policy choice as it relates to gains and losses on certain derivative financial instruments.

Example 3.8.2-1 – consistency of accounting policy choice for the classification of income and expenses related to (1) 'pure financing' liabilities that do not relate to providing financing to customers; and (2) cash and cash equivalents

Applying the requirement of IFRS 18.56(b)(ii) to ensure a consistent accounting policy choice may be challenging to understand, therefore, this worked example demonstrates the requirements.

Fact pattern

Entity G provides financing to customers as a main business activity, as it sells farm equipment and provides extended financing terms to customers. Entity G holds cash and cash equivalents and earns interest income on those assets; however, Entity G does not invest in financial assets as a main business activity and the income and expenses arising from cash and cash equivalents does not relate to providing financing to customers.

To finance its operations, Entity G has several bank loans. Some of the bank loans are specific to providing financing to customers, and others are not. Entity G is able to distinguish between the liabilities that relate to providing financing to customers and those that do not. In accordance with IFRS 18.65(a)(i), Entity G classifies income and expenses on the bank loans that are specific to providing financing to customers in the operating category.

Entity G is assessing how it may classify income and expenses relating to its bank loans that are not specifically related to providing financing to customers.

<u>Assessment</u>

For the bank loans that are not specifically related to providing financing to customers, Entity G has an accounting policy choice to make concerning how related income and expenses are classified. Entity G is able to distinguish between the liabilities that relate to providing financing to customers and those that do not (IFRS 18.66), therefore, it may select either the operating category or financing category.

Entity G's application of the classification flowchart for income and expenses for entities that provide financing to customers as a main business activity is as follows, with the 'non-customer specific' bank loans in green:





At the same time, Entity G must also select an accounting policy relating to how it classifies income and expenses arising from cash and cash equivalents. Entity G's application of the classification flowchart for income and expenses from cash and cash equivalents (see section <u>3.9.1</u>) is as follows, with Entity G's assessment in orange:



Because of its particular facts and circumstances, Entity G now has a choice of classifying income and expenses relating to:

- 'Pure financing' liabilities not related to providing financing to customers in the operating or financing category; and
- Cash and cash equivalents in the operating or investing category.

In accordance with IFRS 18.56(b)(ii), this accounting policy choice must be consistent, meaning, if Entity G classifies income and expenses relating to the 'pure financing' liabilities not related to providing financing to customers in the operating category, it must make the same accounting policy choice for income and expenses relating to cash and cash equivalents.

If Entity G selected the financing category for 'pure financing' liabilities not related to providing financing to customers in the financing category (i.e. not the operating category), Entity G must classify income and expenses relating to cash and cash equivalents in the investing category.

3.8.2.2. Liabilities that arise from transactions that do not involve only the raising of finance ('other liabilities')

For other liabilities, income and expenses are classified into the operating and financing categories based on the general requirements of IFRS 18 (see section <u>3.3</u>). No exceptions exist for entities that provide financing to customers as a main business activity.

Therefore, for example, an entity that provides financing to customers as a main business activity would classify interest expense relating to lease liabilities in the financing category. See section <u>3.3</u> for other examples of the classification of income and expenses relating to 'other liabilities'.

3.8.2.3. Cash and cash equivalents

See section <u>3.9.1</u> for the requirements applicable to classifying income and expenses relating to cash and cash equivalents.

In summary, if an entity provides financing to customers as a main business activity but does not invest in financial assets as a main business activity, the entity may have an accounting policy choice to make concerning the classification of income and expenses relating to cash and cash equivalents subject to the restriction mentioned in section 3.9.1.

3.9. Considerations for specific items of income and expense

This section addresses specific requirements for classification income and expense related to the following assets where the general requirements discussed in section 3.1-3.8 do not apply:

- Cash and cash-equivalent (see section 3.9.1)
- Hybrid contracts (see section 3.9.2)
- ▶ Foreign exchange differences (see section 3.9.3)
- Derivatives (see section 3.9.4)

Section 3.9.5 addresses the classification requirements of IFRS 18 applicable to specific income and expenses where no specific rules apply, but the application of IFRS 18's requirements are illustrated.

3.9.1. Cash and cash equivalents

The classification requirements for income and expenses relating to cash and cash equivalents (e.g. interest income) are summarised as follows:





In summary:

- An entity that invests in financial assets as a main business activity will always classify income and expenses related to cash and cash equivalents in the operating category (IFRS 18.56(a)). This conclusion remains unchanged even if an entity invests in financial assets as a main business activity and it provides financing to customers as a main business activity (i.e. it has more than a single specified main business activity).
- If an entity does not invest in financial assets as a main business activity and does not provide financing to customers as a main business activity, income and expenses related to cash and cash equivalents will always be classified in the investing category (IFRS 18.56).
- Where an entity does not invest in financial assets as a main business activity, but does provide financing to customers as a main business activity, it must determine if the income and expenses related to cash and cash equivalents relate to providing financing to customers (IFRS 18.56(b)):
 - If they do, income and expenses are classified in the operating category (IFRS 18.56(b)(i)).
 - If they do not, an accounting policy choice exists to classify income and expenses either in the operating or the investing category, subject to two restrictions (see below) (IFRS 18.56(b)(ii)).

The accounting policy choice noted above is subject to two restrictions:

- 1. It must be consistent with the accounting policy choice selected for 'pure financing' liabilities that do not relate to providing financing to customers, which is discussed in section <u>3.8.2</u>, see <u>Example 3.8.2-1</u>.
- 2. If an entity cannot distinguish between cash and cash equivalents that relate to providing finance to customers and those that do not, an entity is required to classify income and expenses from all cash and cash equivalents in the operating category (IFRS 18.57).

3.9.2. Hybrid contracts containing a host that is a liability

Hybrid contracts containing a host that is a liability may include, for example:

- ▶ Convertible notes with conversion options that are not closely related to the host contract;
- > Lease liabilities with embedded derivatives not closely related to the host contract; and
- ▶ Insurance contract liabilities with embedded derivatives not closely related to the host contract.

The embedded derivatives may either be separated or if certain criteria are met, an entity may elect to measure the entire hybrid instrument at fair value through profit or loss (IFRS 9.4.3.5). Hybrid contracts may also not have the embedded derivative accounted for separately from the host contract is the embedded derivative is considered closely related to the host contract.

IFRS 18.B56-B57 set out detailed classification requirements applicable to income and expenses related to hybrid contracts containing a host liability, with the primary distinction being whether the embedded derivative is separated from the host liability or not.

3.9.2.1. Hybrid contracts containing a host liability – embedded derivative is separated from the host liability (IFRS 18.B56(a))

The classification requirements for income and expenses relating to hybrid contracts containing a host liability where the embedded derivative is separated are summarised as follows (assuming that the entity does not provide financing to customers as a main business activity):

Income and expenses from hybrid contracts with host liabilities – embedded derivative is separated from the host liability (entities that do not provide financing to customers as a main business activity)



In summary, income and expenses relating to the hybrid instrument are classified separately for the host liability and the separated embedded derivative(s):

- ▶ The general requirements of IFRS 18 apply in classifying income and expenses relating to host liabilities. This flowchart summarises the same requirements as described in section <u>3.3</u>, meaning that they assume that an entity does not provide financing to customers as a main business activity. If an entity does provide financing as a main business activity, it must apply the requirements described in section <u>3.8.2</u> to the host liability.
- ▶ The requirements applicable to derivatives apply to the embedded derivative that is separated from the host liability (see section <u>3.9.4</u>).

These requirements are consistent with the fact that when an embedded derivative is separated from the host contract, the two components (the host liability and the embedded derivative) are accounted for separately from one another. Therefore, the classification requirements of IFRS 18 apply to those two components separately.



Example 3.9.2-1 – classification of income and expenses arising from a convertible note – embedded derivative is separated from the host liability

Fact pattern

Entity M issues a convertible note where the principal of the loan may be converted into Entity M's equity instruments at the option of the holder at any time. The conversion option does not meet the 'fixed for fixed' criterion because the currency of the note differs from the functional currency of Entity M, meaning that the amount of 'cash' that would be settled upon conversion is not fixed in the functional currency of Entity M. Entity M does not elect to use the fair value option in IFRS 9.4.3.5.

Entity M does not provide financing to customers as a main business activity.

Entity M separates the embedded derivative from the host contract at initial recognition, as it does not elect to use the fair value option. The host liability is measured at amortised cost and the embedded derivative is measured at fair value through profit or loss.

During the reporting period where the convertible note remains outstanding, the host liability and the embedded derivative give rise to the following income and expenses:

Component	Income and expenses
Host liability	Interest expense accounted for applying the effective interest method (IFRS 9)
Embedded derivative (conversion option)	Fair value gains and losses (IFRS 9)

Assessment – host liability

Entity M classifies the interest expense relating to the host liability as follows (the relevant assessment is circled in red dotted lines using the classification flowchart from earlier in this section):



Income and expenses from hybrid contracts with host liabilities – embedded derivative is separated from the host liability (entities that do not provide financing to customers as a main business activity)



Explanation:

- ▶ The host liability is a liability that arises from a transaction that involves only the raising of finance because Entity M receives finance in the form of cash and will return cash or its own equity instruments in exchange at a later date (IFRS 18.B50).
- Income and expenses that arise from the initial and subsequent measurement of that liability are therefore classified in the financing category (IFRS 18.60).

Assessment - embedded derivative

Entity M classifies the fair value gains and losses relating to the embedded derivative conversion option as follows (the relevant assessment is circled in red dotted lines using the classification flowchart from section 3.9.4, which addresses the classification of gains and losses on derivatives):



Gains and losses on derivatives and designated hedging instruments (IFRS 18.B70-B76)

Explanation:

- The embedded derivative conversion option is not used to manage identified risks (e.g. foreign currency or interest rate risk).
- ▶ The derivative relates to a transaction that involves only the raising of finance, as the derivative is embedded in a host liability that is considered to be a liability that arises from transactions that involve only the raising of finance. See the section immediately above which explains this classification of the host liability.
- Entity M does not provide financing to customers as a main business activity.
- Therefore, the fair value gains and losses on the derivative are classified in the financing category.



3.9.2.2. Hybrid contracts containing a host liability – embedded derivative is not separated from the host liability (IFRS 18.B56(b)-(c))

The classification requirements for income and expenses relating to hybrid contracts containing a host liability where the embedded derivative is not separated are summarised as follows (assuming that the entity does not provide financing to customers as a main business activity):

Income and expenses from hybrid contracts with host liabilities – embedded derivative is <u>not</u> separated from the host liability (entities that do not provide financing to customers as a main business activity)



In summary, income and expenses relating to the hybrid instrument with a non-separated embedded derivative are classified as follows:

- If the hybrid contract arises from a transaction that involves only the raising of finance, all income and expenses noted in IFRS 18.60 are classified in the financing category, consistent with other 'pure financing' liabilities (IFRS 18.B56(b)).
- ▶ If the hybrid contract arises from a transaction that does not involve only the raising of finance, the classification depends on the nature of the hybrid contract/the host liability (IFRS 18.B56(c)):
 - If the host liability is a financial liability within the scope of IFRS 9 that is measured at amortised cost, then income and expenses specified in IFRS 18.60 arising after initial recognition are classified in the financing category (IFRS 18.B56(c)(i)). All other income and expenses are classified in the operating category.
 - If the hybrid contract is an insurance contract in the scope of IFRS 17, income and expenses are classified in the operating category (IFRS 18.B56(c)(ii)).
 - In all other cases, an entity applies the requirements applicable to liabilities that arise from transactions that do
 not involve only the raising of finance (IFRS 18.B56(c)(iii)). However, if a hybrid contract falls within the scope of
 IFRS 18.B56(c)(iii) and is accounted for at fair value through profit or loss using the fair value option (IFRS 9.4.3.5),

no income and expenses will be classified in the financing category because that contract will not give rise to (1) interest income and expenses; or (2) income and expenses from changes in interest rates (IFRS 18.61). See <u>Example</u> <u>3.9.2-4</u>.

The above flowchart assumes that an entity does not provide financing to customers as a main business activity. If an entity does provide financing to customers as a main business activity, then conclusions will differ. For example, for an entity that provides financing to customers as a main business activity, income and expenses relating to a hybrid contract (with a non-separated embedded derivative) arising from a transaction that involves only the raising of finance will be classified in the operating category rather than the financing category (IFRS 18.65(a)(i)).

BDO comment – reasons why an embedded derivative may not be accounted for separately from the host contract

Note that these requirements apply to hybrid contracts where the embedded derivative is not separated from the host contract regardless of the <u>reason</u> for the embedded derivative not being separated (IFRS 18.B57). This may be the case because of:

- ▶ IFRS 9.4.3.3: an embedded derivative may be identified in a host contract, but it is not separated and accounted for as a derivative if it is considered 'closely related' to the host contract (see criteria in IFRS 9.4.3.3).
- IFRS 9.4.3.5: an entity is permitted to designate the entire hybrid contract as at fair value through profit or loss (meaning the embedded derivative is not accounted for separately) if criteria in IFRS 9.4.3.5 are met.

If the criteria in IFRS 9.4.3.3 are met, an embedded derivative is not accounted for separately from the host contract, however, the contract is still technically a 'hybrid contract', meaning the special considerations in IFRS 18 still apply. See Example 3.9.2-3 for an illustration of the non-separated embedded derivatives hybrid contract flowchart being used to analyse how income and expenses are classified arising from a contract where the embedded derivative is not separated because it is considered closely related to the host contract.

D

Example 3.9.2-2 – classification of income and expenses arising from a convertible note – embedded derivative is <u>not</u> separated from the host liability because the fair value option is elected (IFRS 9.4.3.5)

Fact pattern

Entity N issues a convertible note where the principal of the loan may be converted into Entity N's equity instruments at the option of the holder at any time. The conversion option does not meet the 'fixed for fixed' criterion because the currency of the note differs from the functional currency of Entity N meaning that the amount of 'cash' that would be settled upon conversion is not fixed in the functional currency of Entity N. Entity N elects to use the fair value option in IFRS 9.4.3.5, meaning the entire hybrid instrument is measured at fair value through profit or loss.

Entity N does not provide financing to customers as a main business activity.

During the reporting period where the convertible note remains outstanding, the hybrid contract gives rise to the following income and expenses:

Component	Income and expenses
Hybrid contract	Fair value gains and losses (IFRS 9)

While the convertible note may result in coupon interest payments to note holders, interest expense is not recognised on the hybrid contract because it is not measured at amortised cost. For a financial liability measured at fair value through profit or loss, the entire movement in the carrying amount from one reporting period to the next is reflected as fair value gains and losses (except for changes in credit risk relating to that liability, which are reflected in other comprehensive income – see IFRS 9.5.7.7).

<u>Assessment – hybrid contract</u>

Entity N classifies the fair value gains and losses related to the hybrid contract as follows (the relevant assessment is circled in red dotted lines using the classification flowchart from earlier in this section):

Income and expenses from hybrid contracts with host liabilities – embedded derivative is <u>not</u> separated from the host liability (entities that do not provide financing to customers as a main business activity)



Explanation:

- ▶ The fair value gains and losses on the hybrid contract being remeasured to fair value at each reporting period are classified in the finance category because the hybrid contract arises from a transaction that involves only the raising of finance. That is because Entity N receives finance in the form of cash and will return cash in exchange at a later date (IFRS 18.B50).
- Note that the amount of the change in fair value of the hybrid contract that is attributable to changes in the credit risk of the liability will be presented in other comprehensive income (IFRS 9.5.7.7). Amounts presented in other comprehensive income are not subject to the classification requirements of IFRS 18 other than the requirement to distinguish amounts that will and will not be reclassified to profit or loss when specific conditions are met (IFRS 18.88).



Example 3.9.2-3 – classification of income and expenses arising from a hybrid contract in the scope of IFRS 18.B56(c)(iii) - embedded derivative is not separated from the host liability because it is considered closely related to the host contract (IFRS 9.4.3.3)

Fact pattern

Entity O enters into a lease contract as a lessee for a period of 5 years where the underlying asset is a shop in a shopping mall. Lease payments are CU100 per month, however, Entity O must also pay 5% of its monthly turnover as a variable lease payment.

The turnover-based lease payment term is identified as an embedded derivative in the host contract; however, it is not accounted for separately because it is considered closely related to the host contract. That is because IFRS 9.B4.3.8(f) specifies that variable lease payments related to sales are closely related to the host contract.

During the reporting period where the lease remains outstanding, the hybrid contract gives rise to the following income and expenses:

Component	Income and expenses
Turnover-based lease payments	Variable lease payments recognised in profit or loss as the underlying sales occur (IFRS 16.38(b))
Recognised lease liability (fixed lease payments)	Interest expense on the lease liability recognised (IFRS 16.36)



Assessment

Income and expenses from hybrid contracts with host liabilities – embedded derivative is <u>not</u> separated from the host liability (entities that do not provide financing to customers as a main business activity)



Explanation:

- The hybrid contract (the lease) does not arise from a transaction that involves only the raising of finance. That is because Entity O receives a right-of-use asset, not finance in the form specified by IFRS 18.B50(a).
- The hybrid contract is not (a) a financial liability in the scope of IFRS 9 that is measured at amortised cost; or (b) an insurance contract in the scope of IFRS 17, therefore, the 'other' requirements apply.
- The interest expense recognised on the lease liability is classified in the financing category because it is an interest expense that is identified for the purpose of applying IFRS 16 (IFRS 18.61(a)).
- The variable lease payments recognised in profit or loss are classified in the operating category because they are not (1) interest income and expenses); or (2) income and expenses from changes in interest rates.





Example 3.9.2-4 – classification of income and expenses arising from a hybrid contract (host contract is executory) in the scope of IFRS 18.B56(c)(iii) - embedded derivative is not separated from the host liability because the fair value option is elected (IFRS 9.4.3.5)

Fact pattern

Entity P enters into a contract to purchase a commodity (e.g. oil). The purchase contract contains embedded derivatives and Entity P elects to measure the entire purchase contract at fair value through profit or loss in accordance with IFRS 9.4.3.5.

During the reporting period where the purchase contract remains outstanding, the hybrid contract gives rise to the following income and expenses:

Component	Income and expenses
Hybrid contract	Fair value gains and losses (IFRS 9)



Assessment

Income and expenses from hybrid contracts with host liabilities – embedded derivative is <u>not</u> separated from the host liability (entities that do not provide financing to customers as a main business activity)



Explanation:

- The hybrid contract (the purchase contract) does not arise from a transaction that involves only the raising of finance. That is because Entity P is receiving a commodity, not finance in the form specified by IFRS 18.B50(a).
- The hybrid contract is not (a) a financial liability in the scope of IFRS 9 that is measured at amortised cost; or (b) an insurance contract in the scope of IFRS 17, therefore, the 'other' requirements apply.
- The fair value gains recognised in profit or loss are classified in the operating category because they are not (1) interest income and expenses); or (2) income and expenses from changes in interest rates.





Foreign exchange differences are included in profit or loss when a monetary item is denominated in a currency other than the entity's functional currency. For example, if an entity has the Euro as its functional currency but holds trade receivables denominated in US dollars, foreign exchange gains and losses will be recognised in profit or loss in accordance with IAS 21.

Foreign exchange differences included in profit or loss applying IAS 21 are classified in the same category as the income and expenses from the items that give rise to the foreign exchange differences, unless doing so would involve undue cost or effort (IFRS 18.B65).

For example, an entity classifies foreign exchange differences as follows (IFRS 18.B66):

Source of foreign exchange difference	Explanation
A trade receivable denominated in a foreign currency	Operating category, as trade receivables are assets that do not generate a return individually and largely independently of the entity's other resources, meaning associated income and expenses are classified in the operating category.
Liabilities that arise from transactions that involve only the raising of finance (the entity does not provide financing as a main business activity)	Financing category, as that matches the income and expenses arising from the debt (e.g. interest expense).

Judgement is required to determine the amount of foreign exchange differences that should be classified in different categories.

3.9.3.1. Income and expenses classified in more than one category

In certain circumstances, an entity may classify income and expenses arising from a transaction in more than one category. Note that this is only possible if the transaction does not involve only the raising of financing (i.e. these requirements would not be applicable to a bank loan).

For example, an entity may enter into a service agreement (e.g. consulting services) denominated in a foreign currency with a supplier to pay the supplier on deferred terms (e.g. the first 6 months of services paid over 18 months). This transaction would give rise to two elements:

- 1. The service expenses, which would generally be classified in the operating category as service is provided over the 6-month period; and
- 2. The finance expense due to the deferred credit terms, which would be classified in the financing category because the resulting liability does not arise from transactions that involve only the raising of finance and the finance expense arising from the deferred payment terms are interest expense identified by applying IFRS Accounting Standards (IFRS 18.61).

This transaction would also give rise to foreign exchange differences in the scope of IAS 21 because it is denominated in a foreign currency. Therefore, foreign exchange differences would arise because of the difference between when the financial liability is recognised at the date services are provided and when the financial liability is settled.

An entity uses its judgement to determine how to classify these foreign exchange differences (i.e. in the operating category or the financing category). If the foreign exchange differences relate to a liability that does not arise from transactions that involve only the raising of financing (e.g. the liability owing to the service provider in the example above), then all of the foreign exchange differences must be classified in a single category: operating or financing (IFRS 18.B67). The classification of foreign exchange differences described in this paragraph is not an accounting policy choice, but the result of the application of judgement. After that judgement is applied, all foreign exchange differences relating to that liability (and similar liabilities) must be classified in that single category.

In all other cases, judgement is required to determine the amount of foreign exchange differences that should be classified in different categories.



3.9.3.2. Undue cost or effort

If applying these requirements would involve undue cost or effort, an entity is required to classify the affected foreign exchange differences entirely in the operating category (IFRS 18.B68). This assessment of 'undue cost or effort' is not made for the reporting entity as a whole (i.e. all foreign exchange differences), but for each item that gives rise to foreign exchange differences.

BDO comment – effect on systems and process

Many entities' systems will not be designed to capture the information required by IFRS 18 for the classification of foreign exchange differences. Entities may have captured foreign exchange differences in a single line item in the statement of profit or loss previously, while some entities may have recorded exchange differences relating to financing (e.g. bank loans) as part of 'finance costs'.

The work effort involved to apply these new requirements will differ from entity to entity. 'Undue cost or effort' is generally considered to be a high threshold, meaning that an entity is not permitted to simply classify all foreign exchange differences in the operating category by virtue of there being costs associated with modifying their systems and processes to comply with the requirements of IFRS 18.

3.9.3.3. Hyperinflationary entities - the net monetary position

Entities that apply IAS 29 *Financial Reporting in Hyperinflationary Economies* may present the gain or loss on the net monetary position with other income and expense items associated with the net monetary position (e.g. interest income and expenses and foreign exchange differences). If an entity does not present the gain or loss on the net monetary position in this manner, it must be classified in the operating category (IFRS 18.B69).

3.9.4. Gains and losses on derivatives and designated hedging instruments

The classification requirements for income and expenses relating to derivatives and designated hedging instruments are summarised as follows:





Gains and losses on derivatives and designated hedging instruments (IFRS 18.B70-B76)

The flowchart is explained in the following sub-sections.

3.9.4.1. Derivatives used to manage identified risks

Derivative contracts may be used to manage identified risks, such as:

Derivative financial instrument	Identified risk that may be managed
Interest rate swap that swaps fixed interest payments for variable or vice versa	Interest rate risk
Foreign currency forward contract to buy or sell foreign currency at a future date at a fixed exchange rate	Foreign currency risk
Option contract to sell an equity instrument at a pre- determined price in the future	Other price risk

Derivatives may be designated as hedging instruments in a qualifying hedge accounting relationship in IFRS 9, or they may not. Regardless of whether hedge accounting is applied, gains and losses included in the statement of profit or loss are generally classified in the same category as the income and expenses affected by the risks the financial instrument is used to manage (IFRS 18.B70). The ineffective portion of a gain or loss is also reflected in the same category as the effective portion (IFRS 18.B71).

Gains and losses are not classified in the same category as the income and expenses affected by the risks when hedge accounting is not applied and (i) doing so would require the grossing up of gains or losses; or (ii) doing so would involve undue cost or effort. If either (i) or (ii) apply, then all gains and losses are instead classified in the operating category (IFRS 18.B72).



The undesignated component of a designated hedging instrument has its gains and losses recorded in the same category as the gains and losses on the designated component (IFRS 18.B71).

For example, if an interest rate swap is used to manage interest rate risk on a bank loan that arises from transactions that involve only the raising of finance, the gains and losses recognised in profit or loss relating to that swap are classified in the financing category to 'match' where income and expenses related to the bank loan are classified in accordance with IFRS 18.60. This 'matching' occurs regardless of whether the entity applies hedge accounting, as long as the derivative is used to manage identified risks (IFRS 18.870, B72). Any ineffective portion would also be classified in the financing category (IFRS 18.871).

3.9.4.2. 'Grossing up' of gains and losses on derivatives used to manage identified risks

The 'matching' of the gains and losses on derivatives recognised in profit or loss with the classification of the income and expenses affected by the financial instrument used to manage that risk is required unless in order to achieve this effect, the entity would be required to 'gross up' gains and losses. If this is the case, all such gains and losses on the derivative are classified in the operating category IFRS 18.B70).

Grossing up of gains and losses might arise from situations in which (IFRS 18.B74):

- ► An entity uses such financial instruments to manage the risks of a group of items with offsetting risk positions (see IFRS 9.6.6.1 for the criteria for a group of items to be an eligible hedged item); and
- ▶ The risks managed affect line items in more than one category of the statement of profit or loss.

For example, an entity may use a derivative to manage both the net currency risk on revenue (classified in the operating category) and interest expense (classified in the financing category). In such cases, presenting gain or loss on the derivative in each category would result in grossing up gains and losses, as an entity would present a larger gain or loss in each category than the gain or loss that occurred on the derivative. (IFRS 18.B75).

3.9.4.3. Derivatives used for purposes other than managing identified risks

Entities may hold derivatives that are not used to manage identified risks. For example, an entity may:

- Issue share purchase warrants on their own equity instruments which fail the 'fixed for fixed' test in IAS 32, and are therefore measured at fair value through profit or loss.
- ▶ Hold and/or issue derivatives as part of trading activities with an aim of earning profit rather than managing specific risk exposures. This may be common for banks and financial institutions.

If derivatives are used for purposes other than managing identified risks and they are not related to a transaction that involves only the raising of finance, gains and losses must be classified in the operating category (IFRS 18.B73(b)).

If derivatives are used for purposes other than managing identified risks and they are related to a transaction that involves only the raising of finance, two further assessments are necessary:

- ▶ If the entity does not provide financing to customers as a main business activity, then gains and losses are classified in the financing category (IFRS 18.B73(a)), which for example, results in those gains and losses 'matching' the interest income and expenses arising from the liabilities related to transactions that involve only the raising of finance (IFRS 18.60).
- ▶ If the entity does provide financing to customers as a main business activity:
 - If the derivative does relate to providing finance to customers, gains and losses are classified in the operating category (IFRS 18.B73(a)).
 - If the derivative does not relate to providing finance to customers, the entity has an accounting policy choice of classifying gains and losses in the operating or financing category (IFRS 18.65(a)(ii).

BDO comment – accounting policy choice in IFRS 18.65(a)(ii) available to entities that provide financing to customers as a main business activity

As noted above, if (1) a derivative is used for purposes other than managing identified risks; (2) the derivative is related to a transaction that involves only the raising of finance; (3) the entity provides financing to customers as a main business activity; and (4) the derivative does not relate to providing financing to customers; then an entity has an accounting policy choice to classify gains and losses in the operating or financing category (IFRS 18.65(a)(ii)).

This accounting policy choice is the same as the one discussed in section <u>3.8.2</u>, where for 'pure financing liabilities', an entity that provides financing to customers as a main business activity may classify income and expenses on liabilities that do not relate to providing financing to customers in either the operating or financing category.

As this is the same accounting policy choice, an entity must be consistent. Therefore, two different classes of income and expenses must be classified consistently for entities that provide financing to customers as a main business activity:

- ▶ Gains and losses on derivatives where the accounting policy election described above is available; and
- Other income and expenses (e.g. interest expense) on 'pure financing liabilities' that do not relate to providing financing to customers.

All income and expenses for the two classes of transactions must be classified consistently: all such expenses in the operating category or the financing category.

Additionally, as this is a single accounting policy choice, this choice is subject to the restrictions noted in section 3.8.2 concerning consistency in the policy selected for the classification of cash and cash equivalents (see sections 3.8.2 and 3.9.1).

3.9.5. Guidance on common items of income and expense

The following section provides practical guidance on the classification of common items of income and expense applying the requirements of IFRS 18. Where the classification of the items may differ depending on whether an entity has specified main business activities, that consideration has been noted. Where the classification of that item is addressed in another section of this publication, the reference to that section is provided. The order of items is presented based on the sequencing of the related IFRS Accounting Standard.

This section sets out how items of income and expenses will generally be classified applying the requirements of IFRS 18 discussed in the preceding sections, however, this section should be taken as broad guidance only. The classification of particular income and expenses will depend on facts and circumstances.



Item of income or expense	Classification	Explanation
Inventories expensed, including write-downs to net realisable value and reversals of write-downs (IAS 2)	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). See sections 3.2 and 3.4 .
Income taxes expenses, both current and deferred (IAS 12)	Income taxes	Income and expenses within the scope of IAS 12 are classified in the income taxes category. See section 3.5 .
Depreciation of property, plant and equipment (IAS 16)	Generally operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). Property, plant and equipment typically does not generate a return individually and largely independently of the entity's other resources, however, this may not always be the case. See section <u>3.4</u> .
Gains and losses from the sale of property, plant and equipment (IAS 16)	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). See section <u>3.4</u> .
Finance expense on employee benefit liabilities (IAS 19)	Financing	Liability arises from transactions that do not involve only the raising of finance (IFRS 18.59(b)) and are interest income and expenses identified for purposes of applying IFRS Accounting Standards (IFRS 18.61(a)). See section <u>3.3</u> .
Current service cost on post-employment benefits (IAS 19)	Operating	Liability arises from transactions that do not involve only the raising of finance (IFRS 18.59(b)) but current service cost is not (a) interest income and expenses identified for purposes of applying IFRS Accounting Standards; or (b) income and expenses arising from changes in interest rates. See section <u>3.3</u> .
Government grants	Depends on facts and	Income related to government grants may be:
recognised in income (IAS 20)	circumstances and accounting policy choices	 (a) Recognised in profit or loss (e.g. a wage subsidy), which would result in the effect of the grant being recognised in the operating category because it does not relate to specified assets or liabilities that arise from transactions that involve only the raising of financing. (b) Recognised as a reduction of the cost of an associated asset (e.g. property, plant and equipment), which would result in the effect of the grant being recognised in the same category as the associated depreciation (e.g. operating for property, plant and equipment). (c) Recognised as deferred income and amortised into income on a systematic basis, which would result in the effect of the grant being recognised in the arms being recognised in the anortised into involve only the raising of finance (IFRS 18.59(b)) and the amortisation of income is not (a) interest income and expense or (b) income and expenses arising from changes in interest rates. The classification of other government grants (e.g. those relates to debt or investment properties) will depend on facts and circumstances

Item of income or expense	Classification	Explanation
Foreign exchange differences (IAS 21)	Depends on classification of related income and expenses	See section <u>3.9.3</u> .
Income and expenses related to associates and joint ventures (IAS 28, IFRS 10) – equity method is applied	Investing	If the associate or joint venture is accounted for using the equity method, income and expenses will always be classified in the investing category. See section <u>3.8.1</u> .
Income and expenses related to associates and joint ventures (IAS 28, IFRS 10) – equity method is not applied	Operating or investing	If the equity method is not applied, the classification will depend on whether the entity invests in associates and joint ventures as a main business activity. See section <u>3.8.1</u> .
Gain or loss on net monetary position (IAS 29)	Depends on approach adopted	Entities that apply IAS 29 may present the gain or loss on the net monetary position with other income and expense items associated with the net monetary position (e.g. interest income and expenses and foreign exchange differences). If an entity does not present the gain or loss on the net monetary position in this manner, it must be classified in the operating category (IFRS 18.B69).
Impairment of assets and reversals of impairment where applicable (IAS 36)	Generally operating	Most impairment losses will be recognised in the operating category as the impairment will relate to assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). For example, property, plant and equipment, intangible assets, goodwill, etc.
		In some cases, impairments recognised in accordance with IAS 36 may relate to assets that generate a return individually and largely independently of the entity's other resources (IFRS 18.53(c)) and are within the scope of IAS 36. For example, an investment property measured at cost that is impaired in accordance with IAS 36 would have the resulting impairment classified in the investing category (IFRS 18.B47(e)). If the entity invests in such assets as a main business activity, then the impairment would also be classified in the operating category (IFRS 18.58).
Provisions recognised as an expense (e.g. lawsuits, restructuring provisions) (IAS 37)	Operating	Liability arises from transactions that do not involve only the raising of finance (IFRS 18.59(b)) but such an expense is not (a) interest income and expenses identified for purposes of applying IFRS Accounting Standards; or (b) income and expenses arising from changes in interest rates. See section <u>3.3</u> .
Finance expense on provision liabilities (IAS 37)	Financing	Liability arises from transactions that do not involve only the raising of finance (IFRS 18.59(b)) and are interest income and expenses identified for purposes of applying IFRS Accounting Standards (IFRS 18.61(a)).
		Increases in the discounted amount of a provision due to the passage of time and effects of changes in discount rate, are classified in the financing category (IFRS 18. B54).
		See section <u>3.3</u> .

Item of income or expense	Classification	Explanation
Amortisation of intangible assets (IAS 38)	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). See section <u>3.4</u> .
		If an entity accounts for intangible assets using the revaluation model (e.g. certain intangibles with an active market, including some cryptocurrencies), fair value gains and losses are recognised in other comprehensive income rather than profit or loss, therefore, these classification requirements do not apply.
Depreciation of investment property accounted for using the cost model (IAS 40)	Operating or investing	Generally classified as investing, as these are assets that generate a return individually and largely independently of the entity's other resources (IFRS 18.53(c)) and income and expenses that arise from the subsequent measurement of that asset (IFRS 18.54(b)).
		If the entity invests in such assets as a main business activity, then deprecation would be classified in the operating category (IFRS 18.58).
Fair value gains and losses on investment property accounted for using the fair value model (IAS 40)	Operating or investing	Generally classified as investing, as these are assets that generate a return individually and largely independently of the entity's other resources (IFRS 18.53(c)) and income and expenses that arise from the subsequent measurement of that asset (IFRS 18.54(b)).
		If the entity invests in such assets as a main business activity, then fair value gains and losses would be classified in the operating category (IFRS 18.58).
Fair value gains and losses on biological assets and agricultural produce at the point of harvest (IAS 41)	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). See section <u>3.4</u> .
Share-based payments related to employee compensation (IFRS 2)	Operating	Classified in the operating category because the expense does not meet the criteria to be classified in any of the other four categories.
Share-based payments related to other services	Operating	Expense is not an income tax or related to a discontinued operation.
obtained (IFRS 2)		Expense is not classified in the investing category because it does not relate to investments in associates, joint ventures and unconsolidated subsidiaries, cash equivalents or other assets that generate a return individually and largely independently of the entity's other resources (IFRS 18.53).
		If the transaction is an equity-settled share-based payment, the expense is not classified in the financing category because it does not relate to a recognised liability (IFRS 18.59). If the expense is a cash-settled share-based payment transaction, it may arise from a recognised liability, however, that liability would not arise from transactions that involve only the raising of financing (IFRS 18.59(b)). The expense recognised on such a liability would not be interest income and expenses or income and expenses arising from changes in interest rates (IFRS 18.61), therefore, it must be classified in the operating category.

Item of income or expense	Classification	Explanation
Bargain purchase gain (IFRS 3)	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). IFRS 18.B49(f) requires income and expenses arising from business combinations to be classified in the operating category.
Gain or loss on contingent consideration recognised as a result of a business combination (IFRS 3)	Operating	
Remeasurement of associate or joint venture accounted for using the equity method upon step acquisition of a subsidiary (IFRS 3)	Investing	IFRS 18.B60(c) specifies that the remeasurement of an investment in an associate previously accounted for using the equity method on the step acquisition of a subsidiary is classified in the investing category. See section 3.2 .
Remeasurement of associate or joint venture <u>not</u> accounted for using the equity method upon step acquisition of a subsidiary (IFRS 3)	Operating or investing	If the acquirer does not account for the associate or joint venture using the equity method, the classification will depend on whether the entity invests in associates and joint ventures as a main business activity. See section <u>3.8.1</u> .
Results of discontinued operations (IFRS 5)	Discontinued operations	Income and expenses presented as a single line item in accordance with IFRS 5.33(a) are classified in the discontinued operations category. See section 3.6 .
Income and expenses related to cash and cash equivalents	Operating or investing	For entities without specified main business activities, income and expenses will be classified in the investing category. See section <u>3.9.1</u> .
Income and expenses on hybrid contracts containing a financial liability	Operating or financing	See section <u>3.9.2</u> .
Interest income on financial assets measured at amortised cost or fair value through other comprehensive income (IFRS 9)	Operating or investing	Classification will depend on whether the entity invests in financial assets as a main business activity. See section 3.2 and $3.8.1$.
Interest expense on financial liabilities measured at amortised cost (IFRS 9)	Operating or financing	 Classification will depend on the assessment of multiple factors, including whether: (a) The liability arises from transactions that involve only the raising of finance; (b) The entity provides financing to customers as a main business activity; (c) The financial liability relates to providing financing to customers; and (d) If (a)-(c) are all 'yes', certain accounting policy choices made by the entity. See sections <u>3.3</u> and <u>3.8.2</u> for explanation of classification of income and expenses arising from liabilities.
Gains and losses on investments in debt and equity instruments measured at fair value through profit or loss (IFRS 9)	Operating or investing	Classification will depend on whether the entity invests in financial assets as a main business activity. See section <u>3.2</u> and <u>3.8.1</u> .

Item of income or expense	Classification	Explanation
Gains and losses on derivatives measured at fair value through profit or loss (IFRS 9)	Depends on facts and circumstances	See section <u>3.9.4</u> .
Gains and losses on the derecognition of financial assets (IFRS 9)	Operating or investing	Classification will depend on whether the entity invests in financial assets as a main business activity. See section 3.2 and $3.8.1$.
Gains and losses on the derecognition of financial liabilities (IFRS 9)	Operating or financing	 Classification will depend on the assessment of multiple factors, including whether: (e) The liability arises from transactions that involve only the raising of finance; (f) The entity provides financing to customers as a main business activity; (g) The financial liability relates to providing financing to customers; and (h) If (a)-(c) are all 'yes', certain accounting policy choices made by the entity.
		See sections <u>3.3</u> and <u>3.8.2</u> for explanation of classification of income and expenses arising from liabilities.
Modification gains and losses on financial assets (IFRS 9)	Operating or investing	Classification will depend on whether the entity invests in financial assets as a main business activity. See section 3.2 and $3.8.1$.
Modification gains and losses on financial liabilities (IFRS 9)	Operating or financing	 Classification will depend on the assessment of multiple factors, including whether: (a) The liability arises from transactions that involve only the raising of finance; (b) The entity provides financing to customers as a main business activity; (c) The financial liability relates to providing financing to customers; and (d) If (a)-(c) are all 'yes', certain accounting policy choices made by the entity. See section 3.3 and 3.8.2 for explanation of classification
Expected credit losses on trade receivables (IFRS 9)	Operating	of income and expenses arising from liabilities. Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). See section <u>3.4</u> .
Expected credit losses on related party loans receivable (IFRS 9)	Operating or investing	Assets that generate a return individually and largely independently of the entity's other resources (IFRS 18.53(c)) and income and expenses that arise from the subsequent measurement of that asset (IFRS 18.54(b)), therefore, classified in the investing category for entities without specified main business activities. If the entity invests in such assets as a main business activity, then expected credit losses would be classified in the operating category (IFRS 18.58).
Fair value gains and losses on unconsolidated subsidiaries and investments in associates and joint ventures for investment entities (IFRS 10)	Operating	Investment entities would be expected to invest in such assets as main business activities, meaning that all income and expenses are recognised in the operating category (IFRS 18.58). See section <u>3.2.4</u> .

Item of income or expense	Classification	Explanation
Revenue from contracts with customers (IFRS 15)	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). See section <u>3.4</u> .
Finance income on contract assets (IFRS 15)	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). See section <u>3.4</u> .
Finance expense on contract liabilities (IFRS 15)	Financing	Liability arises from transactions that do not involve only the raising of finance (IFRS 18.59(b)) and are interest income and expenses identified for purposes of applying IFRS Accounting Standards (IFRS 18.61(a)). See section <u>3.3</u> .
Modification gains or losses on contracts (IFRS 15)	Operating	See section <u>3.3</u> .
Depreciation of right-of- use assets not classified as investment property (IFRS 16)	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). See section <u>3.4.</u>
Finance expense on lease liabilities (IFRS 16)	Financing	Liability arises from transactions that do not involve only the raising of finance (IFRS 18.59(b)) and are interest income and expenses identified for purposes of applying IFRS Accounting Standards (IFRS 18.61(a)). See section <u>3.3</u> .
Variable lease payments not included in the measurement of the lease liability recognised in profit or loss (e.g. rent based on a percentage of sales) (IFRS 16)	Operating	See <u>Example 3.9.2-3</u> .
Lease modification gains or losses recognised in profit or loss (IFRS 16)	Operating	See section <u>3.3</u> .
Operating lease income related to assets other than investment property (e.g. property, plant and equipment subject to an operating lease – from the perspective of the lessor) (IFRS 16)	Generally operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). Property, plant and equipment typically does not generate a return individually and largely independently of the entity's other resources, however, this may not always be the case. See section <u>3.4</u> .
Selling profit or loss on finance leases (i.e. the difference between the initial carrying amount of the net investment in the lease and the carrying amount of the asset derecognised) (IFRS 16)	Operating or investing	Finance leases result in the derecognition of the asset subject to the lease (e.g. an item of property, plant and equipment). IFRS 18.B60 requires any income and expenses on the derecognition of an asset to be classified in the same category as the income and expenses from the asset immediately before its derecognition.
		In many cases, this will result in operating classification of the selling profit or loss on the finance lease because the asset subject to the finance lease will have been considered an asset that an entity uses in combination to produce or supply goods or services that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). For example, property, plant and equipment.

Item of income or expense	Classification	Explanation
		The selling profit could be classified in the investing category if the asset being derecognised in the finance lease was considered an asset that generated a return individually and largely independently of the entity's other resources, and therefore, the income and expenses related to it were classified in the investing category prior to derecognition (IFRS 18.B47). For example, an investment property that is derecognised when it is leased under a finance lease. If an entity invests in such assets (e.g. investment property) as a main business activity, then the selling profit would be classified in the operating category (IFRS 18.55(b), IFRS 18.58).
Finance income on finance leases (IFRS 16)	Operating or investing	Finance income on the net investment in the lease is classified in the investing category unless the entity invests in such assets as a main business activity. That is because the net investment in the lease is an asset that generates a return individually and largely independently of the entity's other resources (IFRS 18.53(c)), and the finance income is income generated by that asset (IFRS 18.54(a)).
		If the entity invests in net investments in leases as a main business activity (e.g. the entity is a leasing company), then the finance income would be classified in the operating category (IFRS 18.55(b), IFRS 18.58).
Income and expense arising from insurance contracts (e.g. insurance revenue, insurance service expenses, insurance finance income, etc.) (IFRS 17)	Operating	See sections 3.3 and 7.5.

3.9.6. Interim financial reports prepared in accordance with IAS 34

If an entity applies IAS 34 *Interim Financial Reporting*, IFRS 18 will affect those reports, regardless of whether they are a complete set of financial statements or a set of condensed interim financial statements.

If an entity prepares a complete set of financial statements as part of its interim financial report, all of the requirements of IFRS 18 discussed in this publication apply.

If an entity prepares condensed interim financial statements (the more common approach), then IFRS 18 will also affect how an entity prepares those condensed interim financial statements. Once an entity has transitioned to IFRS 18 (i.e. past the year of transition where special requirements apply – see section 6.1), an entity will use headings and subtotals that were included in the most recent annual financial statements (IAS 34.10). For example, the manner in which an entity classifies its income and expenses and presents mandatory subtotals (e.g. operating profit) and additional subtotals should be consistent from the most recently published annual financial statements and condensed interim financial statements.

IFRS 18's requirements for aggregation and disaggregation also apply to interim financial statements (IAS 34.10).

Management-defined performance measures are not addressed in this publication; however, it should be emphasised that IAS 34.16A(m) will require entities to provide disclosures about these measures as required by IFRS 18.121-125 in interim financial reports.

The effect of IFRS 18 on an entity's condensed interim financial statements in the year of transition to IFRS 18 (e.g. quarters 1, 2 and 3 or half-year interim financial reports in 2027 for entities with calendar year-ends) is discussed in section 6.4.

4. Principles of aggregation and disaggregation



IFRS 18 sets out new principles for aggregating and disaggregating of information. These principles are applicable to not only items of income and expense, but assets, liabilities, cash flows, equity and reserves also (IFRS 18.41). For example, how similar assets are aggregated for presentation purposes in the statement of financial position, and whether an entity presents its operating expenses by function, nature or a mixed presentation model.

Similar items are aggregated for practical reasons, and also to meet the objective of IFRS 18.16, which is for the primary financial statements to provide a 'structured summary' of a reporting entity's assets, liabilities, equity, income, expenses and cash flows. For purposes of presentation, an 'item' comprises an individual item (e.g. a single asset) or any aggregation or disaggregation of such items (IFRS 18.41).

The following sections discusses the requirements of IFRS 18 applicable to where information is presented or disclosed (i.e. the primary financial statements or the notes) and how information is aggregated and disaggregated. Materiality (whether information is presented or disclosed) remains an overriding factor, which is considered in applying the requirements of IAS 8.

BDO comment – aggregation in IAS 1 vs. IFRS 18

IFRS 18 has more specific requirements applicable to the roles of the primary financial statements and notes, how an entity aggregates and disaggregates information more generally and the use of labelling. The extent to which entities must change their existing presentation in IAS 1 (other than the requirements applicable to classification of items of income and expenses and specified subtotals), will depend on an entity's precise facts and circumstances.

The adoption of IFRS 18 may result in entities revisiting decisions made about presentation and disclosure that had been made previously, which are not necessarily incompatible with the requirements of IFRS 18, but by changing presentation and disclosure practices, may improve the usefulness of information provided.

4.1. The roles of the primary financial statements and notes

Primary financial statements (e.g. statements of financial position, profit or loss, cash flows, etc.) – IFRS 18.16	Notes to the financial statements – IFRS 18.17
 To provide structured summaries of a reporting entity's recognised assets, liabilities, equity, income, expenses and cash flows, that are useful to users of financial statements for: (a) obtaining an understandable overview of the entity's recognised assets, liabilities, equity, income, expenses and cash flows; (b) making comparisons between entities, and between reporting periods for the same entity; and (c) identifying items or areas about which users of financial statements may wish to seek additional information in the notes. 	 To provide material information necessary: (a) to enable users of financial statements to understand the line items presented in the primary financial statements; and (b) to supplement the primary financial statements with additional information to achieve the objective of financial statements.

IFRS 18.16 defines the roles of the primary financial statements and IFRS 18.17 defines the role of the notes to the financial statements:

IAS 1 did not define the roles of these two components of financial statements.

These roles have been defined in order to assist entities in determining whether to include information in the primary financial statements or in the notes (IFRS 18.18). Based on the defined roles, information presented in the primary

financial statements (i.e. a structured summary) is more aggregated than information provided in the notes, which typically disaggregates information provided in the primary financial statements (IFRS 18.18(a)). For example, a single line item may be presented in the statement of financial position for 'financial assets measured at fair value through profit or loss', with that amount disaggregated in a corresponding note into different classes of financial assets.

In some cases, a specific IFRS Accounting Standard may specify disaggregation requirements for notes to financial statements (e.g. a disaggregation of items of property, plant and equipment by asset class in IAS 16), however, those presentation and disclosure requirements remain subject to materiality (IFRS 18.19). Entities are also required to consider whether to provide additional disclosures when compliance with specific requirements in IFRS Accounting Standards is insufficient to enable users of financial statements to understand the effect of transactions and other events and conditions on the entity's financial position and performance (IFRS 18.20).

4.2. Principles of aggregation and disaggregation

In determining how to aggregate and disaggregate information presented in the primary financial statements and disclosed in the notes to the financial statements, IFRS 18.41 requires an entity to:

Requirement	Example
(a) classify and aggregate assets, liabilities, equity, income, expenses or cash flows into items based on shared characteristics;	Presentation of financial assets in the statement of financial position by classification in IFRS 9 (e.g. amortised cost, fair value through profit or loss, fair value through OCI).
(b) disaggregate items based on characteristics that are not shared in the primary financial statements and in the notes;	If an entity presents expenses by nature, disaggregating operating expenses by the nature of those expenses (e.g. presenting wages and benefits separately from depreciation).
(c) aggregate or disaggregate items to present line items in the primary financial statements that fulfil the role of the primary financial statements in providing useful structured summaries;	Applying the aggregation and disaggregation principles in (a) and (b) in order to provide a useful structured summary of information in the primary financial statements.
(d) aggregate or disaggregate items to disclose information in the notes that fulfils the role of the notes in providing material information; and	For assets that are aggregated in the statement of financial position (e.g. financial assets), disclosing a disaggregation by asset class in the notes to the financial statements, considering the requirements of IFRS 18 and IFRS 7.
(e) ensure that aggregation and disaggregation in the financial statements do not obscure material information.	Considering materiality in assessing whether to aggregate otherwise similar items, such as an individually material expected credit loss recorded on a single financial asset vs. expected credit losses recorded on a portfolio of smaller financial assets.

In order to meet the requirements of IFRS 18.41, an entity must always disaggregate items whenever the resulting information is material (IFRS 18.42), however, this disaggregation need not always be in the primary financial statements. as Disaggregated information may be provided in the notes to the financial statements, however, this is not a 'free choice' by an entity. Entities must consider the roles of the financial statements in determining where to provide information about the material item.

Examples of income, expenses, assets, liabilities and items of equity that have sufficiently dissimilar characteristics to warrant separate presentation in the primary statements or disclosure in the notes:


Assets and liabilities - statement of financial position (IFRS 18.B111)	Income and expenses – statement of profit or loss (IFRS 18.B79)
 Property, plant and equipment disaggregated into classes (IAS 16); Receivables disaggregated into amounts receivable from trade customers, related parties, prepayments and other amounts; Inventories disaggregated, applying IAS 2, into items such as merchandise, production supplies, materials, work in progress and finished goods; Trade payables disaggregated, applying IAS 7, to provide separately the amounts of those payable that are part of supplier finance arrangements; Provisions disaggregated according to their nature, such as, provisions for employee benefits, decommissioning liabilities, or other items; and Equity capital and reserves disaggregated into various classes, such as paid-in capital, share premium and reserves. 	 Write-downs of inventories and reversals Impairment losses and reversals for property, plant and equipment Income and expenses from restructuring of an entity's activities and reversals Income and expenses from: Disposals of property, plant and equipment; Disposals of investments (e.g. equity accounted for associates); Litigation settlements; Reversals of provisions; and Non-recurring income and expenses.

4.2.1. Line items required by IFRS Accounting Standards

Certain IFRS Accounting Standards may require specific line items to be presented in primary financial statements (e.g. IFRS 18.75(b)(i) – interest revenue calculated using the effective interest method), however, this separate presentation is not required if doing so is not necessary for the statement to provide a useful structured summary (IFRS 18.23). For example, a corporate entity with immaterial amounts of interest revenue calculated using the effective interest rate method may conclude that presenting that line item separately in the statement of profit or loss is not necessary.

4.2.2. Additional line items

An entity is required to present additional line items (and subtotals – see <u>5.2</u>) if such presentation is necessary for a primary financial statement to provide a useful structured summary (IFRS 18.24). However, any additional line items (and subtotals) must (IFRS 18.24):

(a) comprise amounts recognised and measured in accordance with IFRS Accounting Standards;

- (b) be compatible with the statement structure of IFRS 18 (e.g. all categorisation requirements and mandatory subtotals);
- (c) be consistent from period to period; and
- (d) be displayed no more prominently than the totals and subtotals required by IFRS Accounting Standards.

4.2.3. Characteristics to use in assessing the need to aggregate and disaggregate information

Whether an entity is determining how to aggregate or disaggregate information in the statement of profit or loss to provide a useful structured summary of the entity's income and expenses or how to aggregate and disaggregate information more generally (e.g. line items and note disclosure), an entity is required to use judgement (IFRS 18.B78).

This judgement is based on an assessment of whether the items have characteristics that are shared (similar characteristics) or characteristics that are not shared (dissimilar characteristics). Such characteristics include:



Characteristics (IFRS 18.B78)	Example of items with dissimilar characteristics
Nature	Salaries and wages vs. depreciation of property, plant and equipment
Function (role) within the entity's business activities	Operating expenses classified into selling, finance, and research and development
Persistence (including the frequency of the item of income or expense or whether it is recurring or non- recurring)	Impairment of property, plant and equipment arising from poor sales vs. impairment of property, plant and equipment due to change in regulation requiring change of use of the asset
Measurement basis	Financial assets at fair value through profit or loss vs. amortised cost
Measurement uncertainty or outcome uncertainty (or other risks associated with an item)	Equity instruments at fair value through profit or loss (steady-state companies vs. start-ups)
Size	Salaries and wages, which are quantitatively material vs. advertising expenses, which are not quantitatively material
Geographic location or regulatory environment	Revenue in jurisdictions where the majority of income is earned vs. emerging/growing jurisdictions where an entity has recently expanded
Tax effects	If different tax rates apply to items of income or expense
Whether the income or expense arise on initial recognition of a transaction or event from a subsequent change in estimate relating to a transaction or event	'Day 1' gains on financial assets recognised in accordance with IFRS 9 vs. the release of deferred gains to profit or loss over time (IFRS 9.B5.1.2A)

4.2.4. Application of requirements to the statement of profit or loss

After applying the requirements to classify items of income and expense into the five categories (see section $\underline{3}$), an entity must determine how to present those items in the statement of profit or loss, and provide corresponding disclosures in the notes. This publication focuses on the aggregation and disaggregation requirements of IFRS 18 that are applicable to the statement of profit or loss. Therefore, the remaining topics in section $\underline{4}$ are written in the context of the statement of profit or loss.

4.3.Line items in statement of profit or loss

IFRS 18.75-76 requires certain line items to be presented in the statement of profit or loss (subject to materiality). The following table summarises these requirements, comparing them to the requirements of IAS 1:



Line items required to be presented in the statement of profit or loss (IFRS 18.75-76)	Required by IAS 1?
 Revenue, presenting separately: Interest revenue calculated using the effective interest method (IFRS 9); and Insurance revenue (IFRS 17) 	► Yes
Operating expenses, presenting separate line items as required by IFRS 18.78 and 82(a) – see section <u>4.4</u>	Yes, however, IAS 1 did not require this analysis of operating expenses to be provided in the statement of profit or loss; disclosure could be provided in the notes. IFRS 18 requires this analysis to be in the statement of profit or loss – see section <u>4.4</u>
Share of profit or loss of associates and joint ventures accounted for using the equity method (IAS 27, 28)	► Yes
Income tax expense or income (IAS 12)	► Yes
A single amount for the total of discontinued operations (IFRS 5)	► Yes
IFRS 9 amounts:	
 Interest revenue calculated using the effective interest method 	► Yes
 Impairment losses (including reversals of impairment losses or impairment gains); 	► Yes
 Gains and losses arising from the derecognition of financial assets measured at amortised cost; 	► Yes
 Reclassification gains and losses; and 	► Yes
Recycling of gains and losses from OCI to profit or loss.	► Yes
IFRS 17 amounts:	
 Insurance revenue; 	► Yes
 Insurance service expenses; 	► Yes
 Income or expenses from reinsurance contracts held; 	► Yes
 Insurance finance income or expenses; and 	► Yes
 Finance income or expenses from reinsurance contracts held. 	► Yes
An allocation of profit or loss for the reporting period attributable to:	
 Non-controlling interests; and 	► Yes
 Owners of the parents. 	► Yes

BDO comment - line items required by IAS 1 but not required by IFRS 18

IAS 1.82(b) requires 'finance costs' to be presented as a single line item. This requirement was not brought forward into IFRS 18, unlike the majority of the other line item disclosure requirements in IAS 1 (IFRS 18.BC242).

This is a consequence of the fact that IFRS 18 has a financing category for income and expenses (see section 3.3). An entity applies the requirements of IFRS 18 in determining how to aggregate and disaggregate information in the financing category into line items and whether disaggregation is required in the notes.

For example, an entity may determine that interest expenses on bank loans and lease liabilities classified in the financing category should be aggregated into a single 'interest expense' line item in the statement of profit or loss, with corresponding disclosure in the notes to disaggregate that line item, if the resulting information from disaggregation is considered material.

4.4.Analysis of the most useful structured summary of operating expenses – use of characteristics of nature and function

In the operating category, an entity shall classify and present expenses in line items in a way that provides the most useful structured summary of its expenses, using one or both of the following characteristics (IFRS 18.78): (a) The nature of expenses; or

(b) The function of the expenses within the entity.

This analysis of operating expenses must be presented in the statement of profit or loss (IFRS 18.78), whereas IAS 1 permitted the analysis to be provided in the notes to the financial statements (IAS 1.99-100). Entities may still determine that the line items presented for operating expenses in the statement of profit or loss require additional disaggregation in the notes.

An explanation of nature vs function is provided below:

	Nature	Function	'Mixed' presentation
Description	Allocation of expenses based on the nature of the economic resources consumed to accomplish the entity's activities without reference to the activities in relation to which the economic resources were consumed (IFRS 18.80).	Allocation of expenses based on the activity to which the consumed resources relate (IFRS 18.81).	An entity may present an analysis of operating expenses where some expenses are classified by nature and others by function (IFRS 18.B81).
Example	Salaries and wages, depreciation, inventories expensed	Cost of goods sold, research and development, administrative expenses	See <u>Example 4.4-2</u> below

BDO comment – 'mixed' basis of presentation - IAS 1 vs. IFRS 18

IAS 1 did not explicitly permit a mixed basis of presenting line items in the statement of profit or loss, however, many entities have applied this basis of presentation in the past, while providing disclosures required by IAS 1.104 – specific information about the nature of expenses for depreciation, amortisation and employee benefits. IFRS 18.83 expands this disclosure requirement – see section 4.4.3 'presentation of operating expenses by function' below.



4.4.1. Factors considered in assessing which presentation to use

An entity does not have a free accounting policy choice in determining the presentation basis of operating expenses. IFRS 18.B80 requires an entity to consider:

Factor to be considered	Example
(a) What line items provide the most useful information about the main components or drivers of the entity's profitability;	Use of cost of sales by a retail entity, as it provides relevant information about whether the revenue generated from the sale of goods covers what, for retailers, are mainly direct costs, and by what margin.
(b) What line items most closely represent the way the business is managed and how management reports internally;	An entity with multiple major functions may classify expenses by those functions, whereas an entity with a single function (e.g. a lender) may classify expenses by nature.
(c) What standard industry practice entails; and	If certain industries have common practice in how operating expenses are classified, using the industry accepted approach may make comparison easier between entities.
(d) Whether the allocation of particular expenses to function would be arbitrary to the extent that the line item presented would not provide a faithful representation of the function.	Allocating impairment of a large group of assets (e.g. property, plant and equipment, right-of-use assets, etc.) by function (e.g. research and development, administration, etc.) may be arbitrary as assets may be used in multiple functions.

4.4.2. Presentation of operating expenses by nature

If entities present operating expenses by nature, the amounts presented need not be the amounts recognised as an expense for the period (IFRS 18.B84). For example, a manufacturer may incur the following expenses during the year (amounts prior to a portion being recognised in the carrying amount of assets):

Wages: CU100

Electricity: CU80

Depreciation: CU45

A portion of these expenses will be included in the carrying amount of the entity's unsold finished goods inventories for the period. An entity could, for example, present in the statement of profit or loss as follows:

Revenue	CU300
Wages	(CU100)
Electricity	(CU80)
Depreciation	(CU45)
Change in carrying	
amount of inventories	<u>CU25</u> - (IFRS 18.B84(a))
Operating profit	CU100

4.4.3. Presentation of operating expenses by function

Entities that present operating expenses by function are subject to certain additional requirements.

If an entity presents one or more line items comprising expenses classified by function in the operating category of the statement of profit or loss, it is required to (IFRS 18.82):

(a) Present a separate line item for its cost of sales, if the entity classifies operating expenses in functions that include a cost of sales function. That line item shall include the total of inventory expense described in IAS 2.38.

(b) Disclose a qualitative description of the nature of the expenses included in each function line item.

The requirements of IFRS 18.82 also apply to an entity that uses a mixed presentation, as such a presentation will include 'one or more line item comprising expensed classified by function'.

Entities are also required to present a single note for all line items comprising expenses classified by function, disclosing (IFRS 18.83):

- (a) The total for each of:
 - i. Depreciation (IAS 16, IAS 40, IFRS 16)
 - ii. Amortisation (IAS 38)
 - iii. Employee benefits (IAS 19) and share-based payments relating to services received from employees (IFRS 2)
 - iv. Impairment losses and reversals (IAS 36)
 - v. Write-downs and reversals (IAS 2)

(b)For items (a)(i)-(v):

i. The amount related to each line item in the operating category; and

ii. A list of any line items outside the operating category that also include amounts related to the total. This allocation disclosure required by IFRS 18.83 is intended to enhance the understandability of the composition of the line items.

Example 4.4-1 – disclosure of expenses by function

Fact pattern

P

Entity H classifies all of its operating expenses by function in the statement of profit or loss. Certain expenses specified by IFRS 18.83 are also presented in categories other than the operating category (e.g. depreciation of investment property measured at cost, which is presented in the investing category as Entity H does not have main business activities of investing in investment properties).

Entity H must comply with the disclosure requirements of IFRS 18.83.

<u>Analysis</u>

Entity H presents the following note to meet the requirements of IFRS 18.83:

Line items presented in the statement of profit or loss	Depreciation of property, plant and equipment and investment property	Employee benefits	Totals
Operating category:			
 Cost of sales 	70	90	160
 Research and development 	35	45	80
 Administration 	<u>25</u>	<u>65</u>	90
Totals*	130	200	
Investing category:			
 Depreciation of investment property** 	100	-	100

* IFRS 18.83 requires the <u>totals for each</u> of the items noted in IFRS 18.83(a)(i)-(v) classified in the operating category to be disclosed.

** Depreciation of investment property is not presented by function because it is classified in the investing category. Only expenses classified in the operating category may be presented by function.





4.4.4. Presentation of operating expenses - mixed presentation

In considering the factors in IFRS 18.880, an entity may determine that a mixed presentation is most appropriate, where certain line items are presented by function (e.g. cost of sales) and others are presented by nature (e.g. expected credit losses on trade receivables, salaries and wages not included in cost of sales, etc.) (IFRS 18.881).

If an entity uses a mixed presentation, with some line items being comprised of expenses classified by nature and other line items classified by function, each line item must be aggregated on the basis of only one of those characteristics (nature or function) (IFRS 18.79). For example, an entity may determine that it should present 'cost of sales' (a function basis of presentation) while all other expenses are presented on the basis of their nature.

A mixed presentation also requires appropriate labelling to be used or else line items may be misleading. For example, if an entity presents cost of sales, with all other expenses classified on the basis of nature, then certain expenses will be classified in the cost of sales category (e.g. some salaries and depreciation expenses, such as salaries and depreciation included in the cost of inventories expensed) with others are presented outside of this line item (e.g. salaries of the finance department, depreciation of property, plant and equipment that is not allocated to cost of sales). Appropriate labelling must be used to ensure line items clearly identify the expenses included in each line item, such as (IFRS 18.B82):

- Cost of sales
- Salaries other than those included in cost of sales
- Depreciation other than those included in cost of sales

BDO comment - nature and function for income and expenses not classified in the operating category

The requirement to present operating expenses by nature, function or a mixed presentation applies only to items classified in the operating category, not the other four categories being: investing, financing, income taxes and discontinued operations.

Income and expenses classified into categories other than the operating category are presented by their nature.

Example 4.4-2 - operating expense presentation by nature vs. function vs. mixed

Fact pattern

D

Entity D has applied the classification requirements of IFRS 18 and classified the following expenses into the operating category (all amounts in thousands of CU):

Salaries and wages (IAS 19):	100
Depreciation (IAS 16):	150
Inventories expensed (IAS 2):	360
 Distribution expenses 	
(shipping goods to customers):	40
 Share-based payments 	
to employees (IFRS 2):	60
 Depreciation of 	
right-of-use assets (IFRS 16):	85

Entity D also earned CU1,000 of revenue.

The following is a summary of the operating category in the statement of profit loss, under three mutually exclusive scenarios:

- Scenario A: all expenses are presented by nature
- Scenario B: all expenses are presented by function
- Scenario C: cost of sales is presented (function) with all other expenses presented by nature

As noted earlier in this section, Entity D does not have a free choice of which presentation method to use, however, these scenarios are included to demonstrate the effect of the different presentation methods, as well as

judgements made by Entity D in determining how much information to present in the statement of profit or loss and the corresponding notes.

Scenario A: all expenses are presented by nature

Note that Scenario A does not elect to use the option in IFRS 18.884 to present operating expenses before the allocation of amounts to the carrying amount of assets (e.g. inventories). Therefore, inventories expensed represent the amount of inventories expensed in the period applying the requirements of IAS 2.

Line item	Amount (thousands of CU)
Revenue	1,000
Salaries and wages	(100)
Depreciation (Note X)	(235)
Inventories expensed*	(360)
Share-based payments to employees	(60)
Distribution expenses	<u>(40)</u>
Operating profit	205

Note X: depreciation consists of:

 Depreciation of property, 	
plant and equipment:	150
Depreciation of right-of-use assets:	<u>85</u>
▶ Total	235

*In this scenario, Entity D is presenting expenses by nature only, therefore, it is not required to disclose the portion of depreciation, salaries, etc. that are included in inventories expensed in the period, because those amounts were included in the carrying amount of inventories when they were derecognised. In other words, the disclosures requirement in IFRS 18.83 does not apply to Entity D in this scenario.

Scenario B: all expenses are presented by function

Line item	Amount (thousands of CU)
Revenue	1,000
Cost of goods sold*	<u>(400)</u>
Gross profit**	600
Research and development	(300)
Administration	<u>(95)</u>
Operating profit	205

*Cost of goods sold consists of inventoried expensed (360) plus 40 of distribution expenses.

**See section <u>5.2</u> for further information on the use of additional subtotals. Entities are not required to present the 'gross profit' subtotal; however, it may be common for many entities, especially those that present cost of goods sold.

In Scenario B, Entity D must comply with the disclosure requirements of IFRS 18.83, which results in a single note that provides disclosure of certain expenses by nature and their respective allocations to functional categories.

Scenario C: cost of sales is presented (function) with all other expenses presented by nature ('mixed' presentation)

Line item	Amount (thousands of CU)
Revenue	1,000
Cost of goods sold*	<u>(400)</u>
Gross profit	600
Salaries not included in cost of goods sold	(100)
Depreciation not included in cost of goods sold (Note X)	(235)
Share-based payments to employees	<u>(60)</u>
Operating profit	205

Note X: depreciation consists of:

- Depreciation of property, plant and equipment: 150
- Depreciation of right-of-use assets: 85 235
- Total

*Cost of goods sold is calculated in the same manner as Scenario B.

In Scenario C, Entity D must comply with the disclosure requirements of IFRS 18.83, which results in a single note that provides certain expenses by nature and their respective allocations to functional categories.

4.4.5. Consistency of presentation, disclosure and classification

Entities are required to maintain a consistent method of presentation from one period to the next, as the method selected is an accounting policy. Any change in the basis of presentation (e.g. from operating expenses being presented entirely by nature to a mixed presentation basis) must (IFRS 18.30):

- (a) Arise from a significant change in the nature of the entity's operations or arise from a review of its financial statements, and it is concluded that another presentation would be more appropriate considering the criteria in IAS 8; or
- (b) Arise from a requirement in an IFRS Accounting Standard which requires a change in presentation.

Any change in presentation basis (e.g. a change in judgement concerning aggregation of income and expenses) is accounted for as a change in accounting policy, therefore, comparative information must be restated.

4.5. Labels and descriptions used

Entities are required to describe items presented in the primary financial statements (line items and subtotals) or items disclosed in the notes in a way that faithfully represents the characteristics of the item (IFRS 18.43). Labels will often apply to items that have been aggregated (e.g. a 'salaries and wages' line item will generally comprise thousands of individual transactions).

An entity is required to provide all descriptions and explanations necessary for a user of financial statements to understand the item. In some cases, this may be achieved simply (e.g. utility expenses), however, in other cases, more judgement may be required.

4.5.1. Use of 'other'

Entities may only present or disclose items as 'other' if another more informative label cannot be found (IFRS 18.B25).

For example, if many immaterial items are aggregated with the label 'other expenses' being used in the statement of profit or loss, an entity may find a more informative label by aggregating items that share similar characteristics and describing them in a way that faithfully represents the similar characteristics.

If an entity uses the 'other' label because a more informative label cannot be found, additional requirements apply (IFRS 18.B26):

- (a) For any aggregation using the 'other' label: the entity shall use a label that describes the aggregated item as precisely as possible (e.g. 'other operating expenses' or 'other finance expenses' rather than 'other expenses').
- (b) For an aggregation of immaterial items, the entity shall consider whether the aggregated amount is sufficiently large that users might reasonably question whether it includes items for which information could be material (i.e. material items being obscured by immaterial items). In such cases, the entity is required to disclose further information, for example:
 - i. An explanation that no items for which information would be material are included in the amount; or
 - ii. An explanation that the amount comprises several items for which information would not be material, with an indication of the nature and amount of the largest item.

These additional requirements around the use of the 'other' label have been introduced to address user concerns that overuse of the label has obscured useful information in financial statements.

4.5.2. Labelling of subtotals

Requirements applicable to the labelling of mandatory and additional subtotals presented in the statement of profit or loss are discussed in section <u>5</u>.



5. Totals and subtotals presented in the statement of profit or loss



Once an entity has classified individual items of income and expense into the appropriate categories (section $\underline{3}$), and it has aggregated those items into appropriate levels of aggregation for presentation in the statement of profit or loss (section $\underline{4}$), mandatory and additional subtotals are presented as a result of the previous steps.

For example, for an entity without specified main business activities (see section 3.8), a typical statement of profit or loss may be as follows (a mixed presentation method for operating expenses has been used – see section 4.4):

Line item		Classification
Revenue	XXX	
Cost of sales	XXX	
Gross profit	XXX	
Other operating income	XXX	Operating category
Selling expenses	XXX	
Research and development	XXX	
General and administrative	XXX	
Operating profit	XXX	Mandatory specified subtotal
Fair value gains on investments in equity instruments	XXX	Investing category
Profit before financing and income taxes	XXX	Mandatory specified subtotal
Interest expense on borrowings and lease liabilities	XXX	Financing category
Profit before income taxes	XXX	Additional subtotal
Income tax expense	XXX	Income taxes category
Profit from continuing operations	XXX	Additional subtotal
Loss from discontinued operations	XXX	Discontinued operations category
Profit	XXX	Mandatory total

As noted in this example, certain line items are mandatory, and others are additional (i.e. those that are not mandatory but meet the criteria in IFRS 18 to be disclosed in addition to mandatory subtotals). Sections <u>5.1</u> and <u>5.2</u> discuss the requirements of IFRS 18 applicable to mandatory and additional subtotals in the statement of profit or loss.

5.1. Mandatory subtotals

Mandatory subtotals are based on how the entity has classified its income and expenses into the five categories. The categorisation of expenses into these categories (and the assessment of whether the entity has specified main business activities) is necessary as it affects how subtotals are calculated.

IFRS 18 only requires two mandatory subtotals be presented before profit or loss (IFRS 18.69):

Mandatory subtotal – introduced by IFRS 18	Explanation
Operating profit	The total of all income and expenses classified in the operating category.
Profit or loss before financing and income taxes*	The total of operating profit or loss and all income and expenses classified in the investing category.

*Certain entities are not permitted to present this subtotal; see section 5.1.3 below.



Mandatory subtotal – consistent with IAS 1	Explanation
Profit or loss	The total of income less expenses included in the statement of profit or loss.
Total other comprehensive income	The total of amounts recorded in other comprehensive income.
Total comprehensive income	The total of profit or loss and other comprehensive income.

There is no requirement in IFRS 18 for an entity to present:

- Profit or loss before income taxes (all income and expenses other than those classified in the income taxes category, and the discontinued operations category if applicable); or
- Profit or loss from continuing operations (all income and expenses other than those classified in the discontinued operations category).

However, it is expected that many entities will consider these additional subtotals to be useful, and will therefore present them as additional subtotals (see section 5.2).

IFRS 18 also does not require any mandatory subtotal related to the investing category (e.g. total investing income).

5.1.1. Operating profit

As operating profit includes all income and expenses classified in the operating category, operating profit is a default or residual subtotal, meaning that it includes all income and expenses of the entity unless they are classified into another category.

5.1.2. Profit or loss before financing and income taxes

As profit or loss before financing and income taxes is defined as the total of operating profit or loss and all income and expenses classified in the investing category, this subtotal is essentially all income and expenses before income and expenses classified in the remaining three categories: (1) financing; (2) income taxes; and (3) discontinued operations.

The IASB decided to require this subtotal because many users of financial statements seek to analyse an entity's performance independently of how the entity is financed (IFRS 18.BC148). For example, an entity may earn significant operating profits, but due to significant borrowing costs, it may be unprofitable overall. Separating the cost of financing from the operating profits provides useful information to users, including the ability to compare the operations of two or more entities independent of the cost of financing those entities.

In certain cases, entities do not present the profit or loss before financing and income taxes subtotal; see 5.1.3 below.

5.1.3. Profit or loss before financing and income taxes – entities that provide financing to customers as a main business activity

As discussed in sections <u>3.8.2</u> and <u>3.9.4</u>, for entities that provide financing to customers as a main business activity, entities have an accounting policy choice for how income and expenses related to liabilities that arise from transactions that involve only the raising of financing but are not related to providing financing to customers are presented. This choice also affects how gains and losses on certain derivatives and income and expenses relating to cash and cash equivalents are classified (see sections <u>3.9.4</u> and <u>3.9.1</u>).

For entities that elect to classify such income and expenses in the operating category, IFRS 18.73 does not permit an entity to present the profit or loss before financing and income taxes subtotal. This is not an optional exemption – in such cases, IFRS 18.73 forbids the profit or loss before financing and income taxes subtotal from being presented.

However, in this case, entities shall determine whether to present an additional subtotal (see section <u>5.2</u>) after operating profit and before the financing category. However, if an entity presents an additional subtotal in this manner (e.g. the total of operating profit or loss and all income and expenses classified in the investing category), the entity shall not label the subtotal in a way that implies the subtotal excludes financing amounts (e.g. profit before financing) (IFRS 18.74).

5.2. Additional subtotals

IFRS 18 requires entities to present additional line items (including subtotals) if such presentation is necessary for a primary financial statement to provide a useful structured summary. However, any additional line items (and subtotals) must (IFRS 18.24):

- (a) comprise amounts recognised and measured in accordance with IFRS Accounting Standards;
- (b) be compatible with the statement structure described in section 4.1;
- (c) be consistent from period to period; and
- (d) be displayed no more prominently than the totals and subtotals required by IFRS Accounting Standards.

IFRS 18.24 is not optional; it is a requirement for an entity to consider whether additional subtotals should be presented in order for the statement of profit or loss (and all primary financial statements) to provide a useful structured summary.

Additional subtotals may be presented 'within' other subtotals, including those required by IFRS 18. For example, operating profit or loss comprises the total of all income and expenses classified in the operating category, however, entities may also present subtotals such as 'gross profit', 'net interest income', or 'net interest margin', as long as those subtotals meet the requirements of IFRS 18.24. Such subtotals may be common in certain industries.

5.2.1. Additional subtotals and management-defined performance measures

IFRS 18 requires entities to disclose information about management-defined performance measures (MPMs), which are a subtotal of income and expense that (Appendix A of IFRS 18):

- (a) an entity uses in public communications outside financial statements;
- (b) an entity uses to communicate to users of financial statements management's view of an aspect of the financial performance of the entity as a whole; and
- (c) is not listed in IFRS 18.118, or specifically required to be presented or disclosed by IFRS Accounting Standards.

A common example of an MPM could be 'adjusted profit', which excludes share-based payments and impairment of goodwill. In this case, adjusted profit is a subtotal of income and expenses because it comprises only some of the items of income and expenses required to be included in profit or loss by IFRS Accounting Standards.

Unless IFRS 18 exempts an MPM (IFRS 18.118), significant disclosures about MPMs must be included in financial statements.

While this publication does not focus on the requirements of IFRS 18 applicable to MPMs, because MPMs are subtotals of income and expense, it is noteworthy which items of income and expense are deemed to not be MPMs by IFRS 18.118 (exempting them from associated disclosures), despite the fact that they meet the definition of MPMs:

- (a) gross profit or loss (revenue minus cost of sales) and similar subtotals (see below);
- (b) operating profit or loss before depreciation, amortisation and impairments within the scope of IAS 36;
- (c) operating profit or loss and income and expenses from all investments accounted for using the equity method;
- (d) for an entity that applies IFRS 18.73 (see sections <u>3.8.2</u>, <u>3.9.4</u> and <u>5.1</u>) and therefore does not present profit or loss before financing and income taxes, a subtotal comprising operating profit or loss and all income and expenses classified in the investing category;
- (e) profit or loss before income taxes; and
- (f) profit or loss from continuing operations.

'Similar subtotals' to gross profit or loss (IFRS 18.118(a)) depict the difference between a type of revenue and directly related expenses incurred in generating that revenue. Examples include (IFRS 18.8123):

- (a) net interest income;
- (b) net fee and commission income;

(c) insurance service result;
 (d) net financial result (investment income minus insurance finance income and expenses); and
 (e) net rental income.

BDO comment – earnings before interest, taxes, depreciation and amortisation (EBITDA)

The IASB considered whether to directly define EBITDA, which is a commonly used measure in communication to users of financial statements, and no consensus exists on how to calculate the amount (IFRS 18.BC363).

IFRS 18.118(b) exempts 'operating profit or loss before depreciation, amortisation and impairments within the scope of IAS 36' from the MPM requirements of IFRS 18, and this subtotal is similar to EBITDA in that it is comprised of operating profit (and therefore excludes income taxes), and adjusts out depreciation, amortisation and certain impairment expenses.

However, the IASB decided not to directly define EBITDA as the subtotal in IFRS 18.118(b) because doing so could imply that operating profit or loss is the same as what users of financial statements understand 'earnings before interest and tax' to be, which might not be the case (IFRS 18.BC365).

It is possible that in some cases, 'EBITDA' could be equal to the subtotal specified in IFRS 18.118(b). For example, if an entity has no income and expenses classified in the investing category and the only impairments being excluded in that subtotal are impairments within the scope of IAS 36, it may be possible for the 'earnings before interest, taxes, depreciation and amortisation' subtotal to be presented as an additional subtotal in accordance with IFRS 18.24.

BDO comment – additional subtotals and MPMs

If not for the exemption provided by IFRS 18.118 and B123, many common additional subtotals, such as gross profit or loss, profit or loss before income taxes, etc. would technically meet the requirements of IFRS 18 to be disclosed as MPMs.



6. Effective date and transition requirements



6.1. Required transition approach

IFRS 18 is effective for annual reporting periods beginning on or after 1 January 2027 (IFRS 18.C1). Earlier application is permitted, however, early application in many jurisdictions may be subject to endorsement of IFRS 18 (e.g. endorsement by the European Union for entities that comply with IFRS as endorsed by the European Union).

For example, entities with the following annual reporting period ends would be required to apply IFRS 18 in annual financial statements as follows:

Annual reporting period ending	IFRS 18 mandatorily effective for the first time in annual reporting period ending:
31 March	31 March 2028
30 June	30 June 2028
30 September	30 September 2028
31 December	31 December 2027

Specific requirements apply to interim financial reports – see section 6.4.

When an entity applies IFRS 18, it must do so retrospectively. Retrospective application means that an entity must restate all comparative periods presented to comply with the requirements of IFRS 18. Adopting IFRS 18 for the first time will primarily affect the statement of profit or loss, however, it is possible that the application of IFRS 18 may also affect other primary financial statements along with notes to the financial statements if, for example, entities change the level of aggregation in financial statements.

However, an entity is not required to present the quantitative information specified in IAS 8.28(f) (amount of the adjustment for each line item affected and if IAS 33 is applicable, the amount of the adjustment for basic and diluted earnings per share) (IFRS 18.C2). Instead, an entity applies the 'IFRS 18-specific' transition disclosures.

BDO comment – effect of retrospective application

The work effort involved in transitioning to IFRS 18 should not be underestimated simply because IFRS 18 does not modify the measurement requirements of IFRS Accounting Standards, unlike IFRS 9, 15, 16 or 17.

Depending on the level of complexity of an entity, the time and effort involved in transitioning to the requirements of IFRS 18 may vary significantly. For entities with complex operations, particularly large groups, systems and processes will need to be developed around the following areas (this list is not exhaustive):

- Classifying all items of income and expense into the five categories, including assessing the effect of the more complex areas of this process (e.g. assessing whether an entity has specified main business activities, specific requirements applicable to hybrid contracts, derivatives, etc.);
- Assessing the impact of the new requirements of IFRS 18 applicable to the presentation of foreign exchange differences (see section <u>3.9.3</u>), as systems may not capture foreign exchange by individual assets and liabilities;
- Determining the appropriate level of aggregation and disaggregation in the primary financial statements and the notes to the financial statements;
- Making accounting policy choices permitted by IFRS 18 where applicable;
- Assessing the effect of other requirements modified by IFRS 18 (e.g. management-defined performance measures, consequential amendments to earnings per share (IAS 33), the statement of cash flows (IAS 7)).

As IFRS 18 requires retrospective application, entities will need to ensure they have developed the appropriate processes to present their financial statements under the 'old requirements' (IAS 1), and then restate those amounts under the 'new requirements' (IFRS 18 and consequential amendments) for the period that will be

restated as comparatives when IFRS 18 is applied for the first time (e.g. the year-ended 31 December 2026 presented as comparatives in the 31 December 2027 financial statements).

6.2. Disclosures required upon transition

When IFRS 18 is applied for the first time, entities are required to disclose for the comparative period immediately preceding the period in which IFRS 18 is first applied (e.g. the 31 December 2026 comparative for 31 December 2027 financial statements), a reconciliation between (IFRS 18.C3): (a) The restated amounts presented applying IFRS 18; and

(b) The amounts previously presented applying IAS 1.

For example, if an entity previously presented certain line items outside of how the entity defined 'operating profit' under IAS 1, but those amounts are required to be classified in the operating category in accordance with IFRS 18, the entity would need to reconcile the line items affected.

The reconciliation required by IFRS 18.C3 is permitted, but not required, to be provided for the current period or earlier comparative periods (IFRS 18.C6).

6.3. Change in election for associates and joint ventures (IAS 28)

IAS 28.18 permits an accounting policy choice to be made at the initial recognition of each associate or joint venture for certain entities. This accounting policy choice is available for venture capital organisations, mutual funds, unit trusts and similar entities including investment-linked insurance funds. The accounting policy choice permits an entity to elect to measure an investment in an associate or joint venture at fair value through profit or loss in accordance with IFRS 9 rather than to apply the equity method in accordance with IAS 28. This accounting policy choice is available for each associate or joint venture separately.

At the date of initial application of IFRS 18 (e.g. 1 January 2026, the first date of the comparative period for entities with calendar year-ends that provide only a single comparative period), an entity is permitted to change this previous election if the change is from the equity method to fair value through profit or loss in accordance with IFRS 9 (IFRS 18.C7). An entity is not permitted to change its election in the opposite manner (i.e. from fair value through profit or loss in accordance with IFRS 9 to the equity method). If an entity prepares separate financial statements, it must make a corresponding change (IAS 27.11) in those separate financial statements. This change in policy is reflected retrospectively applying IAS 8.

BDO comment - reason for 're-election' being available on transition to IFRS 18

The reason the IASB permitted an entity to change its accounting policy from equity method to fair value through profit or loss when that choice is normally only available on the initial recognition of the investment is because IFRS 18 may change how an entity presents income and expenses associated with equity method investments.

For example, if a venture capital organisation previously elected to use the equity method prior to IFRS 18 being adopted, without the ability to change this election on adoption of IFRS 18, the venture capital organisation would be required to present income and expense associated with the associate or joint venture in the investing category. That is because income and expenses associated with associates or joint ventures accounted for using the equity method are always classified in the investing category (IFRS 18.BC131-132). This may not produce useful information for users of financial statements for venture capital organisations (and similar entities) since investing in associates and joint ventures may be part of the entities' main business activities.

Allowing the entity to change this election and measure associates and joint ventures at fair value through profit or loss in accordance with IFRS 9 will change the classification of income and expenses from investing to operating if the entity invests in such assets as a main business activity (see section <u>3.8.1</u>).



6.4.Interim financial reports

The general effect of IFRS 18 on interim financial reports is discussed in section <u>3.9.6</u>, however, specific requirements apply for condensed interim financial statements in the first year of applying IFRS 18 (e.g. quarters 1, 2 and 3 or half-year condensed interim financial statements in 2027 for entities with an annual period commencing on or after 1 January 2027).

These transition requirements are necessary because IAS 34.10 requires an entity to include 'each of the headings and subtotals that were included in its most recent annual financial statements'. This would imply that for an entity's quarter 1 condensed interim financial statements as at 31 March 2027, they should use headings and subtotals included in the entities 31 December 2026 annual financial statements, which would be prepared in accordance with IAS 1, not IFRS 18.

However, IFRS 18.C4 modifies this requirement for condensed interim financial statements in the first year of applying IFRS 18. IFRS 18.C4 requires entities to present in condensed interim financial statements each heading it <u>expects to</u> <u>use</u> in applying IFRS 18 and the subtotals required by IFRS 18, despite IAS 34.10's requirement. Effectively, IFRS 18.C4 'overrides' the requirement in IAS 34.10.

The 'default' requirement in IAS 34.10 to use the same heading and subtotals as the most recent annual financial statements does not apply again until an entity has issued its first annual financial statements prepared in accordance with IFRS 18.

Example 6.1-1 - condensed interim financial statements (IFRS 18.C4 and IAS 34.10)

Entity W has a calendar year-end and prepared its 31 December 2026 annual financial statements in accordance with the requirements of IAS 1 (i.e. it is not an early adopter of IFRS 18).

Entity W prepares condensed interim financial statements for the half-year ended 30 June each year.

In applying IAS 1, Entity W excluded from operating profit certain items of income and expenses that will be required to be classified in the operating category in accordance with IFRS 18 (the affected items of income and expense).

Entity W is considering on what basis it prepares its annual and condensed interim financial statements.

<u>Analysis</u>

P

	31 December 2026	30 June 2027	31 December 2027	30 June 2028
	annual financial	condensed interim	annual financial	condensed interim
	statements	financial statements	statements	financial statements
Analysis of affected items of income and expenses and applicable subtotals (e.g. operating profit)	Affected items of income and expense are excluded from operating profit as IFRS 18 is not yet applicable.	IFRS 18.C4 applies. Affected items of income and expense are presented using headings that Entity W expects to use in applying IFRS 18 in its 31 December 2027 financial statements and presents subtotals in accordance with IFRS 18. This would include presenting affected items of income and expense in the operating category.	Affected items of income and expense are presented in the operating category as required by IFRS 18.	IAS 34.10 applies. Entity W prepares its condensed interim financial statements using each of the headings and subtotals that were included in its most recent annual financial statements (31 December 2027).



6.4.1. Disclosures required upon transition – interim financial statements

For interim financial statements prepared in the first year of applying IFRS 18, as part of the disclosures required by IAS 34.16A(a) (information about accounting policies and changes therein), entities are required to disclose reconciliations for each line item presented in the statement of profit or loss for the comparative periods immediately preceding the current and cumulative current periods. The reconciliations are required between (IFRS 18.C5):

- (a) the restated amounts presented applying the accounting policies for the comparative period and the cumulative comparative period when the entity applies IFRS 18; and
- (b) the amounts previously presented applying the accounting policies for the comparative period and cumulative comparative period when the entity applied IAS 1.'

This reconciliation is permitted, but not required, to be provided for the current period or earlier comparative periods (IFRS 18.C6).

Example 6.1-2 – condensed interim financial statements and reconciliation disclosure required (IFRS 18.C5)

Entity Y has a calendar year-end and prepared its 31 December 2026 annual financial statements in accordance with the requirements of IAS 1 (i.e. it is not an early adopter of IFRS 18).

Entity Y prepares condensed interim financial statements for each quarter (Q1 - 31 March, Q2 - 30 June, Q3 - 30 September). In each condensed interim financial statement, the statement of profit or loss includes the current interim period and the cumulative interim period for the current financial year to date in accordance with IAS 34.20(b). For example, the Q3 30 September 2027 condensed interim financial statements include a statement of profit or loss for:

- The 3 months ended 30 September 2027 (the current interim period); and
- ▶ The 9 months ended 30 September 2027 (the cumulative interim period).

Comparative information for the immediately preceding financial year:

- ▶ The 3 months ended 30 September 2026 (the prior year comparative interim period); and
- ▶ The 9 months ended 30 September 2026 (the prior year cumulative interim period).

Therefore, Entity Y presents 4 statements of profit or loss in its 30 September 2027 condensed interim financial statements.

In applying IAS 1, Entity Y excluded from operating profit certain items of income and expenses that will be required to be classified in the operating category in accordance with IFRS 18 (the affected items of income and expense).

Entity Y is considering how it complies with the reconciliation disclosure required by IFRS 18.C5 in its 30 September 2027 condensed interim financial statements.

Analysis

To comply with IFRS 18.C5, Entity Y must reconcile both of its 3 months and 9 months ended statements of profit or loss related to the comparative period (30 September 2026). IFRS 18.C5 requires this reconciliation for 'each line item presented in the statement of profit or loss for the comparative periods immediately preceding the current and cumulative current periods'.

Therefore, Entity Y must explain how its line items have changed in the comparative periods as a result of applying IFRS 18 retrospectively. For the 3 months and 9 months ended statements of profit or loss for the periods ended 30 September 2026, Entity Y must reconcile the restated line items (e.g. the affected items of income and expense) to their revised presentation under IFRS 18 and how they were presented in previous condensed interim financial statements in accordance with IAS 1.

7. Examples of statements of profit or loss prepared in accordance with IFRS 18



The following examples use a common set of income and expenses and demonstrate how they would be presented under the following mutually exclusive scenarios:

- Scenario A: Entity S does not have any specified main business activities
- Scenario B: Entity S invests in financial assets as a main business activity
- Scenario C: Entity S provides financing to customers as a main business activity
- Scenario D: Entity S provides financing to customers as a main business activity and invests in investment property measured at fair value as a main business activity

These examples also include Scenario E, which is independent of the previous scenarios and illustrates how an insurance company that invests in financial assets as a main business activity may present its statement of profit or loss along with the line items and subtotals required by IFRS 17.

Common fact pattern for Examples A – D

Entity S has the following income and expenses:

- Revenue from contracts with customers
- Costs of goods sold
- Employee benefits not included in costs of goods sold
- Depreciation not included in costs of goods sold
- Expected credit losses on trade receivables
- Share of profit of associates and joint ventures (accounted for using the equity method)
- ► Fair value gains on equity instruments
- ▶ Fair value gains on investment property
- Interest expense on bank loans
- Interest income on contract assets
- Income tax expense
- Loss from discontinued operations

Entity S has applied the aggregation and disaggregation requirements of IFRS 18 and has determined that each of these classes of income and expense should be presented separately in the statement of profit or loss.

Entity S presents operating expenses by nature, except for the cost of goods sold line item, which is presented using a mixed presentation.



7.1. Example A – entity that does not have any specified main business activities

Scenario A: Entity S does not have any specified main business activities

Income or expense Classification Explanation Revenue from contracts with Operating customers Costs of goods sold Operating Employee benefits not included in Income and expenses from assets that do not generate a Operating costs of goods sold return individually and largely independently of the entity's other resources (IFRS 18.B48). See section 3.4. Depreciation not included in costs Operating of goods sold Expected credit losses on trade Operating receivables Share of profit of associates and Share of profit or loss from associates and joint ventures Investing joint ventures (accounted for accounted for using the equity method is also classified in the using the equity method) investing category. See section 3.2. Fair value gains on equity Financial assets and investment property generate a return Investing instruments individually and largely independently of the entity's other resources (IFRS 18.53(c)), however, Entity S does not invest in Fair value gains on investment Investing those assets as main business activities, therefore, income and property expenses are classified in the investing category. See sections 3.2 and 3.8.1. Bank loans arise from transactions that involve only the Interest expense on bank loans Financing raising of finance (IFRS 18.59(a)), and interest expense arises from the subsequent measurement of the liability (IFRS 18.60). As Entity S does not provide financing to customers as a main business activity, interest expense is classified in the financing category. See sections 3.3 and 3.8.2. Interest income on contract Operating Income and expenses from assets that do not generate a assets return individually and largely independently of the entity's other resources (IFRS 18.B48). See section 3.4. Income tax expense Income taxes Income and expenses within the scope of IAS 12 are classified in the income taxes category. See section 3.5. Loss from discontinued Discontinued Income and expenses presented as a single line item in accordance with IFRS 5.33(a) are classified in the discontinued operations operations operations category. See section 3.6.

The income and expenses described in the common fact pattern are classified as follows:

Entity S considers the requirement in IFRS 18.24 to present additional subtotals beyond those that are mandatory (see section <u>5.1</u>) and concludes that it will present three additional subtotals:

Additional subtotal	Description
Gross profit	The difference between revenue and cost of goods sold.
Profit or loss before income taxes	All income and expenses other than those classified in the income taxes and discontinued operations categories.
Profit or loss from continuing operations	All income and expenses other than those classified in the discontinued operations category.

Entity S presents its statement of profit or loss for the year-ended 31 December 20x7 as follows

	31 December 20x7	31 December 20x6	Explanation
Revenue	xxx	xxx	
Cost of goods sold	xxx	xxx	
Gross profit	XXX	XXX	Additional subtotal: revenue less cost of goods sold
Employee benefits not included in costs of goods sold	XXX	XXX	
Depreciation not included in costs of goods sold	XXX	XXX	
Expected credit losses on trade receivables	xxx	ххх	
Interest income on contract assets	XXX	XXX	
Operating profit	xxx	XXX	Mandatory subtotal: The total of all income and expenses classified in the operating category.
Share of profit of associates and joint ventures	xxx	ххх	
Fair value gains on equity instruments	XXX	ххх	
Fair value gains on investment property	XXX	xxx	
Profit before financing and income taxes	xxx	ххх	Mandatory subtotal: The total of operating profit or loss and all income and expenses classified in the investing category.
Interest expense on bank loans	ххх	xxx	
Profit before income taxes	XXX	XXX	Additional subtotal: all income and expenses other than those classified in the income taxes and discontinued operations categories.
Income tax expense	xxx	xxx	
Profit from continuing operations	XXX	XXX	Additional subtotal: all income and expenses other than those classified in the discontinued operations category.
Loss from discontinued operations	XXX	XXX	
Profit	xxx	xxx	Mandatory total

7.2. Example B – entity that invests in financial assets as a main business activity

Scenario B: Entity S invests in financial assets as a main business activity

Differences from Example A are in <u>red underlined text</u>.

The income and expenses described in the common fact pattern are classified as follows:

Income or expense	Classification	Explanation
Revenue from contracts with customers	Operating	
Costs of goods sold	Operating	
Employee benefits not included in costs of goods sold	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's
Depreciation not included in costs of goods sold	Operating	other resources (IFRS 18.B48). See section <u>3.4</u> .
Expected credit losses on trade receivables	Operating	
Share of profit of associates and joint ventures (accounted for using the equity method)	Investing	Share of profit or loss from associates and joint ventures accounted for using the equity method is also classified in the investing category. See section <u>3.2</u> .
Fair value gains on equity instruments	<u>Operating</u>	Financial assets such as equity instrument generate a return individually and largely independently of the entity's other resources (IFRS 18.53(c)). Entity S invests in financial assets as a main business activity; therefore, income and expenses are classified in the operating category (IFRS 18.58). See sections 3.3 and 3.8.1.
Fair value gains on investment property	Investing	Investment property generates a return individually and largely independently of the entity's other resources (IFRS 18.53(c)), however, Entity S does not invest in investment property as main business activity, therefore, income and expenses are classified in the investing category. See sections <u>3.3</u> and <u>3.8.1</u> .
Interest expense on bank loans	Financing	Bank loans arise from transactions that involve only the raising of finance (IFRS 18.59(a)), and interest expense arises from the subsequent measurement of the liability (IFRS 18.60). As Entity S does not provide financing to customers as a main business activity, interest expense is classified in the financing category. See sections <u>3.3</u> and <u>3.8.2</u> .
Interest income on contract assets	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). See section <u>3.4</u> .
Income tax expense	Income taxes	Income and expenses within the scope of IAS 12 are classified in the income taxes category. See section <u>3.5</u> .
Loss from discontinued operations	Discontinued operations	Income and expenses presented as a single line item in accordance with IFRS $5.33(a)$ are classified in the discontinued operations category. See section <u>3.6</u> .

Entity S considers the requirement in IFRS 18.24 to present additional subtotals beyond those that are mandatory (see section 5.2) and concludes that it will present three additional subtotals:

Additional subtotal	Description
Gross profit	The difference between revenue and cost of goods sold.
Profit or loss before income taxes	All income and expenses other than those classified in the income taxes and discontinued operations categories.
Profit or loss from continuing operations	All income and expenses other than those classified in the discontinued operations category.

Entity S presents its statement of profit or loss for the year-ended 31 December 20x7 as follows.

	31 December 20x7	31 December 20x6	Explanation
Revenue	xxx	xxx	
Cost of goods sold	XXX	xxx	
Gross profit	XXX	XXX	Additional subtotal: revenue less cost of goods sold
Employee benefits not included in costs of goods sold	XXX	XXX	
Depreciation not included in costs of goods sold	xxx	ХХХ	
Expected credit losses on trade receivables	xxx	XXX	
Interest income on contract assets	xxx	XXX	
Fair value gains on equity instruments	xxx	XXX	
Operating profit	XXX	ххх	Mandatory subtotal: The total of all income and expenses classified in the operating category.
Share of profit of associates and joint ventures	ххх	XXX	
Fair value gains on investment property	xxx	XXX	
Profit before financing and income taxes	XXX	ххх	Mandatory subtotal: The total of operating profit or loss and all income and expenses classified in the investing category.
Interest expense on bank loans	xxx	xxx	
Profit before income taxes	XXX	ХХХ	Additional subtotal: all income and expenses other than those classified in the income taxes and discontinued operations categories.
Income tax expense	XXX	xxx	
Profit from continuing operations	XXX	ххх	Additional subtotal: all income and expenses other than those classified in the discontinued operations category.
Loss from discontinued operations	xxx	XXX	
Profit	XXX	xxx	Mandatory total

7.3. Example C – entity that provides financing to customers as a main business activity

Scenario C: Entity S provides financing to customers as a main business activity

Differences from Example A are in red underlined text.

The income and expenses described in the common fact pattern are classified as follows:

Income or expense	Classification	Explanation
Revenue from contracts with customers	Operating	
Costs of goods sold	Operating	
Employee benefits not included in costs of goods sold	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's
Depreciation not included in costs of goods sold	Operating	other resources (IFRS 18.B48). See section <u>3.4</u> .
Expected credit losses on trade receivables	Operating	
Share of profit of associates and joint ventures (accounted for using the equity method)	Investing	Share of profit or loss from associates and joint ventures accounted for using the equity method is also classified in the investing category. See section <u>3.2</u> .
Fair value gains on equity instruments	Investing	Financial assets and investment property generate a return individually and largely independently of the entity's other
Fair value gains on investment property	Investing	resources (IFRS 18.53(c)), however, Entity S does not invest in those assets as main business activities, therefore, income and expenses are classified in the investing category. See sections 3.2 and $3.8.1$.
Interest expense on bank loans	<u>Operating</u>	Bank loans arise from transactions that involve only the raising of finance (IFRS 18.59(a)), and interest expense arises from the subsequent measurement of the liability (IFRS 18.60). Entity S provides financing to customers as a main business activity, and Entity S concludes that the income and expenses arising from the bank loan is related to providing finance to customers (IFRS 18.65(a)(i)). Therefore, interest expense is classified in the operating category. See sections 3.3 and 3.8.2.
Interest income on contract assets	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). See section <u>3.4</u> .
Income tax expense	Income taxes	Income and expenses within the scope of IAS 12 are classified in the income taxes category. See section <u>3.5</u> .
Loss from discontinued operations	Discontinued operations	Income and expenses presented as a single line item in accordance with IFRS $5.33(a)$ are classified in the discontinued operations category. See section <u>3.6</u> .

Entity S considers the requirement in IFRS 18.24 to present additional subtotals beyond those that are mandatory (see section 5.1) and concludes that it will present <u>four</u> additional subtotals:

Additional subtotal	Description
Gross profit	The difference between revenue and cost of goods sold.
Net interest income	The difference between interest income on contract assets and interest expense on bank loans.
Profit or loss before income taxes	All income and expenses other than those classified in the income taxes and discontinued operations categories.
Profit or loss from continuing operations	All income and expenses other than those classified in the discontinued operations category.

Entity S presents its statement of profit or loss for the year-ended 31 December 20x7 as follows.

	31 December 20x7	31 December 20x6	Explanation
Revenue	xxx	xxx	
Cost of goods sold	xxx	xxx	
Gross profit	XXX	ххх	Additional subtotal: revenue less cost of goods sold
Interest income on contract assets	xxx	ххх	
Interest expense on bank loans	xxx	xxx	
Net interest income	XXX	XXX	Additional subtotal: the difference between interest income on contract assets and interest expense on bank loans.
Employee benefits not included in costs of goods sold	xxx	ххх	
Depreciation not included in costs of goods sold	XXX	ххх	
Expected credit losses on trade receivables	XXX	XXX	
Operating profit	XXX	ххх	Mandatory subtotal: The total of all income and expenses classified in the operating category.
Share of profit of associates and joint ventures	xxx	XXX	
Fair value gains on equity instruments	xxx	XXX	
Fair value gains on investment property	xxx	XXX	
Profit before financing and income taxes	XXX	XXX	Mandatory subtotal: The total of operating profit or loss and all income and expenses classified in the investing category.
			In this scenario, Entity S does not have any income and expenses classified in the financing category. However, IFRS 18.69(b) requires this subtotal to be presented for most entities (see 5.1.3 for explanation of situations where entities are forbidden from presenting this subtotal).
Income tax expense	xxx	xxx	
Profit from continuing operations	XXX	XXX	Additional subtotal: all income and expenses other than those classified in the discontinued operations category.
Loss from discontinued operations	xxx	xxx	
Profit	xxx	xxx	Mandatory total

7.4. Example D – entity that provides financing to customers as a main business activity and invests in investment property measured at fair value as a main business activity

Scenario D: Entity S provides financing to customers as a main business activity and invests in investment property measured at fair value as a main business activity

Differences from Example A are in <u>red underlined text</u>.

The income and expenses described in the common fact pattern are classified as follows:

Income or expense	Classification	Explanation
Revenue from contracts with customers	Operating	
Costs of goods sold	Operating	
Employee benefits not included in costs of goods sold	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's
Depreciation not included in costs of goods sold	Operating	other resources (IFRS 18.B48). See section <u>3.4</u> .
Expected credit losses on trade receivables	Operating	
Share of profit of associates and joint ventures (accounted for using the equity method)	Investing	Share of profit or loss from associates and joint ventures accounted for using the equity method is also classified in the investing category. See section <u>3.2</u> .
Fair value gains on equity instruments	Investing	Assets generate a return individually and largely independently of the entity's other resources (IFRS 18.53(c)), however, Entity S does not invest in those assets as a main business activity, therefore, income and expenses are classified in the investing category. See sections <u>3.3</u> and <u>3.8.1</u> .
<u>Fair value gains on investment</u> property	<u>Operating</u>	Financial assets such as investment property generate a return individually and largely independently of the entity's other resources (IFRS 18.53(c)). Entity S invests in investment property as a main business activity; therefore, income and expenses are classified in the operating category (IFRS 18.58). See sections 3.2 and 3.8.1.
Interest expense on bank loans	Operating	Bank loans arise from transactions that involve only the raising of finance (IFRS 18.59(a)), and interest expense arises from the subsequent measurement of the liability (IFRS 18.60). Entity S provides financing to customers as a main business activity, and Entity S concludes that the income and expenses arising from the bank loan is related to providing finance to customers (IFRS 18.65(a)(i)). Therefore, interest expense is classified in the operating category. See sections 3.3 and 3.8.2.
Interest income on contract assets	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). See section <u>3.4</u> .
Income tax expense	Income taxes	Income and expenses within the scope of IAS 12 are classified in the income taxes category. See section <u>3.5</u> .
Loss from discontinued operations	Discontinued operations	Income and expenses presented as a single line item in accordance with IFRS 5.33(a) are classified in the discontinued operations category. See section <u>3.6</u> .

Entity S considers the requirement in IFRS 18.24 to present additional subtotals beyond those that are mandatory (see section 5.1) and concludes that it will present <u>four</u> additional subtotals:

Additional subtotal	Description
Gross profit	The difference between revenue and cost of goods sold.
Net interest income	The difference between interest income on contract assets and interest expense on bank loans.
Profit or loss before income taxes	All income and expenses other than those classified in the income taxes and discontinued operations categories.
Profit or loss from continuing operations	All income and expenses other than those classified in the discontinued operations category.

Entity S presents its statement of profit or loss for the year-ended 31 December 20x7 as follows.



	31 December 20x7	31 December 20x6	Explanation
Revenue	xxx	xxx	
Cost of goods sold	xxx	xxx	
Gross profit	xxx	XXX	Additional subtotal: revenue less cost of goods sold
Interest income on contract assets	XXX	XXX	
Interest expense on bank loans	xxx	XXX	
Net interest income	XXX	XXX	Additional subtotal: the difference between interest income on contract assets and interest expense on bank loans.
Fair value gains on investment property	XXX	XXX	
Employee benefits not included in costs of goods sold_	XXX	XXX	
Depreciation not included in costs of goods sold	xxx	xxx	
Expected credit losses on trade receivables	xxx	xxx	
Operating profit	XXX	XXX	Mandatory subtotal: The total of all income and expenses classified in the operating category.
Share of profit of associates and joint ventures	xxx	xxx	
Fair value gains on equity instruments	xxx	xxx	
Profit before financing and income taxes	xxx	XXX	Mandatory subtotal: The total of operating profit or loss and all income and expenses classified in the investing category.
			In this scenario, Entity S does not have any income and expenses classified in the financing category. However, IFRS 18.69(b) requires this subtotal to be presented for most entities (see 5.1.3 for explanation of situations where entities are forbidden from presenting this subtotal).
Profit before income taxes	XXX	XXX	Additional subtotal: all income and expenses other than those classified in the income taxes and discontinued operations categories.
Income tax expense	XXX	xxx	
Profit from continuing operations	XXX	XXX	Additional subtotal: all income and expenses other than those classified in the discontinued operations category.
Loss from discontinued operations	xxx	xxx	
Profit	xxx	ххх	Mandatory total

7.5. Example E – insurance company that invests in financial assets as a main business activity

Scenario E: insurance company that invests in financial assets as a main business activity

Fact pattern

Scenario E is independent of Scenarios A – D.

Entity T is an insurance company and has the following income and expenses:

- Insurance revenue
- Insurance service expenses
- Net expense from reinsurance contracts
- ► Finance expense from insurance contracts
- ► Finance income from reinsurance contracts
- Employee benefits
- Depreciation
- Changes in fair value from investment in associates and joint ventures (measured at fair value through profit or loss in accordance with IAS 28.18)
- Interest expense on bank loans
- Interest revenue calculated using the effective interest rate method
- ► Fair value gains on equity instruments
- Income tax expense
- Loss from discontinued operations

Entity T has applied the aggregation and disaggregation requirements of IFRS 18 and has determined that each of these classes of income and expense should be presented separately in the statement of profit or loss. Entity T has also considered the requirements of IFRS 18.75(c) to present separate line items for income and expenses related to insurance contracts (see section $\frac{4.3}{2}$).

Entity T presents operating expenses by nature.

Analysis

The income and expenses are classified as follows:

Income or expense	Classification	Explanation
Insurance revenue	Operating	
Insurance service expenses	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's
Net expense from reinsurance contracts	Operating	other resources (IFRS 18.B48). See section <u>3.4</u> .
Finance expense from insurance contracts	Operating	IFRS 18.64(b) requires insurance finance income and expenses
Finance income from reinsurance contracts	Operating	to be classified in the operating category. See section 3.3 .
Employee benefits	Operating	
Depreciation	Operating	Income and expenses from assets that do not generate a return individually and largely independently of the entity's other resources (IFRS 18.B48). See section <u>3.4</u> .

⁵ IAS 28.18 provides an entity an irrevocable election to measure certain investments in associates and joint ventures at fair value through profit or loss. This includes investment-linked insurance funds.

Change in fair value from investments in associates and joint ventures	Operating	The associates and joint ventures are not accounted for using the equity method; therefore, it is necessary to assess whether Entity T invests in associates and joint ventures as a main business Activity. Entity T concludes that it does,
		therefore, income and expenses are classified in the operating category. See section $3.8.1$.
Interest expense on bank loans	Financing	Bank loans arise from transactions that involve only the raising of finance (IFRS 18.59(a)), and interest expense arises from the subsequent measurement of the liability (IFRS 18.60). As Entity T does not provide financing to customers as a main business activity, interest expense is classified in the financing category. See sections <u>3.3</u> and <u>3.8.2</u> .
Interest revenue calculated using the effective interest rate method	Operating	Financial assets such as equity instrument and financial assets at amortised cost generate returns individually and largely
Fair value gains on equity instruments	Operating	independently of the entity's other resources (IFRS 18.53(c)). Entity T invests in financial assets as a main business activity; therefore, income and expenses are classified in the operating category (IFRS 18.58). See sections <u>3.2</u> and <u>3.8.1</u> .
Income tax expense	Income taxes	Income and expenses within the scope of IAS 12 are classified in the income taxes category. See section <u>3.5</u> .
Loss from discontinued operations	Discontinued operations	Income and expenses presented as a single line item in accordance with IFRS 5.33(a) are classified in the discontinued operations category. See section <u>3.6</u> .

Entity T considers the requirement in IFRS 18.24 to present additional subtotals beyond those that are mandatory (see section 5.1) and concludes that it will present four additional subtotals:

Additional subtotal	Description
Total investment income	All income and expenses relating to financial assets classified in the operating category. This subtotal is presented as Entity T invests in financial assets as a main business activity, so a sum of those activities is considered useful to users.
Profit or loss before income taxes	All income and expenses other than those classified in the income taxes and discontinued operations categories.
Profit or loss from continuing operations	All income and expenses other than those classified in the discontinued operations category.

Entity also considers the requirements of IFRS 17 to disaggregate information, resulting in these additional subtotals:

Additional subtotal	Description
Insurance service result	Comprises insurance revenue and insurance service expenses, including the net expense from reinsurance contracts (IFRS 17.80(a)).
Net insurance financial result	Comprises insurance finance income from reinsurance contracts and insurance finance expense from insurance contracts (IFRS 17.80(b)), 82).

Entity T presents its statement of profit or loss for the year-ended 31 December 20x7 as follows.

	31 December 20x7	31 December 20x6	Explanation
Insurance revenue	ххх	ххх	
Insurance service expenses	xxx	XXX	
Net expense from reinsurance contracts	XXX	XXX	
Insurance service result	XXX	XXX	Additional subtotal: comprises insurance revenue and insurance service expenses, including the net expense from reinsurance contracts (IFRS 17.80(a)).
Interest revenue calculated using the effective interest rate method	ххх	XXX	
Change in fair value from investments in associates and joint ventures	XXX	ххх	
Fair value gains on equity instruments	xxx	XXX	
Total investment income	XXX	XXX	Additional subtotal: all income and expenses relating to financial assets classified in the operating category.
Finance expense from insurance contracts	xxx	ХХХ	
Finance income from reinsurance contracts	xxx	XXX	
Net insurance financial result	ххх	XXX	Additional subtotal: comprises insurance finance income from reinsurance contracts and insurance finance expense from insurance contracts (IFRS 17.80(b)), 82).
Employee benefits	XXX	XXX	
Depreciation	xxx	XXX	
Operating profit and profit before financing and income taxes	XXX	XXX	Mandatory subtotal: The total of all income and expenses classified in the operating category. Mandatory subtotal: The total of operating
			profit or loss and all income and expenses classified in the investing category.
			Entity T does not have any income and expenses classified in the investing category; therefore, these subtotals are equal to one another.
Interest expense on bank loans	xxx	xxx	
Profit before income taxes	XXX	XXX	Additional subtotal: all income and expenses other than those classified in the income taxe and discontinued operations categories.

Income tax expense	XXX	xxx	
Profit from continuing operations	XXX	ххх	Additional subtotal: all income and expenses other than those classified in the discontinued operations category.
Loss from discontinued operations	xxx	xxx	
Profit	ххх	XXX	Mandatory total



Appendices

7.6. Appendix 1 – Defined terms

Appendix 1 includes terms defined in IFRS 18 that are used in this publication. See Appendix A to IFRS 18 for a complete list of defined terms in IFRS 18.

Defined term	Definition
aggregation	The adding together of assets, liabilities, equity, income, expenses or cash flows that share characteristics and are included in the same classification.
classification	The sorting of assets, liabilities, equity, income, expenses and cash flows based on shared characteristics.
disaggregation	The separation of an item into component parts that have characteristics that are not shared.
IFRS Accounting Standards	Accounting standards issued by the International Accounting Standards Board. They comprise: (a) International Financial Reporting Standards; (b) International Accounting Standards; (c) IFRIC Interpretations; and (d) SIC Interpretations. IFRS Accounting Standards were previously known as International Financial Reporting
management-defined performance measure	 Standards, IFRS, IFRSs and IFRS Standards. A subtotal of income and expenses that: (a) an entity uses in public communications outside financial statements; (b) an entity uses to communicate to users of financial statements management's view of an aspect of the financial performance of the entity as a whole; and (c) is not listed in paragraph 118 of IFRS 18, or specifically required to be presented or disclosed by IFRS Accounting Standards.
material information	Information is material if omitting, misstating or obscuring it could reasonably be expected to influence decisions that the primary users of general purpose financial statements make on the basis of those financial statements, which provide financial information about a specific reporting entity.
notes	Information in financial statements provided in addition to that presented in the primary financial statements.
operating profit or loss	The total of all income and expenses classified in the operating category.
other comprehensive income	Items of income and expense (including reclassification adjustments) that are recognised outside profit or loss as required or permitted by other IFRS Accounting Standards.
primary financial statements	The statement(s) of financial performance, the statement of financial position, the statement of changes in equity and the statement of cash flows.
profit or loss	The total of income less expenses included in the statement of profit or loss.
profit or loss before financing and income taxes	The total of operating profit or loss and all income and expenses classified in the investing category.
reclassification adjustments	Amounts reclassified to profit or loss in the current reporting period that were included in other comprehensive income in the current or prior periods.

total comprehensive income	The change in equity during a reporting period resulting from transactions and other events, other than those changes resulting from transactions with owners in their capacity as owners.
useful structured summary	 A structured summary provided in a primary financial statement of a reporting entity's recognised assets, liabilities, equity, income, expenses and cash flows that is useful for: (a) obtaining an understandable overview of the entity's recognised assets, liabilities, equity, income, expenses and cash flows; (b) making comparisons between entities, and between reporting periods for the same entity; and (c) identifying items or areas about which users of financial statements may wish to seek additional information in the notes.



Contact

For further information about how BDO can assist you and your organisation, please get in touch with one of our key contacts listed below.

Alternatively, please visit <u>www.bdo.global</u> where you can find full lists of regional and country contacts.

EUROPE

Evy Borry	Belgium	evy.borry@bdo.be
Anne Catherine Farlay	France	annecatherine.farlay@bdo.fr
Jens Freiberg	Germany	jens.freiberg@bdo.de
Ehud Greenberg	📼 Israel	<u>ehudg@bdo.co.il</u>
Stefano Bianchi	Italy	stefano.bianchi@bdo.it
Alon Levy	Netherlands	<u>alon.levy@bdo.nl</u>
Nina Servold Oppi	🔚 Norway	<u>nina.servold.oppi@bdo.no</u>
David Cabaleiro	S pain	david.cabaleiro@bdo.es
René Füglister	 Switzerland 	<u>rene.fueglister@bdo.ch</u>
Moses Serfaty	🚟 United Kingdom	moses.serfaty@bdo.co.uk
ASIA PACIFIC		
Aletta Boshoff	📷 Australia	aletta.boshoff@bdo.com.au
Hu Jian Fei	China	hu.jianfei@bdo.com.cn
Fanny Hsiang	Hong Kong	fannyhsiang@bdo.com.hk
Siddharth Iyer	India	siddharthiyer@mska.in
Khoon Yeow Tan	🖴 Malaysia	tanky@bdo.my
Ng Kian Hui	Singapore	kianhui@bdo.com.sg
LATIN AMERICA		
Marcello Canetti	💳 Argentina	mcanetti@bdoargentina.com
Luis Fernando Gomez Gonzalez	Colombia	lgomezg@bdo.com.co
Ernesto Bartesaghi	블 Uruguay	ebartesaghi@bdo.com.uy
-		
NORTH AMERICA & CARIBBEAN		
Craig Cross	Canada	ccross@bdo.ca
Wendy Hambleton	usa 🔤	whambleton@bdo.com
MIDDLE EAST		
Ayez Qureshi	Bahrain	ayez.qureshi@bdo.bh
Antoine Gholam	- Lebanon	agholam@bdo-lb.com
		-
SUB SAHARAN AFRICA		
Theunis Schoeman	🔀 South Africa	tschoeman@bdo.co.za

This publication has been carefully prepared, but it has been written in general terms and should be seen as broad guidance only. The publication cannot be relied upon to cover specific situations and you should not act, or refrain from acting, upon the information contained therein without obtaining specific professional advice. Neither BDO IFR Advisory Limited, and/or any other entity of BDO network, nor their respective partners, employees and/or agents accept or assume any liability or duty of care for any loss arising from any action taken or not taken by anyone in reliance on the information in this publication or for any decision based on it.

The BDO network (referred to as the 'BDO network' or the 'Network') is an international network of independent public accounting, tax and advisory firms which are members of BDO International Limited and perform professional services under the name and style of BDO (hereafter 'BDO member firms'). BDO International Limited is a UK company limited by guarantee. It is the governing entity of the BDO network.

Service provision within the BDO network in connection with International Financial Reporting Standards (IFRS) Accounting Standards (comprising IFRS Accounting Standards, International Accounting Standards, and Interpretations developed by the IFRS Interpretations Committee and the former Standing Interpretations Committee), and other documents, as issued by the International Accounting Standards Board, IFRS Sustainability Disclosure Standards as issued by the International Sustainability Standards Board, and European Sustainability Reporting Standards (ESRS) published in the Official Journal of the European Union, is provided by BDO IFR Advisory Limited, a UK registered company limited by guarantee. Service provision within the BDO network is coordinated by Brussels Worldwide Services BV, a limited liability company incorporated in Belgium.

Each of BDO International Limited, Brussels Worldwide Services BV, BDO IFR Advisory Limited and the BDO member firms is a separate legal entity and has no liability for another entity's acts or omissions. Nothing in the arrangements or rules of the BDO network shall constitute or imply an agency relationship or a partnership between BDO International Limited, Brussels Worldwide Services BV, BDO IFR Advisory Limited and/or the BDO member firms. Neither BDO International Limited nor any other central entities of the BDO network provide services to clients.

BDO is the brand name for the BDO network and for each of the BDO member firms.

© 2024 BDO IFR Advisory Limited, a UK registered company limited by guarantee. All rights reserved.

